UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 QUARTERLY PERIOD ENDED June 30, 2004

Commission File Number 0-2525

Huntington Bancshares Incorporated

Maryland (State or other jurisdiction of incorporation or organization) 31-0724920 (I.R.S. Employer Identification No.)

41 South High Street, Columbus, Ohio 43287

Registrant's telephone number (614) 480-8300

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗵 No 🗆

There were 229,685,385 shares of Registrant's without par value common stock outstanding on July 31, 2004.

Huntington Bancshares Incorporated

INDEX

| Part I. | Financial Information | |
|--------------|--|----|
| Item 1. | Financial Statements (Unaudited) | |
| | Condensed Consolidated Balance Sheets at June 30, 2004, December 31, 2003, and June 30, 2003 | 3 |
| | Condensed Consolidated Statements of Income for the three and six months ended June 30, 2004 and 2003 | 4 |
| | Condensed Consolidated Statements of Changes in Shareholders' Equity for the six months ended June 30, 2004 and 2003 | 5 |
| | Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2004 and 2003 | 6 |
| | Notes to Unaudited Condensed Consolidated Financial Statements | 7 |
| Item 2. | Management's Discussion and Analysis of Financial Condition and Results of Operations | 18 |
| Item 3. | Quantitative and Qualitative Disclosures about Market Risk | 74 |
| Item 4. | Controls and Procedures | 74 |
| Part II. | Other Information | |
| Item 2. | Changes in Securities, Use of Proceeds, and Issuer Purchases of Equity Securities | 75 |
| Item 4. | | |
| | Submission of Matters to a Vote of Shareholders | 75 |
| Item 6. | Exhibits and Reports on Form 8-K | 76 |
| Signatures 1 | | 77 |
| | 2 | |

Part 1. Financial Information Item 1. Financial Statements Huntington Bancshares Incorporated Condensed Consolidated Balance Sheets

| (in thousands, except number of shares) | June 30, 2004 | December 31, 2003 | June 30, 2003 |
|---|------------------|----------------------|------------------|
| | (Unaudited) | | (Unaudited) |
| Assets | | | |
| Cash and due from banks | \$ 1,162,995 | \$ 899,689 | \$ 1,153,108 |
| Federal funds sold and securities purchased under resale agreements | 193,772 | 96,814 | 74,473 |
| Interest bearing deposits in banks | 24,009 | 33,627 | 44,906 |
| Trading account securities | 20,577 | 7,589 | 19,426 |
| Loans held for sale | 314,262 | 226,729 | 713,722 |
| Securities available for sale - at fair value | 4,988,270 | 4,925,232 | 3,702,761 |
| Investment securities held to maturity - fair value \$3,224; \$3,937 and \$6,780, respectively | 3,169 | 3,828 | 6,593 |
| Loans and leases | 21,775,669 | 21,075,118 | 19,059,533 |
| Allowance for loan and lease losses | (286,935) | (299,732) | (307,667) |
| Net loans and leases | 21,488,734 | 20,775,386 | 18,751,866 |
| Operating lease assets | 888,612 | 1,260,440 | 1,672,608 |
| Bank owned life insurance | 944,892 | 927,671 | 906,823 |
| Premises and equipment | 354,534 | 349,712 | 332,916 |
| Goodwill and other intangible assets | 216,215 | 217,009 | 218,080 |
| Customers' acceptance liability | 6,613 | 9,553 | 8,372 |
| Accrued income and other assets | 814,552 | 786,047 | 731,559 |
| Total Assets | \$ 31,421,206 | \$ 30,519,326 | \$ 28,337,213 |
| *** | | | |
| Liabilities | 0 10 4CE 14C | ¢ 10 407 205 | ¢ 10.271.250 |
| Deposits | \$ 19,465,146 | \$ 18,487,395 | \$ 18,371,359 |
| Short-term borrowings | 1,130,830 | 1,452,304 | 918,771 |
| Federal Home Loan Bank advances | 1,270,455 | 1,273,000 | 1,273,000 |
| Other long-term debt Subordinated notes | 4,557,373 | 4,544,509 | 3,508,397 |
| | 1,011,506 | 990,470 | 496,666 |
| Company obligated mandatorily redeemable preferred capital securities of subsidiary trusts holding solely | | | 200.000 |
| junior subordinated debentures of the parent company | 31,193 | 25 522 | 300,000 |
| Allowance for unfunded loan commitments and letters of credit Bank acceptances outstanding | 6.613 | 35,522 9,553 | 33,280 |
| 1 8 | -) | -) | 8,372 |
| Accrued expenses and other liabilities | 1,561,721 | 1,451,571 | 1,225,169 |
| Total Liabilities | 29,034,837 | 28,244,324 | 26,135,014 |
| Shareholders' Equity | | | |
| Preferred stock - authorized 6,617,808 shares; none outstanding | _ | _ | _ |
| Common stock - without par value; authorized 500,000,000 shares; issued 257,866,255 shares; outstanding 229,475,821; 229,008,088 and 228,660,038 shares, respectively | 2,482,069 | 2,483,542 | 2.483.105 |
| Less 28,390,434; 28,858,167 and 29,206,217 treasury shares, respectively | (539,852) | (548,576) | (555,176) |
| Accumulated other comprehensive income (loss) | (27,204) | 2,678 | 40,817 |
| Retained earnings | 471,356 | 337,358 | 233,453 |
| Total Shareholders' Equity | 2,386,369 | 2,275,002 | 2,202,199 |
| Total Liabilities and Shareholders' Equity | \$ 31,421,206 | \$ 30,519,326 | \$ 28,337,213 |
| | | | |

See notes to unaudited condensed consolidated financial statements

Huntington Bancshares Incorporated Condensed Consolidated Statements of Income (Unaudited)

| | | nths Ended e 30, | Six Months Ended June 30, | | |
|--|--------------------|---------------------|------------------------------|------------------|--|
| (in thousands, except per share amounts) | 2004 | 2003 | 2004 | 2003 | |
| Interest and fee income | | | | | |
| Loans and leases | | | | | |
| Taxable | \$ 268,409 | \$ 265,694 | \$ 538,771 | \$ 535,939 | |
| Taxable-exempt | 444 | 675 | 950 | 1,409 | |
| Securities | | | | | |
| Taxable | 43,926 | 37,014 | 88,072 | 74,139 | |
| Taxable-exempt | 7,247 | 5,019 | 14,760 | 9,972 | |
| Other | 4,141 | 8,923 | 7,545 | 15,880 | |
| Total Interest Income | 324,167 | 317,325 | 650,098 | 637,339 | |
| Interest expenses | | | | | |
| Deposits | 59,372 | 76,383 | 118,998 | 156,093 | |
| Short-term borrowings | 2,789 | 4,313 | 6,102 | 9,872 | |
| Federal Home Loan Bank advances | 8,098 | 5,634 | 16,139 | 11,219 | |
| Subordinated notes and other long-term debt including preferred capital securities | 31,345 | 28,554 | 63,611 | 55,955 | |
| Total Interest Expense | 101,604 | 114,884 | 204,850 | 233,139 | |
| | | | | | |
| Net Interest Income | 222,563 | 202,441 | 445,248 | 404,200 | |
| Provision for credit losses | 5,027 | 49,193 | 30,623 | 86,037 | |
| Net Interest Income After Provision for Credit Losses | 217,536 | 153,248 | 414,625 | 318,163 | |
| Operating lease income | 78,706 | 128,574 | 167,573 | 266,767 | |
| Service charges on deposit accounts | 43,596 | 40,914 | 85,433 | 80,783 | |
| Trust services | 16,708 | 15,580 | 33,031 | 30,783 | |
| Brokerage and insurance income | 13,523 | 14,196 | 28,720 | 29,693 | |
| Mortgage banking | 23,322 | 7,185 | 19,026 | 18,310 | |
| Bank owned life insurance income | 11,309 | 11,043 | 21,794 | 22,180 | |
| Gain on sales of automobile loans | 4,890 | 13,496 | 13,894 | 23,751 | |
| Other service charges and fees | 10,645 | 11,372 | 20,158 | 21,710 | |
| Securities gains (losses) | (9,230) | 6,887 | 5,860 | 8,085 | |
| Other | 24,659 | 27,704 | 50,278 | 48,105 | |
| Total Non-Interest Income | 218,128 | 276,951 | 445,767 | 549,875 | |
| | 110 515 | 105.242 | 241 220 | 010 001 | |
| Personnel costs | 119,715 | 105,242 | 241,339 | 218,331 | |
| Operating lease expense | 62,563 | 102,939 | 133,273 | 214,527 | |
| Outside data processing and other services | 17,563 | 16,104 | 36,025 | 32,683 | |
| Equipment | 16,228 | 16,341 | 32,314 | 32,753 31,986 | |
| Net occupancy | 16,258 7,836 | 15,377 9,872 | 33,021 | 19,157 | |
| Professional services Marketing | 8,069 | 8,454 | 15,135 15,908 | 15,080 | |
| Telecommunications | 4,638 | 5,394 | 9,832 | 11,095 | |
| Printing and supplies | 3,098 | 2,253 | 6,114 | 5,934 | |
| Amortization of intangibles | 204 | 2,255 | 408 | 408 | |
| Restructuring reserve releases | 204 | (5,315) | 400 | (6,315) | |
| Other | 25,981 | 20,168 | 44,438 | 36,873 | |
| Total Non-Interest Expense | 282,153 | 297,033 | 567,807 | 612,512 | |
| Income Before Income Taxes | 153,511 | 133,166 | 292,585 | 255,526 | |
| Provision for income taxes | 43,384 | 36,676 | 78,285 | 67,306 | |
| Net Income | \$ 110,127 | \$ 96,490 | \$ 214,300 | \$188,220 | |
| Auguage Common Shares | | | | | |
| | 229,429 | 228,633 | 229,328 | 229,987 | |
| Average Common Shares: Basic | | | | | |
| Average Common Shares: Basic Diluted | 232.659 | 230,572 | 232.787 | 231.684 | |
| Basic | 232,659 | 230,572 | 232,787 | 231,684 | |
| Basic Diluted | 232,659 \$ 0.48 | 230,572 \$ 0.42 | 232,787 \$ 0.93 | | |
| Basic Diluted Per Common Share: | | | | | |

See notes to unaudited condensed consolidated financial statements

Huntington Bancshares Incorporated Consolidated Statements of Changes in Shareholders' Equity

| 2 | Common Stock | | Treasury Shares | | Accumulated Other | | | |
|--|--------------|--------------|-----------------|--------------|----------------------|-----------------------|----------------------|--------------|
| (in thousands) | Shares | Amount | Shares | Amount | | nprehensive Income | Retained Earnings | Total |
| Six Months Ended June 30, 2003 (Unaudited): | | | | | | | | |
| Balance, beginning of period | 257,866 | \$ 2,484,421 | (24,987) | \$ (475,399) | \$ | 62,300 | \$118,471 | \$ 2,189,793 |
| Comprehensive Income: | | | | | | | | |
| Net income | | | | | | | 188,220 | 188,220 |
| Unrealized net holding losses on securities available for sale arising during the period, net of reclassification adjustment for net | | | | | | (1.201) | | (4.201) |
| gains included in net income | | | | | | (4,391) | | (4,391) |
| Unrealized losses on derivative instruments used in cash flow hedging relationships | | | | | | (17,092) | | (17,092) |
| Total comprehensive income | | | | | | | | 166,737 |
| Cash dividends declared (\$0.32 per share) | | | | | | | (73,238) | (73,238) |
| Stock options exercised | | (1,316) | 118 | 1,902 | | | (,) | 586 |
| Treasury shares purchased | | ()) | (4,300) | (81,061) | | | | (81,061) |
| Other | | | (37) | (618) | | | | (618) |
| Balance, end of period (Unaudited) | 257,866 | \$ 2,483,105 | (29,206) | \$ (555,176) | \$ | 40,817 | \$ 233,453 | \$ 2,202,199 |
| Six Months Ended June 30, 2004 (Unaudited): | | | | | | | | |
| Balance, beginning of period | 257,866 | \$ 2,483,542 | (28,858) | \$ (548,576) | \$ | 2,678 | \$ 337,358 | \$ 2,275,002 |
| Comprehensive Income: | | | | | | | | |
| Net income | | | | | | | 214,300 | 214,300 |
| Unrealized net holding losses on securities available for sale arising during the period, net of reclassification adjustment | | | | | | | | |
| for net gains included in net income | | | | | | (52,165) | | (52,165) |
| Unrealized gains on derivative instruments used in cash flow hedging relationships | | | | | | 22,283 | | 22,283 |
| Total comprehensive income | | | | | | | | 184,418 |
| Cash dividends declared (\$0.35 per share) | | | | | | | (80,302) | (80,302) |
| Stock options exercised | | (951) | 442 | 8,467 | | | | 7,516 |
| Other | | (522) | 26 | 257 | | | | (265) |
| Balance, end of period (Unaudited) | 257,866 | \$ 2,482,069 | (28,390) | \$ (539,852) | \$ | (27,204) | \$ 471,356 | \$ 2,386,369 |
| • · · · | | | | , | | | | |

5

See notes to unaudited condensed consolidated financial statements.

Huntington Bancshares Incorporated Consolidated Statements of Cash Flows (Unaudited)

| | Six Months E | nded June 30, |
|--|--------------|---------------|
| in thousands) | 2004 | 2003 |
| perating Activities | | |
| Net Income | \$ 214,300 | \$ 188,22 |
| Adjustments to reconcile net income to net cash provided by operating activities | | |
| Provision for credit losses | 30,623 | 86,03 |
| Depreciation on operating lease assets | 132,388 | 187,91 |
| Other depreciation and amortization | 38,995 | 42,20 |
| Deferred income tax expense | 66,243 | 31,47 |
| Increase in trading account securities | (12,988) | (19,18 |
| Increase in mortgages held for sale | (87,783) | (185,34 |
| Gains on sales of securities available for sale | (5,860) | (8,08 |
| Gains on sales/securitizations of loans | (15,407) | (30,79 |
| Restructuring reserve releases | (13,407) | (6,31 |
| Other, net | (8,146) | (68,00 |
| | | 210.10 |
| Net Cash Provided by Operating Activities | 352,365 | 218,13 |
| vesting Activities | | ·= |
| Decrease (increase) in interest bearing deposits in banks | 9,618 | (7,60 |
| Proceeds from: | | |
| Maturities and calls of investment securities held to maturity | 670 | 95 |
| Maturities and calls of securities available for sale | 544,419 | 945,53 |
| Sales of securities available for sale | 885,554 | 591,49 |
| Purchases of securities available for sale | (1,457,477) | (1,649,72 |
| Proceeds from sales/securitizations of loans | 1,382,596 | 1,390,37 |
| Net loan and lease originations, excluding sales | (2,234,989) | (2,119,93 |
| Net decrease in operating lease assets | 240,523 | 340,00 |
| Proceeds from sale of premises and equipment | 334 | 4,04 |
| Purchases of premises and equipment | (29,298) | (22,22 |
| Proceeds from sales of other real estate | 6,460 | 4,87 |
| Net Cash Used for Investing Activities | (651,590) | (522,19 |
| | | |
| nancing Activities | 003 101 | 0.00.01 |
| Increase in deposits | 982,401 | 869,31 |
| Decrease in short-term borrowings | (321,474) | (1,222,24 |
| Proceeds from issuance of subordinated notes | 148,830 | |
| Maturity of subordinated notes | (100,000) | (250,00 |
| Proceeds from Federal Home Loan Bank advances | 455 | 270,00 |
| Maturity of Federal Home Loan Bank advances | (3,000) | (10,00 |
| Proceeds from issuance of long-term debt | 625,000 | 1,235,00 |
| Maturity of long-term debt | (600,000) | (225,00 |
| Dividends paid on common stock | (80,239) | (73,71 |
| Repurchases of common stock | _ | (81,06 |
| Net proceeds from issuance of common stock | 7,516 | 58 |
| Net Cash Provided by Financing Activities | 659,489 | 512,87 |
| | <u></u> | |
| Change in Cash and Cash Equivalents | 360,264 | 208,81 |
| Cash and Cash Equivalents at Beginning of Period | 996,503 | 1,018,76 |
| Cash and Cash Equivalents at End of Period | \$ 1,356,767 | \$ 1,227,58 |
| | | , , , |
| Income taxes paid | \$ 9,490 | \$ 65,85 |
| Interest paid | | |
| | 206,500 | 247,12 |
| Non-cash activities | | |
| Residential mortgage loans securitized and retained in securities available for sale | 115,929 | 171,58 |
| Common stock dividends accrued not paid | 31,562 | 27,93 |

See notes to unaudited condensed consolidated financial statements.

Notes to Unaudited Condensed Consolidated Financial Statements

Note 1 - Basis of Presentation

The accompanying unaudited condensed consolidated financial statements of Huntington Bancshares Incorporated (Huntington) reflect all adjustments consisting of normal recurring accruals, which are, in the opinion of Management, necessary for a fair presentation of the consolidated financial position, the results of operations, and cash flows for the periods presented. These unaudited condensed consolidated financial statements have been prepared according to the rules and regulations of the Securities and Exchange Commission (SEC) and, therefore, certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States (GAAP) have been omitted. The Notes to the Consolidated Financial Statements appearing in Huntington's 2003 Annual Report on Form 10-K (2003 Form 10-K), which include descriptions of significant accounting policies, as updated by the information contained in this report, should be read in conjunction with these interim financial statements.

Certain amounts in the prior year's financial statements have been reclassified to conform to the 2004 presentation.

For statement of cash flows purposes, cash and cash equivalents are defined as the sum of "Cash and due from banks" and "Federal funds sold and securities purchased under resale agreements".

Note 2 - New Accounting Pronouncements

Emerging Issues Task Force Issue No. 03-1, *The Meaning of Other-Than-Temporary Impairments and Its Application to Certain Investments*(EITF 03-1): The Emerging Issues Task Force reached a consensus about the criteria that should be used to determine when an investment is considered impaired, whether that impairment is other than temporary, and the measurement of an impairment loss. EITF 03-1 also included accounting considerations subsequent to the recognition of an other-than-temporary impairment and requires certain disclosures about unrealized losses that have not been recognized as other-than-temporary impairments. EITF 03-1 will apply to the evaluation of other-than-temporary impairment in reporting periods beginning after June 15, 2004.

At June 30, 2004, Huntington had \$3.3 billion of securities whose fair value was less than book value. The unrealized loss in these securities totaled \$88.9 million. Of these securities, Huntington held \$3.1 billion of securities, with an unrealized loss of \$80.0 million, where the security cannot be contractually prepaid or where Huntington would recover substantially all of its cost if the securities were contractually prepaid. Management continues to evaluate its options with respect to adopting this new accounting guidance and cannot currently estimate the impact of adopting EITF 03-1.

Emerging Issues Task Force Issue No. 03-16 *Accounting for Investment in Limited Liability Companies* (EITF 03-16): The Task Force reached a consensus that an investment in a limited liability company (LLC) that maintains a "specific ownership account" for each investor should be viewed as similar to an investment in a limited partnership for purposes of determining whether a noncontrolling investment in an LLC should be accounted for using the cost method or the equity method. The current rules require a noncontrolling investment in a limited partnership to be accounted for under the equity method unless the interest is so minor that the limited partner may have virtually no influence over the partnership operating and financial policies. The guidance for evaluating an investment in a LLC should be applied for reporting periods beginning after June 15, 2004. The impact of EITF 03-16 is not expected to be material to Huntington's financial condition, results of operations, or cash flows.

SEC Staff Accounting Bulletin No. 105, *Application of Accounting Principles to Loan Commitments* (SAB 105): On March 9, 2004, the SEC issued SAB 105, which summarizes the views of the SEC staff regarding the application of generally accepted accounting principles to loan commitments accounted for as derivative instruments. Specifically, SAB 105 indicated that the fair value of loan commitments that are required to follow derivative accounting under FAS 133, *Accounting for Derivative Instruments and Hedging Activities*, should not consider the expected future cash flows related to the associated servicing of the future loan. Prior to SAB 105, Huntington did not consider the expected future cash flows related to the associated servicing in determining the fair value of loan commitments. The adoption of SAB 105 did not have a material effect on Huntington's financial results.

FASB Staff Position No. 106-2, *Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003*(FSP **106-2**): In December 2003, a law was approved that expands Medicare benefits, primarily adding a prescription drug benefit for Medicare-eligible retirees beginning in 2006. The law also provides a federal subsidy to companies that sponsor postretirement benefit plans providing prescription drug coverage. FSP 106-2 was issued in May 2004 and supersedes FSP 106-1 issued in January 2004. FSP 106-2 specifies that any Medicare subsidy must be taken into account in measuring the employer's postretirement health care benefit obligation and will also reduce the net periodic postretirement cost in future periods. The new guidance is effective for the reporting periods beginning on or after June 15, 2004. Accordingly, the postretirement benefit obligations and net periodic costs reported in the accompanying financial statements and notes do not reflect the impact of this legislation. The impact of this new pronouncement is not expected to be material to Huntington's financial condition, results of operations, or cash flows.

AICPA Statement of Position No. 03-3, Accounting for Certain Loans or Debt Securities Acquired in a Transfer (SOP 03-3): In December 2003, the Accounting Standards Executive Committee of the American Institute of Certified Public Accountants issued SOP 03-3 to address accounting for differences between the contractual cash flows of certain loans and debt securities and the cash flows expected to be collected when loans or debt securities are acquired in a transfer and those cash flow differences are attributable, at least in part, to credit quality. As such, SOP 03-3 applies to loans and debt securities acquired in purchase business combinations and does not apply to originated loans. The application of SOP 03-3 limits the interest income, including accretion of purchase price discounts, that may be recognized for certain loans and debt securities. Additionally, SOP 03-3 requires that the excess of contractual cash flows over cash flows expected to be collected (nonaccretable difference) not be recognized as an adjustment of yield or valuation allowance, such as the allowance for credit losses. Subsequent to the initial investment, increases in expected cash flows generally should be recognized prospectively through adjustment of the yield on the loan or debt security over its remaining life. Decreases in expected cash flows should be recognized as impairment. SOP 03-3 is effective for loans and debt securities acquired in fiscal years beginning after December 15, 2004, with early application encouraged. The impact of this new pronouncement is not expected to be material to Huntington's financial condition, results of operations, or cash flows.

FASB Statement No. 148, Accounting for Stock-Based Compensation—Transition and Disclosure (FAS 148): FAS 148 was issued in December 2002, as an amendment of Statement No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition to FAS 123's fair value method of accounting for stock-based employee compensation. FAS 148 also amends the disclosure provisions of FAS 123 and Accounting Principles Board (APB) Opinion No. 28, Interim Financial Reporting (APB 28), to require disclosure in the summary of significant accounting policies of the effects of an entity's accounting policy with respect to stock-based employee compensation on reported net income and earnings per share in annual and interim financial statements. While FAS 148 does not require companies to account for employee stock options using the fair value method, the disclosure provisions of FAS 148 are applicable to all companies with stock-based employee compensation, regardless of whether they account for that compensation using the fair value method of FAS 123 or the intrinsic value method of APB 25, which is the method currently used by Huntington. See note 11 for the disclosures.

Note 3 - Securities and Exchange Commission Investigation

As previously disclosed, the Securities and Exchange Commission (SEC) is conducting a formal investigation regarding certain financial accounting and disclosure matters, including certain matters that were the subject of prior restatements by Huntington. The SEC staff has notified Huntington that the staff is considering recommending to the SEC that it institute enforcement action against Huntington and certain of its senior officers for, among other possible matters, violations of various provisions of the Securities Exchange Act of 1934 and the Securities Act of 1933 in connection with certain financial accounting matters relating to fiscal years 2002 and earlier, and certain related disclosure matters.

Huntington is presently in negotiations with the staff of the SEC regarding a settlement of its investigation. Huntington's chief executive officer has indicated he accepts responsibility in his position as the chief executive officer for matters that occur on his watch, and has indicated a willingness to negotiate a settlement with the SEC of these matters. Huntington expects that a settlement of this matter would involve the entry of an order requiring, among other possible matters, Huntington to comply with various provisions of the Securities Exchange Act of 1934 and the Securities Act of 1933, along with the imposition of a fine and other possible measures. No assurances, however, can be provided as to the ultimate timing or outcome of these matters.

Huntington also remains in active dialogue with its bank regulators concerning these matters and is working diligently with the regulators to resolve them in a manner that permits it to proceed with its pending Unizan acquisition. No assurances, however, can be provided as to the ultimate timing or outcome of these matters.

Donald R. Kimble, who joined Huntington earlier this year as executive vice president in the Finance area, has been named chief financial officer and controller. Prior to joining Huntington, Kimble was executive vice president and controller of AmSouth Bancorporation, and previously held various subsidiary chief financial officer or accounting positions at Bank One Corporation. Michael J. McMennamin and John D. Van Fleet have relinquished their positions of chief financial officer and controller, respectively. Both remain with the company.

Note 4 - Pending Acquisition

On January 27, 2004, Huntington announced the signing of a definitive agreement to acquire Unizan Financial Corp. (Unizan), a financial holding company based in Canton, Ohio, with \$2.7 billion of assets at December 31, 2003. Under the terms of the agreement, Unizan shareholders will receive 1.1424 shares of Huntington common stock, on a tax-free basis, for each share of Unizan. Based on the \$23.10 closing price of Huntington's common stock on January 26, 2004, this represented a price of \$26.39 per Unizan share, and valued the transaction at approximately \$587 million. Both boards unanimously approved the merger, and on May 25, 2004, Unizan shareholders approved the merger. Approval by Huntington shareholders is not required.

As reported on June 16, 2004, the Federal Reserve Board had informed Huntington that it had extended its review period to coordinate further with the staff of the SEC regarding the SEC's ongoing formal investigation of Huntington and to complete its review of the Community Reinvestment Act aspects of the merger. Huntington remains in active dialogue with its bank regulators concerning these matters and is working diligently with the regulators to resolve them in a manner that permits it to proceed with its pending Unizan acquisition. No assurances, however, can be provided as to the ultimate timing or outcome of these matters (see Note 3 for additional discussion).

Huntington and Unizan are ready to close the merger, subject to the receipt of all necessary regulatory approvals. After the merger, Huntington also intends to purchase approximately 2.5 million common shares from time-to-time in the open market or through privately negotiated transactions depending on market conditions, to offset the dilutive effect of issuing shares to Unizan shareholders.

Note 5 - Stock Repurchase Plan

Effective April 27, 2004, the board of directors authorized a new share repurchase program (the 2004 Repurchase Program) which cancelled the 2003 Repurchase Program and authorized Management to repurchase not more than 7,500,000 shares of Huntington common stock. Any share repurchases will be made under this authorization. Purchases will be made from time-to-time in the open market or through privately negotiated transactions depending on market conditions. No share repurchases were made under the 2004 repurchase program.

Note 6 - Operating Lease Assets

Operating lease assets at June 30, 2004, December 31, 2003, and June 30, 2003, were as follows:

| (in thousands) | June 30, 2004 | December 31, 2003 | June 30, 2003 |
|---|------------------|----------------------|------------------|
| Cost of assets under operating leases | \$ 1,629,552 | \$ 2,136,502 | \$ 2,689,413 |
| Deferred lease origination fees and costs | (1,033) | (2,117) | (44,586) |
| Accumulated depreciation | (739,907) | (873,945) | (972,219) |
| | | | |
| Operating Lease Assets, Net | \$ 888,612 | \$ 1,260,440 | \$ 1,672,608 |
| | ······ | | |

Depreciation and residual losses at termination related to operating lease assets was \$57.4 million and \$91.4 million for the three months ended June 30, 2004 and 2003, respectively. For the respective six-month periods, depreciation and residual losses at termination was \$121.3 million and \$190.7 million.

Note 7 – Securities Available for Sale

Securities available for sale at June 30, 2004, December 31, 2003, and June 30, 2003 were as follows:

| | June 30, 2004 | | Decembe | er 31, 2003 | June 30, 2003 | |
|---|-------------------|--------------|-------------------|----------------|-------------------|--------------|
| (in thousands of dollars) | Amortized Cost | Fair Value | Amortized Cost | Fair Value | Amortized Cost | Fair Value |
| U.S. Treasury | | | | | | |
| Under 1 year | \$ 796 | \$ 800 | \$ 1,374 | \$ 1,376 | \$ 327 | \$ 331 |
| 1-5 years | 24,480 | 24,404 | 31,356 | 31,454 | 38,930 | 39,543 |
| 6-10 years | 754 | 824 | 271,271 | 275,540 | 64,063 | 66,158 |
| Over 10 years | _ | _ | | | | |
| | | | | | | |
| Total U.S. Treasury | 26,030 | 26,028 | 304,001 | 308,370 | 103,320 | 106,032 |
| Federal Agencies | | | | | | |
| Mortgage backed securities | | | | | | |
| 1-5 years | 14,181 | 14,548 | 19,899 | 20,434 | 23,879 | 24,661 |
| 6-10 years | 155,460 | 155,628 | 198,755 | 201,995 | 248,896 | 254,564 |
| Over 10 years | 1,352,082 | 1,331,790 | 1,593,139 | 1,595,594 | 1,668,613 | 1,699,875 |
| 2 | | | | | | |
| Total Mortgage-Backed | 1,521,723 | 1,501,967 | 1,811,793 | 1,818,023 | 1,941,388 | 1,979,100 |
| Other Federal Agencies | | | | | | |
| Under 1 year | 116,357 | 118,776 | 173,181 | 175,505 | 137,797 | 141,375 |
| 1-5 years | 728,472 | 719,339 | 585,561 | 593,662 | 294,977 | 313,029 |
| 6-10 years | 343,226 | 322,398 | 403,953 | 390,164 | 171,343 | 169,976 |
| Over 10 years | | | 201 | 192 | | |
| 2 | | | | | | |
| Total Other Agencies | 1,188,055 | 1,160,512 | 1,162,896 | 1,159,523 | 604,117 | 624,380 |
| Total U.S. Treasury and Other Federal Agencies | 2,735,808 | 2,688,507 | 3,278,690 | 3,285,916 | 2,648,825 | 2,709,512 |
| Municipal Securities | | | | | | |
| Under 1 year | 6,679 | 6,731 | 6,594 | 6,663 | 6,639 | 6,728 |
| 1-5 years | 14,191 | 14,424 | 20,015 | 20,569 | 20,402 | 21,150 |
| 6-10 years | 69,861 | 68,928 | 69,511 | 71,013 | 52,995 | 54,707 |
| Over 10 years | 311,972 | 303,309 | 332,181 | 334,188 | 219,011 | 223,072 |
| over to years | 511,972 | 505,507 | 552,101 | | 219,011 | 223,072 |
| Total Municipal Securities | 402,703 | 393,391 | 428,301 | 432,433 | 299,047 | 305,657 |
| Private Label CMO | | | | | | |
| Under 1 year | | | 1,973 | 1,973 | | |
| 1-5 years | | | 1,975 | 1,975 | | |
| 6-10 years | — | — | — | — | — | — |
| Over 10 years | 585,920 | 577,013 | 388,933 | 388,684 | 197,080 | 198,617 |
| Over 10 years | 363,920 | 577,013 | 588,955 | 588,084 | 197,080 | 198,017 |
| Fotal Private Label CMO | 585,920 | 577,013 | 390,906 | 390,657 | 197,080 | 198,617 |
| Asset Backed Securities | | | | | | |
| Under 1 year | | | | | | |
| 1-5 years | 30,000 | 30,038 | 30,000 | 29,944 | 30,000 | 29,944 |
| 6-10 years | 11,187 | 11,339 | 20,000 | 19,984 | 50,000 | 2),)++ |
| Over 10 years | 1,074,239 | 1,075,608 | 590,826 | 589,788 | 96,875 | 96,693 |
| Total Asset Backed Securities | 1,115,426 | 1,116,984 | 640,826 | 639,716 | 126,875 | 126,637 |
| | | | | | | |
| Other | 4 (44 | 1 (10 | 500 | 500 | 1.001 | 1.002 |
| Under 1 year | 1,611 | 1,642 | 500 | 502 | 1,081 | 1,003 |
| 1-5 years | 9,703 | 9,877 | 7,169 | 7,346 | 8,605 | 8,641 |
| 6-10 years | 2,854 | 2,948 | 5,047 | 5,510 | 4,994 | 5,267 |
| Over 10 years | 193,652 | 190,545 | 145,103 | 146,685 | 136,226 | 132,948 |
| Retained interest in securitizations | | _ | 5,593 | 6,356 | 148,177 | 163,664 |
| Marketable equity securities | 6,658 | 7,364 | 8,547 | 10,111 | 50,809 | 50,815 |
| Total Other | 214,479 | 212,375 | 171,959 | 176,510 | 349,892 | 362,338 |
| Total Saguritias Available for Sale | \$ 5,054,335 | \$ 4,988,270 | \$ 4,910,682 | \$ 4,925,232 | \$ 3,621,719 | \$ 3,702,761 |
| Total Securities Available for Sale | \$ 5,054,335 | \$ 4,988,270 | \$ 4,910,082 | ۵ 4,923,232 | \$ 3,621,719 | \$ 3,702,761 |

The growth from year-end and the year-ago quarter primarily consisted of over 10-year variable-rate asset backed securities.

Note 8 - Segment Reporting

Huntington has three distinct lines of business: Regional Banking, Dealer Sales, and the Private Financial Group (PFG). A fourth segment includes Huntington's Treasury functions and capital markets activities and other unallocated assets, liabilities, revenue, and expense. Line of business results are determined based upon Huntington's management reporting system, which assigns balance sheet and income statement items to each of the business segments. The process is designed around Huntington's organizational and management structure and, accordingly, the results below are not necessarily comparable with similar information published by other financial institutions.

Management relies on "operating earnings" for review of performance and for critical decision making purposes. Operating earnings exclude the impact of the significant items listed in the reconciliation table below. See Note 12 for further discussions regarding Restructuring Reserves.

The following provides a brief description of the four operating segments of Huntington:

Regional Banking: This segment provides products and services to retail, business banking, and commercial customers. These products and services are offered in seven operating regions within the five states of Ohio, Michigan, West Virginia, Indiana, and Kentucky through the company's traditional banking network. Each region is further divided into Retail and Commercial Banking units. Retail products and services include home equity loans and lines of credit, first mortgage loans, direct installment loans, business loans, personal and business deposit products, as well as sales of investment and insurance services. Retail products and services comprise 57% and 81% of total Regional Banking loans and deposits, respectively. These products and services are delivered to customers through banking offices, ATMs, Direct Bank—Huntington's customer service center, and Web Bank at huntington.com. Commercial banking serves middle-market and commercial banking relationships, which use a variety of banking products and services including, commercial loans, international trade, cash management, leasing, interest rate protection products, capital market alternatives, 401(k) plans, and mezzanine investment capabilities.

Dealer Sales: This segment serves automotive dealerships within Huntington's primary banking markets, as well as in Arizona, Florida, Georgia, Pennsylvania, and Tennessee. This segment finances the purchase of automobiles by customers of the automotive dealerships, purchases automobiles from dealers and simultaneously leases the automobiles under long-term direct financing leases, finances dealership floor plan inventories, real estate, or working capital needs, and provides other banking services to the automotive dealerships and their owners.

Private Financial Group: This segment provides products and services designed to meet the needs of the company's higher net worth customers. Revenue is derived through trust, asset management, investment advisory, brokerage, insurance, and private banking products and services.

Treasury/Other: This segment includes revenue and expense related to assets, liabilities, and equity that are not directly assigned or allocated to one of the three business segments. Assets included in this segment include bank owned life insurance, investment securities, and mezzanine loans originated through Huntington Capital Markets Group.

A match-funded transfer pricing system is used to attribute appropriate interest income and interest expense to other business segments. The Treasury/Other segment includes the net impact of interest rate risk management, including derivative activities. Furthermore, this segment's results include the investment securities portfolios and capital markets activities. Additionally, income or expense and provision for income taxes, not allocated to other business segments, are also included.

Listed below is certain reported financial information reconciled to Huntington's three and six month 2004 and 2003 operating results by line of business.

| | Three Months Ended June 30, | | | | | | | |
|--|-----------------------------|-----------------|-----------|--------------------|----------------------------|--|--|--|
| Income Statements (in thousands) | Regional Banking | Dealer Sales | PFG | Treasury/ Other | Huntington Consolidated | | | |
| 2004 | | | | | | | | |
| Net interest income | \$ 155,083 | \$ 43,586 | \$ 11,166 | \$ 12,728 | \$ 222,563 | | | |
| Provision for credit losses | 3,949 | (8,261) | (654) | (61) | (5,027) | | | |
| Non-Interest income | 82,475 | 85,983 | 27,680 | 21,990 | 218,128 | | | |
| Non-Interest expense | (147,840) | (85,768) | (28,559) | (19,986) | (282,153) | | | |
| Provision for income taxes | (32,783) | (12,439) | (3,372) | 5,210 | (43,384) | | | |
| Net income, as reported | 60,884 | 23,101 | 6,261 | 19,881 | 110,127 | | | |
| Gain on sale of automobile loans, net of tax | _ | (2,068) | | (1,110) | (3,178) | | | |
| Operating Earnings | \$ 60,884 | \$ 21,033 | \$ 6,261 | \$ 18,771 | \$ 106,949 | | | |
| 2003 | | | | | | | | |
| Net interest income | \$ 150,418 | \$ 11,639 | \$ 9,785 | \$ 30,599 | \$ 202,441 | | | |
| Provision for credit losses | (40,525) | (9,191) | 457 | 66 | (49,193) | | | |
| Non-Interest income | 71,790 | 153,303 | 27,850 | 24,008 | 276,951 | | | |
| Non-Interest expense | (143,319) | (125,464) | (25,886) | (2,364) | (297,033) | | | |
| Provision for income taxes | (13,427) | (10,600) | (4,272) | (8,377) | (36,676) | | | |
| Net income, as reported | 24,937 | 19,687 | 7,934 | 43,932 | 96,490 | | | |
| Gain on sale of automobile loans, net of tax | _ | (2,216) | | (6,556) | (8,772) | | | |
| Restructuring releases, net of taxes | _ | | _ | (3,455) | (3,455) | | | |
| Operating Earnings | \$ 24,937 | \$ 17,471 | \$ 7,934 | \$ 33,921 | \$ 84,263 | | | |
| 1 0 0 | · · · · · | , . | | . , | , | | | |

| | Six Months Ended June 30, | | | | | | |
|--|---------------------------|-----------------|-----------|--------------------|----------------------------|--|--|
| Income Statements (in thousands) | Regional Banking | Dealer Sales | PFG | Treasury/ Other | Huntington Consolidated | | |
| 2004 | | | | | | | |
| Net interest income | \$ 306,145 | \$ 73,891 | \$ 22,295 | \$ 42,917 | \$ 445,248 | | |
| Provision for credit losses | 1,844 | (29,916) | (97) | (2,454) | (30,623) | | |
| Non-Interest income | 154,526 | 196,538 | 56,307 | 38,396 | 445,767 | | |
| Non-Interest expense | (294,932) | (177,137) | (58,020) | (37,718) | (567,807) | | |
| Provision for income taxes | (58,654) | (22,182) | (7,170) | 9,721 | (78,285) | | |
| Net income, as reported | 108,929 | 41,194 | 13,315 | 50.862 | 214,300 | | |
| Gain on sale of automobile loans, net of tax | | (8,214) | | (817) | (9,031) | | |
| | ¢ 100.030 | | | | ¢ 205 260 | | |
| Operating Earnings | \$ 108,929 | \$ 32,980 | \$ 13,315 | \$ 50,045 | \$ 205,269 | | |
| 2003 | | | | | | | |
| Net interest income | \$ 296,832 | \$ 27,286 | \$ 19,280 | \$ 60,802 | \$ 404,200 | | |
| Provision for credit losses | (64,078) | (20,576) | (1,443) | 60 | (86,037) | | |
| Non-Interest income | 143,389 | 311,819 | 55,063 | 39,604 | 549,875 | | |
| Non-Interest expense | (283,623) | (259,633) | (52,521) | (16,735) | (612,512) | | |
| Provision for income taxes | (32,382) | (20,614) | (7,133) | (7,177) | (67,306) | | |
| Net income, as reported | 60,138 | 38,282 | 13,246 | 76,554 | 188,220 | | |
| Gain on sale of automobile loans, net of tax | | (4,807) | | (10,631) | (15,438) | | |
| Restructuring releases, net of taxes | — | — | — | (4,105) | (4,105) | | |
| Operating Earnings | \$ 60.138 | \$ 33,475 | \$ 13,246 | \$ 61,818 | \$ 168,677 | | |
| operating Darmings | \$ 00,130 | \$ 55,475 | \$ 13,240 | \$ 51,010 | \$ 100,077 | | |

| | | Total Deposits at | | | | |
|--|------------------|-------------------|------------------|------------------|----------------------|------------------|
| Period-end Balance Sheet Data (in millions) | June 30, 2004 | December 31, 2003 | June 30, 2003 | June 30, 2004 | December 31, 2003 | June 30, 2003 |
| Regional Banking | \$ 16,526 | \$ 14,971 | \$ 14,585 | \$ 16,435 | \$ 15,539 | \$ 16,628 |
| Dealer Sales | 6,162 | 7,335 | 6,607 | 71 | 77 | 67 |
| PFG | 1,540 | 1,461 | 1,328 | 1,016 | 1,164 | 1,027 |
| Treasury / Other | 7,193 | 6,752 | 5,817 | 1,943 | 1,707 | 649 |
| Total | \$ 31,421 | \$ 30,519 | \$ 28,337 | \$ 19,465 | \$ 18,487 | \$ 18,371 |
| | • =) | | . , | . , | | |

Note 9 – Comprehensive Income

The components of Huntington's Other Comprehensive Income in the three and six months ended June 30 were as follows:

| | Three Mon June | | Six Months Ended June 30, | |
|---|-------------------|------------|------------------------------|-------------|
| (in thousands) | 2004 | 2003 | 2004 | 2003 |
| Unrealized holding (losses) gains on securities available for sale arising during the period: | | | | |
| Unrealized net (losses) gains | (136,458) | 9,046 | (74,755) | 1,799 |
| Related tax benefit (expense) | 47,760 | (3,162) | 26,399 | (935) |
| Net | (88,698) | 5,884 | (48,356) | 864 |
| Less: Reclassification adjustment for net gains (losses) included in net income: | | | | <u> </u> |
| Realized net gains (losses) | 9,230 | (6,887) | (5,860) | (8,085) |
| Related tax (expense) benefit | (3,231) | 2,410 | 2,051 | 2,830 |
| Net | 5,999 | (4,477) | (3,809) | (5,255) |
| 1101 | | | (5,005) | |
| Total unrealized holding (losses) gains on securities available for sale arising during the period, net of reclassification adjustment for net gains included in net income. | | | | |
| | (92 (00) | 1 407 | (53.1(5) | (4.201) |
| Net | (82,699) | 1,407 | (52,165) | (4,391) |
| Unrealized gains (losses) on derivatives used in cash flow hedging relationships arising during the period: | | | | |
| Unrealized net gains (losses) | 52,315 | (23,415) | 34,282 | (26,295) |
| Related tax (expense) benefit | (18,310) | 8,195 | (11,999) | 9,203 |
| Net | 34,005 | (15,220) | 22,283 | (17,092) |
| Total Other Comprehensive Loss | \$ (48,694) | \$(13,813) | \$ (29,882) | \$ (21,483) |
| | | . (. ,) | | . (,) |

Activity in Accumulated Other Comprehensive Income for the six months ended June 30, 2004 and 2003 was as follows:

| (in thousands) | Minimum pension liability | Unrealized gains (losses) on securities available for sale | Unrealized gains (losses) on derivative instruments used in cash flow hedging relationships | Total |
|----------------------------|---------------------------------|---|---|-------------|
| Balance, December 31, 2002 | \$ (195) | \$ 56,856 | \$ 5,639 | \$ 62,300 |
| Period change | | (4,391) | (17,092) | (21,483) |
| Balance, June 30, 2003 | \$ (195) | \$ 52,465 | \$ (11,453) | \$ 40,817 |
| | | | | |
| Balance, December 31, 2003 | \$ (1,309) | \$ 9,429 | \$ (5,442) | \$ 2,678 |
| Period change | — ⁽ | (52,165) | 22,283 | (29,882) |
| | | | | |
| Balance, June 30, 2004 | \$ (1,309) | \$ (42,736) | \$ 16,841 | \$ (27,204) |
| | | | | |

Note 10 - Earnings per Share

Basic earnings per share is the amount of earnings for the period available to each share of common stock outstanding during the reporting period. Diluted earnings per share is the amount of earnings available to each share of common stock outstanding during the reporting period adjusted for the potential issuance of common shares upon the exercise of stock options. The calculation of basic and diluted earnings per share for each of the three and six months ended June 30 is as follows:

| | | Three Mo Jun | nths En e 30, | ded | Six Months Ended June 30, | | | led |
|--|----------|------------------|------------------|------------------|------------------------------|------------------|----|------------------|
| (in thousands, except per share amount) | _ | 2004 | | 2003 | | 2004 | | 2003 |
| Net Income | \$ | 110,127 | \$ | 96,490 | \$ | 214,300 | \$ | 188,220 |
| Average common shares outstanding Dilutive effect of common stock equivalents | | 229,429 3,230 | | 228,633 1,939 | _ | 229,328 3,459 | | 229,987 1,697 |
| Diluted Average Common Shares Outstanding | <u> </u> | 232,659 | \$ | 230,572 | \$ | 232,787 | \$ | 231,684 |
| | . | 232,037 | φ | 250,572 | φ | 252,101 | \$ | 251,004 |
| Earnings Per Share | | | | | | | | |
| Basic | \$ | 0.48 | \$ | 0.42 | \$ | 0.93 | \$ | 0.82 |
| Diluted | \$ | 0.47 | \$ | 0.42 | \$ | 0.92 | \$ | 0.81 |

The average market price of Huntington's common stock for the period was used in determining the dilutive effect of outstanding stock options. Common stock equivalents are computed based on the number of shares subject to stock options that have an exercise price less than the average market price of Huntington's common stock for the period.

Approximately 6.8 million and 5.1 million stock options were vested and outstanding at June 30, 2004 and 2003, respectively, but were not included in the computation of diluted earnings per share because the options' exercise price was greater than the average market price of the common shares for the period and, therefore, the effect would be antidilutive. The weighted average exercise price for these options was \$22.06 per share and \$23.73 per share at the end of the same respective periods.

On July 30, 2004, Huntington entered into an agreement with the former shareholders of LeaseNet, Inc. to issue in early 2005 up to 86,118 shares of Huntington common stock previously held in escrow subject to LeaseNet meeting certain contractual performance criteria. A total of 366,576 common shares, previously held in escrow, will be returned to Huntington. All shares in escrow had been accounted for as treasury stock.

Note 11 - Stock-Based Compensation

Huntington's stock-based compensation plans are accounted for based on the intrinsic value method promulgated by APB Opinion 25, Accounting for Stock Issued to Employees, and related interpretations. Compensation expense for employee stock options is generally not recognized if the exercise price of the option equals or exceeds the fair value of the stock on the date of grant.

The following pro forma disclosures for net income and earnings per diluted common share is presented as if Huntington had applied the fair value method of accounting of Statement No. 123 in measuring compensation costs for stock options. The fair values of the stock options granted were estimated using the Black-Scholes option-pricing model. This model assumes that the estimated fair value of the options is amortized over the options' vesting periods and the compensation costs would be included in personnel expense on the income statement. The following table also includes the weighted-average assumptions that were used in the option-pricing model for options granted in each of the quarters presented:

| | | Three Months Ended June 30, | | s Ended 30, |
|--|----------|--------------------------------|----------|----------------|
| | 2004 | 2003 | 2004 | 2003 |
| Stock Options Outstanding at period end (in thousands) | 19,252 | 17,399 | 19,252 | 17,399 |
| Assumptions | | | | |
| Risk-free interest rate | 3.89% | 4.46% | 3.82% | 4.30% |
| Expected dividend yield | 3.33% | 3.26% | 3.28% | 3.30% |
| Expected volatility of Huntington's common stock | 30.9% | 33.8% | 30.9% | 33.8% |
| Pro Forma Results (in millions of dollars) | | | | |
| Net income, as reported | \$ 110.1 | \$ 96.5 | \$ 214.3 | \$ 188.2 |
| Less pro forma expense, net of tax, related to options granted | 3.1 | 2.9 | 6.4 | 5.9 |
| | | | | |
| Pro Forma Net Income | \$ 107.0 | \$ 93.6 | \$ 207.9 | \$ 182.3 |
| | | | | |
| Net Income Per Common Share: | | | | |
| Basic, as reported | \$ 0.48 | \$ 0.42 | \$ 0.93 | \$ 0.82 |
| Basic, pro forma | \$ 0.47 | \$ 0.41 | \$ 0.91 | \$ 0.79 |
| Diluted, as reported | \$ 0.47 | \$ 0.42 | \$ 0.92 | \$ 0.81 |
| Diluted, pro forma | \$ 0.46 | \$ 0.41 | \$ 0.89 | \$ 0.79 |

Note 12 – Restructuring Reserves

On a quarterly basis, Huntington assesses its remaining restructuring reserves and makes adjustments to those reserves as necessary. Huntington had remaining reserves for restructuring of \$8.3 million, \$9.7 million, and \$9.4 million as of June 30, 2004, December 31, 2003, and June 30, 2003, respectively. Huntington expects that the reserves will be adequate to fund the estimated future cash outlays.

Note 13 - Benefit Plans

Huntington sponsors the Huntington Bancshares Retirement Plan (the Plan), a non-contributory defined benefit pension plan covering substantially all employees. The Plan provides benefits based upon length of service and compensation levels. The funding policy of Huntington is to contribute an annual amount that is at least equal to the minimum funding requirements but not more than that deductible under the Internal Revenue Code. In addition, Huntington has an unfunded defined benefit post-retirement plan that provides certain healthcare and life insurance benefits to retired employees who have attained the age of 55 and have at least 10 years of vesting service under this plan. For any employee retiring on or after January 1, 1993, post-retirement healthcare benefits are based upon the employee's number of months of service and are limited to the actual cost of coverage. Life insurance benefits are a percentage of the employee's base salary at the time of retirement, with a maximum of \$50,000 of coverage.

The following table shows the components of net periodic benefit expense:

| | Three | sion Benefits Months Ended June 30, | Post Retiren Three Moi June | ths Ended | |
|--|--------------------------------|---|-----------------------------------|---------------|--|
| (in thousands) | 2004 | 2003 | 2004 | 2003 | |
| | | ÷ 0.454 | | * 2 00 | |
| Service cost | \$ 3,040 | \$ 2,454 | \$ 326 | \$ 280 | |
| Interest cost | 4,371 | 4,162 | 802 | 870 | |
| Expected return on plan assets | (5,383) | | | | |
| Amortization of transition asset | _ | (63) | 276 | 276 | |
| Amortization of prior service cost | | — | 146 | 151 | |
| Settlements | 1,000 | 1,089 | _ | | |
| Recognized net actuarial loss | 1,984 | 444 | — | | |
| | | | | | |
| Benefit Expense | \$ 5,012 | \$ 1,801 | \$ 1,550 | \$ 1,577 | |
| | Pension E Six Month June | s Ended | Post Retirem Six Month June | s Ended | |
| (in thousands) | 2004 | 2003 | 2004 | 2003 | |
| Service cost | \$ 6,078 | \$ 4,909 | \$ 650 | \$ 561 | |
| Interest cost | 8,741 | 8,323 | 1,604 | 1,739 | |
| Expected return on plan assets | (10,764) | (12,568) | _ | | |
| Amortization of transition asset | _ | (126) | 552 | 551 | |
| Amortization of prior service cost | _ | _ | 291 | 303 | |
| | | | | | |
| Settlements | 2,000 | 2,176 | — | | |
| Settlements Recognized net actuarial loss | 2,000 3,968 | 2,176 886 | _ | | |
| | | | | — | |

Huntington also sponsors other retirement plans. One of those plans is an unfunded Supplemental Executive Retirement Plan. This plan is a nonqualified plan that provides certain former officers of Huntington and its subsidiaries with defined pension benefits in excess of limits imposed by federal tax law. Other plans, including plans assumed in various past acquisitions, are unfunded, nonqualified plans that provide certain active and former officers of Huntington and its subsidiaries nominated by Huntington's compensation committee with deferred compensation, post-employment, and/or defined pension benefits in excess of the qualified plan limits imposed by federal tax law.

Huntington has a 401(k) plan, which is a defined contribution plan that is available to eligible employees. Matching contributions by Huntington equal 100% on the first 3%, then 50% on the next 2%, of participant elective deferrals. The cost of providing this plan was \$2.3 million and \$2.1 million for the three months ended June 30, 2004 and 2003, respectively. For the respective six-month periods, the cost was \$4.7 million and \$4.3 million.

Note 14 - Commitments and Contingent Liabilities

In the ordinary course of business, Huntington makes various commitments to extend credit that are not reflected in the financial statements. The contract amount of these financial agreements at June 30, 2004, December 31, 2003, and June 30, 2003, were as follows:

| (in millions) | June 30, 2004 | December 31, 2003 | June 30, 2003 |
|--|------------------|----------------------|------------------|
| Contract amount represents credit risk | | | |
| Commitments to extend credit | | | |
| Commercial | \$ 4,993 | \$ 5,712 | \$ 4,924 |
| Consumer | 3,868 | 3,652 | 3,157 |
| Commercial real estate | 586 | 952 | 891 |
| Standby letters of credit | 962 | 983 | 978 |
| Commercial letters of credit | 231 | 166 | 215 |

Commitments to extend credit:

Commitments to extend credit generally have fixed expiration dates, are variable-rate, and contain clauses that permit Huntington to terminate or otherwise renegotiate the contracts in the event of a significant deterioration in the customer's credit quality. These arrangements normally require the payment of a fee by the customer, the pricing of which is based on prevailing market conditions, credit quality, probability of funding, and other relevant factors. Since many of these commitments are expected to expire without being drawn upon, the contract amounts are not necessarily indicative of future cash requirements. The interest rate risk arising from these financial instruments is insignificant as a result of their predominantly short-term, variable-rate nature.

Standby letters of credit are conditional commitments issued to guarantee the performance of a customer to a third party. These guarantees are primarily issued to support public and private borrowing arrangements, including commercial paper, bond financing, and similar transactions. Most of these arrangements mature within two years. The carrying amount of deferred revenue associated with these guarantees was \$3.2 million, \$3.8 million, and \$3.3 million at June 30, 2004, December 31, 2003, and June 30, 2003, respectively.

Commercial letters of credit represent short-term, self-liquidating instruments that facilitate customer trade transactions and generally have maturities of no longer than 90 days. The merchandise or cargo being traded normally secures these instruments.

Litigation:

In the ordinary course of business, there are various legal proceedings pending against Huntington and its subsidiaries. In the opinion of management, the aggregate liabilities, if any, arising from such proceedings are not expected to have a material adverse effect on Huntington's consolidated financial position. (See also Note 3.)



Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

INTRODUCTION

Huntington Bancshares Incorporated (Huntington or the company) is a multi-state diversified financial holding company organized under Maryland law in 1966 and headquartered in Columbus, Ohio. Through its subsidiaries, Huntington is engaged in providing full-service commercial and consumer banking services, mortgage banking services, automobile financing, equipment leasing, investment management, trust services, and discount brokerage services, as well as reinsuring credit life and disability insurance, and selling other insurance and financial products and services. Huntington's banking offices are located in Ohio, Michigan, West Virginia, Indiana, and Kentucky. Selected financial services are also conducted in other states including Arizona, Florida, Georgia, Maryland, New Jersey, Pennsylvania, and Tennessee. Huntington has a foreign office in the Cayman Islands and a foreign office in Hong Kong. The Huntington National Bank (the Bank), organized in 1866, is Huntington's only bank subsidiary.

The following discussion and analysis provides investors and others with information that Management believes to be necessary for an understanding of Huntington's financial condition, changes in financial condition, results of operations, and cash flows, and should be read in conjunction with the financial statements, notes, and other information contained in this report.

Forward-Looking Statements

This report, including Management's Discussion and Analysis of Financial Condition and Results of Operations, contains forward-looking statements about Huntington. These include descriptions of products or services, plans or objectives of Management for future operations, including pending acquisitions, and forecasts of revenues, earnings, cash flows, or other measures of economic performance. Forward-looking statements can be identified by the fact that they do not relate strictly to historical or current facts.

By their nature, forward-looking statements are subject to numerous assumptions, risks, and uncertainties. A number of factors could cause actual conditions, events, or results to differ significantly from those described in the forward-looking statements. These factors include, but are not limited to, those set forth below and under the heading "Business Risks" included in Item 1 of Huntington's Annual Report on Form 10-K for the year ended December 31, 2003 (2003 Form 10-K), and other factors described in this report and from time to time in other filings with the Securities and Exchange Commission.

Management encourages readers of this report to understand forward-looking statements to be strategic objectives rather than absolute forecasts of future performance. Forward-looking statements speak only as of the date they are made. Huntington assumes no obligation to update forward-looking statements to reflect circumstances or events that occur after the date the forward-looking statements were made or to reflect the occurrence of unanticipated events.

Risk Factors

Huntington, like other financial companies, is subject to a number of risks, many of which are outside of Management's control, though Management strives to manage those risks while optimizing returns. Among the risks assumed are: (1) <u>credit risk</u>, which is the risk that loan and lease customers or other counter parties will be unable to perform their contractual obligations, (2) <u>market risk</u>, which is the risk that changes in market rates and prices will adversely affect Huntington's financial condition or results of operations, (3) <u>liquidity risk</u>, which is the risk that Huntington and / or the Bank will have insufficient cash or access to cash to meet operating needs, and (4<u>perational risk</u>, which is the risk of loss resulting from inadequate or failed internal processes, people, or systems, or external events. The description of Huntington's business contained in Item 1 of its 2003 Form 10-K, while not all inclusive, discusses a number of business risks that, in addition to the other information in this report, readers should carefully consider.

Securities and Exchange Commission Investigation

As previously disclosed, the Securities and Exchange Commission (SEC) is conducting a formal investigation regarding certain financial accounting and disclosure matters, including certain matters that were the subject of prior restatements by Huntington. The SEC staff has notified Huntington that the staff is considering recommending to the SEC that it institute enforcement action against Huntington and certain of its senior officers for, among other possible matters, violations of various provisions of the Securities Exchange Act of 1934 and the Securities Act of 1933 in connection with certain financial accounting matters relating to fiscal years 2002 and earlier, and certain related disclosure matters.

Huntington is presently in negotiations with the staff of the SEC regarding a settlement of its investigation. Huntington's chief executive officer has indicated he accepts responsibility in his position as the chief executive officer for matters that occur on his watch, and has indicated a willingness to negotiate a settlement with the SEC of these matters. Huntington expects that a settlement of this matter would involve the entry of an order requiring, among other possible matters, Huntington to comply with various provisions of the Securities Exchange Act of 1934 and the Securities Act of 1933, along with the imposition of a fine and other possible measures. No assurances, however, can be provided as to the ultimate timing or outcome of these matters.

Huntington also remains in active dialogue with its bank regulators concerning these matters and is working diligently with the regulators to resolve them in a manner that permits it to proceed with its pending Unizan acquisition. No assurances, however, can be provided as to the ultimate timing or outcome of these matters.

Donald R. Kimble, who joined Huntington earlier this year as executive vice president in the Finance area, has been named chief financial officer and controller. Prior to joining Huntington, Kimble was executive vice president and controller of AmSouth Bancorporation, and previously held various subsidiary chief financial officer or accounting positions at Bank One Corporation. Michael J. McMennamin and John D. Van Fleet have relinquished their positions of chief financial officer and controller, respectively. Both remain with the company.

Critical Accounting Policies and Use of Significant Estimates

Note 1 to the company's Notes to Consolidated Financial Statements included in Huntington's 2003 Form 10-K lists significant accounting policies used in the development and presentation of its financial statements. These significant accounting policies, as well as the following discussion and analysis and other financial statement disclosures, identify and address key variables and other qualitative and quantitative factors that are necessary for an understanding and evaluation of Huntington, its financial position, results of operations, and cash flows.

The preparation of financial statements in conformity with accounting principles generally accepted in the United States (GAAP) requires Huntington's management to establish critical accounting policies and make accounting estimates, assumptions, and judgments that affect amounts recorded and reported in its financial statements. An accounting estimate requires assumptions about uncertain matters that could have a material effect on the financial statements of Huntington if a different amount within a range of estimates were used or if estimates changed from period to period. Readers of this interim report should understand that estimates are made under facts and circumstances at a point in time and changes in those facts and circumstances could produce actual results that differ from when those estimates were made. Huntington's management has identified the most significant accounting estimates and their related application in Huntington's 2003 Form 10-K.

SUMMARY DISCUSSION OF RESULTS

Earnings comparisons from the second quarter of 2003 through the second quarter of 2004 were impacted by a number of factors, some related to changes in the economic and competitive environment, while others reflected specific Management strategies or changes in accounting practices. Understanding the nature and implications of these factors on financial results is important in understanding the company's income statement, balance sheet, and credit quality trends and the comparison of the current quarter and year-to-date performance with comparable prior-year periods. The key factors impacting the current reporting period comparisons are more fully described in the Significant Factors Influencing Financial Performance Comparisons section, which follows the summary of results below.

2004 Second Quarter versus 2003 Second Quarter

Huntington's 2004 second quarter earnings were \$110.1 million, or \$0.47 per common share, up 14% and 12%, respectively, from \$96.5 million and \$0.42 per common share in the year-ago quarter. This increase in earnings reflected:

- 90% decrease in provision for credit losses,
- 10% increase in fully taxable equivalent net interest income, and
- 5% decline in non-interest expenses

Partially offset by a 21% decline in non-interest income.

The return on average assets (ROA) and return on average equity (ROE), were 1.41% and 19.1%, respectively, compared with 1.38% and 18.0% in the year-ago quarter (see Table 1).

2004 Second Quarter versus 2004 First Quarter

Compared with 2004 first quarter net income of \$104.2 million, or \$0.45 per common share, 2004 second quarter earnings were up 6% and 4%, respectively. This increase in earnings primarily reflected:

- 80% decline in provision for credit losses, and
- 1% decline in non-interest expenses

Partially offset by a 4% decline in non-interest income.

The ROA and ROE were 1.41% and 19.1%, respectively, in the current quarter, compared with 1.36% and 18.4% in the 2004 first quarter (see Table 1).



Table 1 - Selected Quarterly Income Statement Data

| Table 1 - Selecieu Quarierty income Statement Data | 200 | 14 | 2003 | | 2Q04 vs. 2Q03 | | |
|--|------------------|------------------|------------------|------------------|------------------|-------------|--------------|
| (in thousands, except per share amounts) | Second | First | Fourth | Third | Second | Amount | % |
| Total interest income | \$ 324,167 | \$ 325,931 | \$ 335,097 | \$ 333,320 | \$ 317,325 | \$ 6,842 | 2.2% |
| Total interest expense | 101,604 | 103,246 | 110,782 | 112,849 | 114,884 | (13,280) | (11.6)% |
| Net interest income | 222,563 | 222,685 | 224,315 | 220,471 | 202,441 | 20,122 | 9.9% |
| Provision for credit losses | 5,027 | 25,596 | 26,341 | 51,615 | 49,193 | (44,166) | (89.8)% |
| | | | | | | | |
| Net interest income after provision for credit losses | 217,536 | 197,089 | 197,974 | 168,856 | 153,248 | 64,288 | 42.0% |
| Operating lease income | 78,706 | 88,867 | 105,307 | 117,624 | 128,574 | (49,868) | (38.8)% |
| Service charges on deposit accounts | 43,596 | 41,837 | 44,763 | 42,294 | 40,914 | 2,682 | 6.6% |
| Trust services Brokerage and insurance | 16,708 13,523 | 16,323 15,197 | 15,793 14,344 | 15,365 13,807 | 15,580 14,196 | 1,128 (673) | 7.2% |
| Mortgage banking | 23,322 | (4,296) | 9,677 | 30,193 | 7,185 | 16,137 | N.M. |
| Bank owned life insurance | 11,309 | 10,485 | 10,410 | 10,438 | 11,043 | 266 | 2.4% |
| Gain on sale of automobile loans | 4,890 | 9,004 | 16,288 | | 13,496 | (8,606) | (63.8)% |
| Gain on sale of branch office | .,0>0 | | | 13,112 | | (0,000) | N.M. |
| Other service charges and fees | 10,645 | 9,513 | 9,237 | 10,499 | 11,372 | (727) | (6.4)% |
| Securities gains (losses) | (9,230) | 15,090 | 1,280 | (4,107) | 6,887 | (16,117) | N.M. |
| Other | 24,659 | 25,619 | 19,411 | 23,543 | 27,704 | (3,045) | (11.0)% |
| Total non-interest income | 218,128 | 227,639 | 246,510 | 272,768 | 276,951 | (58 872) | (21.2)% |
| i otar non-interest income | 218,128 | 227,039 | 246,310 | 272,708 | 276,931 | (58,823) | (21.2)% |
| Personnel costs | 119,715 | 121,624 | 115,762 | 113,170 | 105,242 | 14,473 | 13.8% |
| Operating lease expense | 62,563 | 70,710 | 85,609 | 93,134 | 102,939 | (40,376) | (39.2)% |
| Outside data processing and other services | 17,563 | 18,462 | 15,957 | 17,478 | 16,104 | 1,459 | 9.1% |
| Equipment | 16,228 | 16,086 | 16,840 | 16,328 | 16,341 | (113) | (0.7)% |
| Net occupancy | 16,258 | 16,763 | 14,925 | 15,570 | 15,377 | 881 | 5.7% |
| Professional services | 7,836 | 7,299 | 12,175 | 11,116 | 9,872 | (2,036) | (20.6)% |
| Marketing | 8,069 | 7,839 | 6,895 | 5,515 | 8,454 | (385) | (4.6)% |
| Telecommunications | 4,638 | 5,194 | 5,272 | 5,612 | 5,394 | (756) | (14.0)% |
| Printing and supplies | 3,098 | 3,016 | 3,417 | 3,658 | 2,253 | 845 | 37.5% |
| Amortization of intangibles | 204 | 204 | 204 | 204 | 204 | | N.M. |
| Loss on early extinguishment of debt Restructuring reserve releases | _ | | 15,250 (351) | | (5,315) | 5,315 | N.M. N.M. |
| Other | 25,981 | 18,457 | 25,510 | 18,397 | 20,168 | 5,813 | 28.8% |
| | | | | | | | |
| Total non-interest expense | 282,153 | 285,654 | 317,465 | 300,182 | 297,033 | (14,880) | (5.0)% |
| Income before provision for income taxes | 153,511 | 139,074 | 127,019 | 141,442 | 133,166 | 20,345 | 15.3% |
| Provision for income taxes | 43,384 | 34,901 | 33,758 | 37,230 | 36,676 | 6,708 | 18.3% |
| Income before cumulative effect effect of change | | | | | | | |
| in accounting principle | 110,127 | 104,173 | 93,261 | 104,212 | 96,490 | 13,637 | 14.1% |
| Cumulative effect of change in accounting principle, net of tax ⁽¹⁾ | | | | (13,330) | | | N.M. |
| | | | | (13,330) | | | IN.IVI. |
| Net Income | \$110,127 | \$ 104,173 | \$ 93,261 | \$ 90,882 | \$ 96,490 | \$ 13,637 | 14.1% |
| Per Common Share | | | | | | | |
| Income before cumulative effect of change in | | | | | | | |
| accounting principle - Diluted | \$ 0.47 | \$ 0.45 | \$ 0.40 | \$ 0.45 | \$ 0.42 | \$ 0.05 | 11.9% |
| Net income - Diluted | 0.47 | 0.45 | 0.40 | 0.39 | 0.42 | 0.05 | 11.9% |
| Cash dividends declared | 0.175 | 0.175 | 0.175 | 0.175 | 0.16 | 0.03 | 9.4% |
| Return on: | | | | | | | |
| Average total assets ⁽²⁾ | 1.41% | 1.36% | 1.22% | 1.38% | 1.38% | 0.03% | 2.2% |
| Average total shareholders' equity ⁽²⁾ | 19.1% | 18.4% | 16.6% | 18.5% | 18.0% | 1.1% | 6.1% |
| Net interest margin ⁽³⁾ | 3.29% | 3.36% | 3.42% | 3.46% | 3.47% | (0.18)% | (5.2)% |
| Efficiency ratio (4) | 62.3% | 65.1% | 67.1% | 60.0% | 62.5% | (0.2)% | (0.3)% |
| Effective tax rate | 28.3% | 25.1% | 26.6% | 26.3% | 27.5% | 0.8% | 2.9% |
| Revenue - Fully Taxable Equivalent (FTE) | | | | | | | |
| Net interest income | \$ 222,563 | \$ 222,685 | \$ 224,315 | \$ 220,471 | \$ 202,441 | \$ 20,122 | 9.9% |
| Tax equivalent adjustment ⁽³⁾ | 2,919 | 3,023 | 2,954 | 2,558 | 2,076 | 843 | 40.6% |
| Net Interest Income | 225,482 | 225,708 | 227,269 | 223,029 | 204,517 | 20,965 | 10.3% |
| Non-Interest Income | 218,128 | 227,639 | 246,510 | 272,768 | 276,951 | (58,823) | (21.2)% |
| Total Revenue | \$ 443,610 | \$453,347 | \$473,779 | \$ 495,797 | \$481,468 | \$ (37,858) | (7.9)% |
| | | , | | , | , | | |

⁽¹⁾ Due to the adoption of FASB Interpretation No. 46 for variable interest entities.
 ⁽²⁾ Based on income before cumulative effect change in accounting principle, net of tax.
 ⁽³⁾ On a fully taxable equivalent (FTE) basis assuming a 35% tax rate.

N.M.- Not Meaningful

⁽⁴⁾ Non-interest expense less amortization of intangibles divided by the sum of FTE net interest income and non-interest income excluding securities gains (losses).

Table 2 - Selected Year to Date Income Statement Data

| | | nths Ending ne 30, | 2004 vs. 2003 | | |
|---|-----------------|-----------------------|------------------|-----------------|--|
| (in thousands, except per share amounts) | 2004 | 2003 | Amount | % | |
| Total interest income | \$ 650,098 | \$ 637,339 | \$ 12,759 | 2.0% | |
| Total interest expense | 204,850 | 233,139 | (28,289) | (12.1)% | |
| Net interest income | 445,248 | 404,200 | 41,048 | 10.2% | |
| Provision for credit losses | 30,623 | 86,037 | (55,414) | (64.4)% | |
| Net interest income after provision for credit losses | 414,625 | 318,163 | 96,462 | 30.3% | |
| Operating lease income | 167,573 | 266,767 | (99,194) | (37.2)% | |
| Service charges on deposit accounts | 85,433 | 80,783 | 4,650 | 5.8% | |
| Trust services | 33,031 | 30,491 | 2,540 | 8.3% | |
| Brokerage and insurance | 28,720 | 29,693 | (973) | (3.3)% | |
| Mortgage banking | 19,026 | 18,310 | 716 | 3.9% | |
| Bank owned life insurance | 21,794 | 22,180 | (386) | (1.7)% | |
| Gain on sale of automobile loans | 13,894 | 23,751 | (9,857) | (41.5)% | |
| Other service charges and fees | 20,158 | 21,710 | (1,552) | (7.1)% | |
| Securities gains Other | 5,860 50,278 | 8,085 48,105 | (2,225) 2,173 | (27.5)% 4.5% | |
| Total non-interest income | 445,767 | 549,875 | (104,108) | (18.9)% | |
| | | | | <u> </u> | |
| Personnel costs | 241,339 | 218,331 | 23,008 | 10.5% | |
| Operating lease expense | 133,273 | 214,527 | (81,254) | (37.9)% | |
| Outside data processing and other services | 36,025 | 32,683 | 3,342 | 10.2% | |
| Equipment | 32,314 | 32,753 | (439) | (1.3)% | |
| Net occupancy | 33,021 | 31,986 | 1,035 | 3.2% | |
| Professional services | 15,135 | 19,157 | (4,022) 828 | (21.0)% | |
| Marketing Telecommunications | 15,908 9,832 | 15,080 11,095 | (1,263) | 5.5% | |
| Printing and supplies | 6,114 | 5,934 | (1,203) | (11.4)% 3.0% | |
| Amortization of intangibles | 408 | 408 | 180 | N.M. | |
| Restructuring reserve releases | 400 | (6,315) | 6,315 | N.M. | |
| Other | 44,438 | 36,873 | 7,565 | 20.5% | |
| Total non-interest expense | 567,807 | 612,512 | (44,705) | (7.3)% | |
| Income before provision for income taxes | 292,585 | 255,526 | 37,059 | 14.5% | |
| Provision for income taxes | 78,285 | 67,306 | 10,979 | 16.3% | |
| Net Income | \$ 214,300 | \$188,220 | \$ 26,080 | 13.9% | |
| | | | | | |
| Per Common Share | | | | | |
| Net income - Diluted | \$ 0.92 | \$ 0.81 | \$ 0.11 | 13.6% | |
| Cash dividends declared | \$ 0.35 | \$ 0.32 | \$ 0.03 | 9.4% | |
| Return on: | | | | | |
| Average total assets | 1.36% | 1.35% | 0.01% | 0.8% | |
| Average total shareholders' equity | 18.4% | 17.3% | 1.1% | 6.1% | |
| Net interest margin ⁽¹⁾ Efficiency ratio ⁽²⁾ | 3.32% 63.7% | 3.52% 64.4% | (0.20)% | (5.7)% | |
| Effective tax rate | 03.7% 26.8% | 26.3% | (0.7)% 0.5% | (1.1)% 1.7% | |
| Revenue - Fully Taxable Equivalent (FTE) | 20.8% | 20.370 | 0.570 | 1./70 | |
| Net interest income | \$ 445,248 | \$404,200 | \$ 41,048 | 10.2% | |
| Tax equivalent adjustment ⁽¹⁾ | 5,942 | 4,172 | 1,770 | 42.4% | |
| Net Interest Income | 451,190 | 408,372 | 42,818 | 10.5% | |
| Non-Interest Income | 445,767 | 549,875 | (104,108) | (18.9)% | |
| Total Revenue | \$ 896,957 | \$ 958,247 | \$ (61,290) | (6.4)% | |
| | | | | | |

(1)

On a fully taxable equivalent (FTE) basis assuming a 35% tax rate. Non-interest expense less amortization of intangibles divided by the sum of FTE net interest income and non-interest income excluding securities gains. (2)

2004 First Six Months versus 2003 First Six Months

Earnings for the first six months of 2004 were \$214.3 million, or \$0.92 per common share, both up 14% from the comparable year-ago period earnings of \$188.2 million, or \$0.81 per common share. This increase in earnings reflected:

- 64% decrease in provision for credit losses,
- 10% increase in fully taxable equivalent net interest income, and
- 7% decline in non-interest expenses

Partially offset by a 19% decline in non-interest income.

The ROA and ROE were 1.36% and 18.4%, respectively, compared with 1.35% and 17.3% in the year-ago six-month period (see Table 2).

Significant Factors Influencing Financial Performance Comparisons

Earnings comparisons from the second quarter of 2003 through the second quarter of 2004 were impacted by a number of factors, some related to changes in the economic and competitive environment, while others reflected specific Management strategies or changes in accounting practices. Those key factors are summarized below.

Automobile leases originated through April 2002 accounted for as operating leases. Automobile leases originated before May 2002 are accounted for using the operating lease method of accounting because they do not qualify as direct financing leases. Operating leases are a non-interest earning asset with the related rental income, other revenue, and credit recoveries reflected as operating lease income, a component of non-interest income. Under this accounting method, depreciation expenses, as well as other costs and charge-offs, are reflected as operating lease expense, a component of non-interest expense. With no new operating leases originated since April 2002, the operating lease assets are rapidly decreasing and will eventually run-off, along with related operating lease income and expense. Since operating lease income and expense a significant percentage of total non-interest income and expense, respectively, throughout this reporting period, their downward trend influences total non-interest income and non-interest expense trends.

All automobile leases originated since April 2002 are accounted for as direct financing leases, an interest earning asset included in total loans and leases with the related income reflected as interest income and included in the calculation of the net interest margin. Credit charge-offs and recoveries are reflected in the allowance for loan and lease losses (ALLL), with related changes in the ALLL reflected in provision for credit losses. The relative newness and rapid growth of this portfolio has resulted in higher reported automobile lease growth rates than in a more mature portfolio. To better understand overall trends in automobile lease exposure it is helpful to compare trends of the combined total of automobile leases plus operating leases (see the company's 2003 Form 10-K for a full discussion).

- Transition from a weak economic environment in 2003 to a recovering economic environment in 2004. The weak economic environment resulted in significant reductions in non-performing assets (NPAs), charge-offs, and the ALLL, as well as continued weak demand for commercial and industrial (C&I) loans, which contributed to declining C&I loans through the first quarter of 2004.
- 3. Declining interest rates in 2003 with generally increasing interest rates in 2004 Interest rates impacted, among other factors, loan and deposit growth, the net interest margin, and the valuation of mortgage servicing rights (MSRs) and investment securities.
 - The historically low interest rate environment resulted in strong demand and resultant growth in residential real estate, home equity, and commercial real
 estate (CRE) loans generally throughout this period. Mortgage banking revenue was also favorably impacted by the significant mortgage origination activity.



- As interest rates fell in 2003 and attained historically low absolute levels, it became increasingly difficult to lower interest rates offered on deposit accounts commensurate with the overall decline in interest rates and yields on earning assets. This created an extremely competitive environment in which to grow deposits. These factors resulted in an inability to lower deposit rates commensurate with the overall decline in earning asset rates, which contributed to the decline in the net interest margin throughout this period.
- Since the second quarter of 2002, the company generally has retained the servicing on mortgage loans it originates and sells. The mortgage servicing right, or MSR, represents the present value of expected future net servicing income for the loan. MSR values are very sensitive to movements in interest rates. Expected future net servicing income depends on the projected outstanding principal balances of the underlying loans, which can be greatly affected by prepayments. Prepayments usually increase when interest rates decline and decrease when interest rates rise. Thus, as interest rates decline, less future income is expected and the value of MSRs declines and becomes impaired when the valuation is less than the recorded book value. The company recognizes temporary impairment due to change in interest rates through a valuation reserve and records a direct write-down of the book value of its MSRs for other-than-temporary declines in valuation. Changes in interest rate levels between quarters resulted in some quarters reporting an MSR temporary impairment, with others reporting a recovery of previously reported MSR temporary impairment. Such swings in MSR valuations have significantly impacted quarterly mortgage banking income throughout this period (see Table 3).
- The company uses gains or losses on investment securities to offset MSR temporary valuation changes. As a result, changes in interest rate levels have also resulted in securities gains or losses. As such, in quarters where an MSR temporary impairment is recognized, investment securities were sold resulting in a gain on sale, and vice versa. The earnings impact of the MSR valuation change and securities gain/loss may not exactly offset due to, among other factors, the difference in the timing of when the MSR valuation is determined and recorded, compared with when the securities are sold and any gain or loss is recorded (see Table 3).
- <u>Management strategies to lower the overall credit risk profile of the balance sheet</u>. Throughout this period, certain strategies were implemented to lower the overall credit risk profile of the balance sheet with the objective of lowering the volatility of earnings.
 - Automobile loan sales One strategy has been to lower the credit exposure to automobile loans and leases to 20% of total credit exposure, as manifested
 through the sale of automobile loans. These sales of higher-rate, higher-risk loans impact results in a number of ways including: lower growth rates in
 automobile, total consumer, and total company loans; the generation of gains reflected in non-interest income; lower net interest income than otherwise
 would be the case if the loans were not sold; and lower net interest margin (see Table 3).
 - Reduction in large-individual C&I and CRE credits This strategy has been reflected in the reduction in shared national credits, as well as other, mostly C&I, loans. In addition, the company sold and charged-off lower-quality C&I and CRE credits in 2003 and 2004. This strategy was a contributing factor in the declines in C&I loan balances, NPAs, and the ALLL. In certain quarters, this strategy contributed to higher C&I net charge-offs.
- 5. <u>Adoption of FIN 46</u> Effective July 1, 2003, the company adopted Financial Accounting Standards Board (FASB) Interpretation No. 46 (FIN 46), *Consolidation of Variable Interest Entities*. The adoption of FIN 46 resulted in the consolidation of \$1.0 billion of previously securitized automobile loans and a \$13.3 million after-tax charge in the 2003 third quarter for the cumulative effect of a change in accounting principle (see Tables 1 and 2).
- 6. <u>Corporate Restructuring Charges</u> The 2001 strategic refocusing plan included the intent to sell the Florida banking and insurance operations, credit-related and other actions to strengthen the balance sheet and financial performance, and the consolidation of numerous non-Florida banking offices. As a result, non-interest expenses in 2001 and 2002 were higher than they otherwise would have been as they included net restructuring charges based on estimated costs associated with implementing these strategic initiatives. In contrast, 2003 non-interest expense reflected recoveries of previously established reserves, which were no longer needed (see Table 3) and lowered 2003 non-interest expense (see Note 21 of the company's 2003 Form 10-K Notes to Consolidated Financial Statements). There were no increases or releases of restructuring charges in the first half of 2004.

 Single commercial recovery – A single commercial credit recovery in the 2004 second quarter on a loan previously charged off in the 2002 fourth quarter favorably impacted the 2004 second quarter provision expense (see Table 3), as well as C&I, total commercial, and total net charge-offs for the quarter (see Table 11).

The following table quantifies the earnings impact of changes in MSR and investment securities valuation, sales of automobile loans, restructuring reserve releases, and a single, large commercial credit recovery on the specified periods.

Table 3 - Significant Items Influencing Earnings Performance Comparisons

| (in millions, except per share) | Pre-tax | Impact After-tax ⁽¹⁾ | EPS |
|--|----------|------------------------------------|---------|
| Three Months Ended: | | | |
| June 30, 2004 - GAAP earnings | \$ 153.5 | \$ 110.1 | \$ 0.47 |
| Gain on sale of automobile loans | 4.9 | 3.2 | 0.01 |
| Mortgage servicing right (MSR) temporary impairment recovery | 10.4 | 6.8 | 0.03 |
| Investment securities losses | (9.2) | (6.0) | (0.03) |
| Single commercial credit recovery | 9.7 | 6.3 | 0.03 |
| June 30, 2003 - GAAP earnings | \$ 133.2 | \$ 96.5 | \$ 0.42 |
| Gain on sale of automobile loans | 13.5 | 8.8 | 0.04 |
| MSR temporary impairment | (6.4) | (4.1) | (0.02) |
| Investment securities gains | 6.9 | 4.5 | 0.02 |
| Restructuring reserve releases | 5.3 | 3.5 | 0.01 |
| Six Months Ended: | | | |
| June 30, 2004 - GAAP earnings | \$ 292.6 | \$ 214.3 | \$ 0.92 |
| Gain on sale of automobile loans | 13.9 | 9.0 | 0.04 |
| MSR temporary impairment recovery | 0.3 | 0.2 | _ |
| Investment securities gain on sale | 5.9 | 4.0 | 0.02 |
| Single commercial credit recovery | 9.7 | 6.3 | 0.03 |
| June 30, 2003 - GAAP earnings | \$ 255.5 | \$ 188.2 | \$ 0.81 |
| Gain on sale of automobile loans | 23.8 | 15.0 | 0.06 |
| MSR temporary impairment | (6.4) | (4.1) | (0.02) |
| Investment securities gains | 8.1 | 5.0 | 0.02 |

⁽¹⁾ Increase (decrease) to GAAP earnings.

Restructuring reserve releases

24

4.0

6.3

0.02

RESULTS OF OPERATIONS

Net Interest Income

2004 Second Quarter versus 2003 Second Quarter

Fully taxable equivalent net interest income increased \$21.0 million, or 10%, from the year-ago quarter, reflecting the favorable impact of a 16% increase in average earning assets, partially offset by an 18 basis point, or an effective 5%, decline in the net interest margin. The fully taxable equivalent net interest margin decreased to 3.29% from 3.47%, reflecting the continued impact of historically low interest rates and the strategic repositioning of portfolios to reduce automobile loans and increase the relative proportion of lower-rate, and lower-risk, residential real estate-related loans and investment securities.

Average total loans and leases increased \$2.5 billion, or 13%, from the 2003 second quarter due primarily to a \$2.2 billion, or 23%, increase in average consumer loans. Contributing to the consumer loan growth was a \$1.1 billion, or 58%, increase in average residential mortgages and a \$0.8 billion, or 23%, increase in average home equity loans. Average total automobile loans and leases increased \$0.3 billion, or 8%. This growth from the year-ago quarter reflected the positive impact of underlying new automobile loan originations, the 2003 third quarter consolidation of a \$1.0 billion automobile loan securitization trust, and the rapid growth in direct financing leases due to the migration from operating lease assets, which are no longer being originated. Offsetting these positive impacts was the sale of \$2.4 billion of automobile loans over this 12-month period (see Loan and Lease Composition for additional discussion and Table 10).

Average total C&I and CRE loans increased \$0.3 billion, or 3%, from the year-ago quarter, reflecting an 11% increase in small business C&I and CRE loans and a 9% increase in middle-market CRE loans. Average middle-market C&I loans were down 4% from the year-ago period and reflected both weak demand and the impact from continued strategies to specifically lower exposure to large individual commercial credits, including shared national credits.

Average investment securities increased \$1.6 billion, or 43%, from the year-ago quarter, primarily reflecting the investment of a portion of the proceeds from the automobile loan sales. The growth from the year-ago quarter primarily consisted of 10-year variable-rate securities.

Average total core deposits in the second quarter were \$16.2 billion, up \$0.8 billion, or 5%, from the year-ago quarter. This growth primarily reflected a \$1.1 billion, or 18%, increase in interest bearing demand deposits, partially offset by a \$0.4 billion, or 14%, decline in retail CDs.



Tables 4 and 5 reflect quarterly average balance sheets and rates earned and paid on interest-earning assets and interest-bearing liabilities:

Table 4 - Condensed Consolidated Quarterly Average Balance Sheets

| Table 4 - Condensed Consolidated Quarterly Average Balance Sheet: | | | | | | | | |
|---|-----------|-----------|-----------|-----------|-----------|---------------|---------|--|
| (in millions) Fully Tax Equivalent Basis | 20 | 004 | | 2003 | | 2Q04 vs. 2Q03 | | |
| | Second | First | Fourth | Third | Second | Amount | Percent | |
| Assets | | | | | | | | |
| Interest bearing deposits in banks | \$69 | \$ 79 | \$ 83 | \$ 90 | \$ 45 | \$ 24 | 53.39 | |
| Trading account securities | 28 | 16 | 11 | 11 | 22 | 6 | 27.3 | |
| Federal funds sold and securities purchased under resale agreements | 168 | 92 | 117 | 103 | 69 | 99 | N.M. | |
| Loans held for sale | 254 | 207 | 295 | 898 | 601 | (347) | (57.7) | |
| Securities: | | | | | | | | |
| Taxable | 4,861 | 4,646 | 4,093 | 3,646 | 3,385 | 1,476 | 43.6 | |
| Tax exempt | 410 | 437 | 424 | 358 | 291 | 119 | 40.9 | |
| Total Securities | 5,271 | 5,083 | 4,517 | 4,004 | 3,676 | 1,595 | 43.4 | |
| | | | | | | | | |
| Loans and Leases: | E E26 | 5 265 | 5 292 | 5 280 | 5 626 | (00) | (1.6) | |
| Commercial and industrial Real Estate | 5,536 | 5,365 | 5,382 | 5,380 | 5,626 | (90) | (1.6) | |
| Construction | 1,322 | 1,322 | 1,297 | 1,258 | 1,239 | 83 | 6.7 | |
| | , | | | | | | | |
| Commercial | 2,906 | 2,876 | 2,830 | 2,744 | 2,621 | 285 | 10.9 | |
| Consumer | | 2 0 4 1 | 2.520 | 2 50 4 | 0.000 | (102) | (17.4) | |
| Automobile loans | 2,337 | 3,041 | 3,529 | 3,594 | 2,830 | (493) | (17.4) | |
| Automobile leases | 2,139 | 1,988 | 1,802 | 1,590 | 1,306 | 833 | 63.8 | |
| Automobile loans and leases | 4,476 | 5,029 | 5,331 | 5,184 | 4,136 | 340 | 8.2 | |
| ······································ | | | | | | | | |
| Home equity | 4,142 | 3,880 | 3,678 | 3,503 | 3,359 | 783 | 23.3 | |
| Residential mortgage | 2,986 | 2,674 | 2,501 | 2,075 | 1,887 | 1,099 | 58.2 | |
| Other loans | 399 | 356 | 387 | 367 | 379 | 20 | 5.3 | |
| Total Consumer | 12,003 | 11,939 | 11,897 | 11,129 | 9,761 | 2,242 | 23.0 | |
| Total Loans and Leases | 21,767 | 21 502 | 21.406 | 20.511 | 19,247 | 2,520 | 13.1 | |
| Total Loans and Leases | 21,/0/ | 21,502 | 21,406 | 20,511 | 19,247 | 2,320 | 13.1 | |
| Allowance for loan and lease losses | 310 | 313 | 350 | 330 | 304 | 6 | 2.0 | |
| Net loans and leases | 21,457 | 21,189 | 21,056 | 20,181 | 18,943 | 2,514 | 13.3 | |
| Total earning assets | 27,557 | 26,979 | 26,429 | 25,617 | 23,660 | 3,897 | 16.5 | |
| On writing lange energy | 077 | 1.1((| 1 255 | 1.5(5 | 1.000 | (925) | (45.9) | |
| Operating lease assets | 977 | 1,166 | 1,355 | 1,565 | 1,802 | (825) | (45.8) | |
| Cash and due from banks | 772 | 740 | 766 | 747 | 735 | 37 | 5.0 | |
| Intangible assets | 216 | 217 | 217 | 218 | 218 | (2) | (0.9) | |
| All other assets | 2,101 | 2,046 | 2,005 | 2,067 | 1,988 | 113 | 5.7 | |
| Total Assets | \$ 31,313 | \$ 30,835 | \$ 30,422 | \$ 29,884 | \$ 28,099 | \$ 3,214 | 11.4% | |
| Liabilities and Shareholders' Equity | | | | | | | | |
| Core deposits | | | | | | | | |
| Non-interest bearing deposits | \$ 3,223 | \$ 3,017 | \$ 3,131 | \$ 3,218 | \$ 3,046 | \$ 177 | 5.8% | |
| Interest bearing demand deposits | 7,168 | 6,609 | 6,466 | 6,558 | 6,100 | 1,068 | 17.5 | |
| Savings deposits | 2,839 | 2,819 | 2,824 | 2,808 | 2,804 | 35 | 1.2 | |
| Retail certificates of deposit | 2,400 | 2,399 | 2,492 | 2,561 | 2,798 | (398) | (14.2) | |
| Other domestic time deposits | 600 | 637 | 631 | 656 | 673 | (73) | (10.8) | |
| Total core deposits | 16,230 | 15,481 | 15,544 | 15,801 | 15,421 | 809 | 5.2 | |
| | | | · | | | | | |
| Domestic time deposits of \$100,000 or more | 795 | 788 | 828 | 803 | 808 | (13) | (1.6) | |
| Brokered time deposits and negotiable CDs | 1,737 | 1,907 | 1,851 | 1,421 | 1,241 | 496 | 40.0 | |
| Foreign time deposits | 542 | 549 | 522 | 536 | 426 | 116 | 27.2 | |
| Total deposits | 19,304 | 18,725 | 18,745 | 18,561 | 17,896 | 1,408 | 7.9 | |
| Short-term borrowings | 1,396 | 1,603 | 1,433 | 1,393 | 1,635 | (239) | (14.6) | |
| Federal Home Loan Bank advances | 1,270 | 1,273 | 1,273 | 1,273 | 1,267 | 3 | 0.2 | |
| Subordinated notes and other long-term debt, | -, | -,,- | -,=+0 | -, | -,=07 | 2 | | |
| including preferred capital securities | 5,623 | 5,557 | 5,432 | 5,197 | 4,010 | 1,613 | 40.2 | |
| Total interest bearing liabilities | 24,370 | 24,141 | 23,752 | 23,206 | 21,762 | 2,608 | 12.0 | |
| All other liabilities | 1,397 | 1,399 | 1,311 | 1,221 | 1,140 | 257 | 22.5 | |
| an oner naonnes | 2,323 | 2,278 | 2,228 | 2,239 | 2,151 | 172 | 8.0 | |
| Shareholders' equity | -, | | _, | _, | -, | | 0.0 | |

Shareholders' equity

| Total Liabilities and Shareholders' Equity | \$ 31,313 | \$ 30,835 | \$ 30,422 | \$ 29,884 | \$ 28,099 | \$ 3,214 | 11.4% |
|--|-----------|-----------|-----------|-----------|-----------|----------|-------|
| | | | | | | | |
| | | | | | | | |

Table 5 - Condensed Consolidated Quarterly Net Interest Margin Analysis

| | Average Rates ⁽²⁾ | | | | | | |
|---|------------------------------|-------|--------|-------|--------|--|--|
| (in millions) | 2004 | L | | 2003 | | | |
| Fully Tax Equivalent Basis ⁽¹⁾ | Second | First | Fourth | Third | Second | | |
| Assets | | | | | | | |
| Interest bearing deposits in banks | 1.07% | 0.71% | 0.60% | 0.51% | 1.58% | | |
| Trading account securities | 3.02 | 3.98 | 2.39 | 4.70 | 4.15 | | |
| Federal funds sold and securities purchased under resale agreements | 1.21 | 1.41 | 1.30 | 1.92 | 2.19 | | |
| Loans held for sale | 5.17 | 5.33 | 5.31 | 5.16 | 5.42 | | |
| Securities: | | | | | | | |
| Taxable | 3.83 | 4.06 | 4.24 | 4.23 | 4.58 | | |
| Tax exempt | 7.07 | 6.88 | 6.91 | 6.93 | 6.91 | | |
| * | | | | | | | |
| Total Securities | 4.09 | 4.30 | 4.49 | 4.47 | 4.77 | | |
| | | | | | | | |
| Loans and Leases: ⁽²⁾ | | | | | | | |
| Commercial and industrial | 4.25 | 4.49 | 4.82 | 4.84 | 5.26 | | |
| Real Estate | | | | | | | |
| Construction | 3.70 | 3.68 | 4.24 | 4.17 | 4.07 | | |
| Commercial | 4.57 | 4.70 | 4.99 | 5.22 | 5.28 | | |
| Consumer | | | | | | | |
| Automobile loans | 7.20 | 6.93 | 6.90 | 7.19 | 7.74 | | |
| Automobile leases | 5.06 | 4.94 | 4.98 | 4.99 | 4.69 | | |
| | | | | | | | |
| Automobile loans and leases | 6.17 | 6.14 | 6.25 | 6.51 | 6.78 | | |
| Home equity | 4.92 | 4.82 | 4.87 | 5.09 | 5.02 | | |
| Residential mortgage | 5.30 | 5.44 | 5.20 | 5.32 | 5.76 | | |
| Other loans | 7.48 | 7.24 | 7.19 | 7.38 | 7.22 | | |
| | 7:40 | 7.24 | 7.17 | 7.50 | 1.22 | | |
| Total Consumer | 5.49 | 5.52 | 5.64 | 5.87 | 5.99 | | |
| | 3.47 | 5.52 | 5.04 | 5.87 | 5.99 | | |
| Total Loans and Leases | 4.95 | 5.04 | 5.26 | 5.41 | 5.56 | | |
| | 4.76% | 4.89% | 5.11% | 5.23% | 5.42% | | |

| Elabilities and Shareholders' Equity | | | | | |
|---|-------|-------|-------|-------|-------|
| Core deposits | | | | | |
| Non-interest bearing deposits | | | | | |
| Interest bearing demand deposits | 0.81% | 0.88% | 0.91% | 1.04% | 1.39% |
| Savings deposits | 0.82 | 0.94 | 1.22 | 1.35 | 1.55 |
| Retail certificates of deposit | 3.27 | 3.47 | 3.54 | 3.51 | 3.75 |
| Other domestic time deposits | 3.19 | 3.48 | 3.69 | 3.89 | 3.85 |
| | | | | | |
| Total core deposits | 1.45 | 1.53 | 1.65 | 1.76 | 2.09 |
| | | | | | |
| Domestic time deposits of \$100,000 or more | 2.37 | 2.14 | 2.37 | 2.32 | 2.55 |
| Brokered time deposits and negotiable CDs | 1.57 | 1.51 | 1.52 | 1.63 | 1.79 |
| Foreign time deposits | 0.76 | 0.72 | 0.75 | 0.85 | 1.03 |
| | | | | | |
| Total deposits | 1.48 | 1.53 | 1.64 | 1.75 | 2.06 |
| | | | | | |
| Short-term borrowings | 0.80 | 0.83 | 0.78 | 0.85 | 1.06 |
| Federal Home Loan Bank advances | 2.52 | 2.50 | 2.24 | 1.81 | 1.76 |
| Subordinated notes and other long-term debt, including preferred capital securities | 2.24 | 2.33 | 2.63 | 2.78 | 2.85 |
| | | | | | |
| Total interest bearing liabilities | 1.66% | 1.71% | 1.85% | 1.93% | 2.11% |
| | | | | | |
| Net interest rate spread | 3.10% | 3.18% | 3.26% | 3.30% | 3.31% |
| Impact of non-interest bearing funds on margin | 0.19 | 0.18 | 0.16 | 0.16 | 0.16 |
| | | | | | |
| Net Interest Margin | 3.29% | 3.36% | 3.42% | 3.46% | 3.47% |
| ů – – – – – – – – – – – – – – – – – – – | | | | | |
| | | | | | |

Fully tax equivalent (FTE) yields are calculated assuming a 35% tax rate. See page 20 for the fully taxable equivalent adjustment. Loan and lease and deposit average rates include impact of applicable derivatives and non-deferrable fees. (1)

(2)

2004 Second Quarter versus 2004 First Quarter

Compared with the 2004 first quarter, fully taxable equivalent net interest income decreased only \$0.2 million, reflecting the adverse impact of automobile loan sales. The fully taxable equivalent net interest margin declined 7 basis points, or an effective 2%, to 3.29% from 3.36%. Average earnings assets, despite the negative impact from the sale of automobile loans, increased \$0.6 billion, or 2%. The decline in the net interest margin from the first quarter reflected the same factors as those impacting the decrease from the year-ago quarter.

Average total loans and leases in the second quarter increased \$0.3 billion, or 1%, from the first quarter, as the growth rate was mitigated by a \$0.7 billion, or 23%, decline in average automobile loans due to the first quarter sale of \$868 million of automobile loans. Growth in mortgage-related consumer loans remained strong with average residential mortgages up \$0.3 billion, or 12%, and average home equity loans up \$0.3 billion, or 7%. Total average C&I and CRE loans increased \$0.2 billion, or 2%, reflecting a 2% increase in small business C&I and CRE loans, a 3% increase in middle-market C&I loans, and a 1% increase in middle-market CRE loans.

Average investment securities increased \$0.2 billion, or 4%, from the first quarter, primarily reflecting the investment of a portion of the proceeds from the first quarter automobile loan sale. However, investment securities at June 30, 2004, were down \$0.5 billion, or 9%, from the end of the first quarter. Average loans held for sale decreased \$0.3 billion, or 58%, from the year-ago quarter.

Average total core deposits in the second quarter increased \$0.7 billion, or 5%, reflecting strong growth in interest bearing demand deposits, up \$0.6 billion, or 8%, as well as non-interest bearing deposits, up \$0.2 billion, or 7%.

2004 First Six Months versus 2003 First Six Months

Fully taxable equivalent net interest income for the first six months of 2004 increased \$42.8 million, or 10%, from the comparable year-ago period, reflecting the favorable impact of a 17% increase in average earning assets, partially offset by an 20 basis point, or an effective 6%, decline in the net interest margin. The fully taxable equivalent net interest margin decreased to 3.32% from 3.52%, reflecting the impact of lower rates and the strategic repositioning of portfolios to reduce automobile loans and increase the relative proportion of lower-rate, and lower-risk, residential real estate-related loans and investment securities.

Average total loans and leases increased \$2.6 billion, or 13%, from the first six months of 2003 due primarily to a \$2.3 billion, or 24%, increase in average consumer loans. Contributing to the consumer loan growth was a \$1.0 billion, or 52%, increase in average residential mortgages and a \$0.7 billion, or 22%, increase in average home equity loans. Average total automobile loans and leases increased \$0.6 billion, or 16%. This growth from the year-ago six-month period reflected the positive impact of underlying new automobile loan originations, the 2003 third quarter consolidation of a \$1.0 billion automobile loan securitization trust, and the rapid growth in direct financing leases due to the migration from operating lease assets, which are no longer being originated. Partially offsetting these positive impacts was the sale of automobile loans over this period.

Average total C&I and CRE loans in the first six months of 2004 increased \$0.2 billion, or 2%, from the comparable year-ago period, reflecting an 11% increase in small business C&I and CRE loans and a 10% increase in middle-market CRE loans. Average middle-market C&I loans were down 5% from the year-ago period and reflected both weak demand and the impact from continued strategies to specifically lower exposure to large individual commercial credits, including shared national credits.

Average investment securities increased \$1.7 billion, or 48%, from the year-ago six-month period, primarily reflecting the investment of a portion of the proceeds from the automobile loan sales.

Average total core deposits in the first six months of 2004 were \$15.9 billion, up \$0.7 billion, or 4%, from the year-ago period. This growth primarily reflected a \$1.0 billion, or 17%, increase in interest bearing demand deposits, partially offset by a \$0.5 billion, or 17%, decline in retail CDs.



Table 6 reflects average balance sheets and rates earned and paid on interest-earning assets and interest-bearing liabilities, respectively for the first six-month periods of 2004 and 2003:

Table 6 - Condensed Consolidated YTD Average Balance Sheets and Net Interest Margin Analysis

| | | | 2004 vs. | 2003 | YTD Average Rates ⁽³⁾ | | |
|--|-----------|-----------|----------|--------|----------------------------------|-------|--|
| Fully Tax Equivalent Basis | 2004 | 2003 | Amount | % | 2004 | 2003 | |
| Assets | | | | | | | |
| Interest bearing deposits in banks | \$ 74 | \$ 41 | \$ 33 | 80.5 | 0.88% | 1.58% | |
| Trading account securities | 22 | 17 | 5 | 29.4 | 3.36 | 4.40 | |
| Federal funds sold and securities purchased under resale agreements | 130 | 63 | 67 | N.M. | 1.28 | 2.15 | |
| Mortgages held for sale | 231 | 531 | (300) | (56.5) | 5.25 | 5.47 | |
| Securities: | | | | | | | |
| Taxable | 4,753 | 3,202 | 1,551 | 48.4 | 3.94 | 4.85 | |
| Tax exempt | 423 | 290 | 133 | 45.9 | 6.97 | 6.88 | |
| Total Securities | 5,176 | 3,492 | 1,684 | 48.2 | 4.19 | 5.02 | |
| Loans and Leases: | | | | | | | |
| Commercial and industrial | 5,451 | 5,625 | (174) | (3.1) | 4.37 | 5.28 | |
| Real Estate | -, | -, | (-, -) | (())) | | | |
| Construction | 1,322 | 1,214 | 108 | 8.9 | 3.69 | 4.06 | |
| Commercial | 2,891 | 2,593 | 298 | 11.5 | 4.63 | 5.35 | |
| Consumer | , | , | | | | | |
| Automobile loans | 2,689 | 2,954 | (265) | (9.0) | 7.05 | 7.73 | |
| Automobile leases | 2,064 | 1,153 | 911 | 79.0 | 5.02 | 5.29 | |
| | | | | | | | |
| Automobile loans and leases | 4,753 | 4,107 | 646 | 15.7 | 6.17 | 7.05 | |
| TT '/ | 4.011 | 2 200 | 712 | 21.6 | 4.00 | 5.00 | |
| Home equity | 4,011 | 3,298 | 713 | 21.6 | 4.88 | 5.08 | |
| Residential mortgage | 2,830 | 1,859 | 971 | 52.2 | 5.37 | 5.80 | |
| Other loans | 377 | 383 | (6) | (1.6) | 7.37 | 6.85 | |
| Total Consumer | 11,971 | 9,647 | 2,324 | 24.1 | 5.51 | 6.13 | |
| Total Loans and Leases | 21,635 | 19,079 | 2,556 | 13.4 | 5.00 | 5.64 | |
| Allowance for loan and lease losses | 328 | 343 | (15) | (4.4) | | · | |
| Net loans and leases | 21,307 | 18,736 | 2,571 | 13.7 | | | |
| | · | | , | | 4.020/ | | |
| Total earning assets | 27,268 | 23,223 | 4,045 | 17.4 | 4.83% | 5.52% | |
| Operating lease assets | 1,070 | 1,937 | (867) | (44.8) | | | |
| Cash and due from banks | 756 | 738 | 18 | 2.4 | | | |
| Intangible assets | 217 | 218 | (1) | (0.5) | | | |
| All other assets | 2,075 | 1,974 | 101 | 5.1 | | | |
| Total Assets | \$ 31,058 | \$ 27,747 | \$ 3,311 | 11.9 | | | |
| Liabilities and Shareholders' Equity | | | | | | | |
| Core deposits | | | | | | | |
| Non-interest bearing deposits | \$ 3,120 | \$ 2,984 | \$ 136 | 4.6 | | | |
| Interest bearing demand deposits | 6,889 | 5,868 | 1,021 | 17.4 | 0.92% | 1.44% | |
| Savings deposits | 2,829 | 2,788 | 41 | 1.5 | 0.88 | 1.63 | |
| Retail certificates of deposit | 2,400 | 2,880 | (480) | (16.7) | 3.37 | 3.81 | |
| Other domestic time deposits | 618 | 678 | (60) | (8.8) | 3.34 | 3.92 | |
| Total core deposits | 15,856 | 15,198 | 658 | 4.3 | 1.49 | 2.18 | |
| Domestic time deposits of \$100,000 or more | 792 | 789 | 3 | 0.4 | 2.26 | 2.66 | |
| Brokered time deposits of a roo,000 of hore | 1,822 | 1,198 | 624 | 52.1 | 1.54 | 1.88 | |
| Foreign time deposits | | | | | | | |
| | 545 | 470 | 75 | 16.0 | 0.74 | 1.05 | |
| Total deposits | 19,015 | 17,655 | 1,360 | 7.7 | 1.51 | 2.15 | |
| Short-term borrowings | 1,499 | 1,790 | (291) | (16.3) | 0.82 | 1.11 | |
| Federal Home Loan Bank advances | 1,272 | 1,242 | 30 | 2.4 | 2.51 | 1.80 | |
| Subordinated notes and other long-term debt, including preferred capital securities | 5,590 | 3,791 | 1,799 | 47.5 | 2.28 | 2.97 | |
| | · | | | | | | |
| Total interest bearing liabilities | 24,256 | 21,494 | 2,762 | 12.9 | 1.68% | 2.19% | |
| | 1,381 | 1,111 | 270 | 24.3 | | | |

| Shareholders' equity | 2,301 | 2,158 | 143 | 6.6 | | |
|--|-----------|-----------|----------|------|-------|-------|
| | | | | | | |
| Total Liabilities and Shareholders' Equity | \$ 31,058 | \$ 27,747 | \$ 3,311 | 11.9 | | |
| | | | | | | |
| Net interest rate spread | | | | | 3.15% | 3.33% |
| Impact of non-interest bearing funds on margin | | | | | 0.17 | 0.19 |
| | | | | | | |
| Net Interest Margin | | | | | 3.32% | 3.52% |
| | | | | | | |

Provision for Credit Losses

The provision for credit losses is the expense necessary to maintain the ALLL and the allowance for unfunded loan commitments (AULC) at levels adequate to absorb Management's estimate of inherent losses in the total loan and lease portfolio, unfunded loan commitments, and letters of credit. Taken into consideration are such factors as current period net charge-offs that are charged against these allowances, current period loan and lease growth and any related estimate of likely losses associated with that growth based on historical experience, the current economic outlook, and the anticipated impact on credit quality of existing loans and leases and unfunded commitments and letters of credit (see Allowances for Credit Losses for additional discussion and Table 14).

The provision for credit losses in the 2004 second quarter was \$5.0 million, a \$44.2 million reduction from the year-ago quarter and a \$20.6 million decline from the 2004 first quarter. This reduction in provision expense reflected the \$9.7 million commercial loan recovery and overall improved portfolio quality performance, as well as an improved economic outlook, only partially offset by provision expense related to loan growth. As previously disclosed, effective January 1, 2004, the company adopted a more quantitative approach to calculating the economic reserve component of the ALLL, making this component more responsive to changes in economic conditions. This change, combined with the existing quantitative approach for determining the transaction reserve component, as well as changes to the specific reserve component, will result in more volatility in the total ALLL and corresponding provision for credit losses (see Credit Risk for additional discussion).

The provision for credit losses in the first six months of 2004 was \$30.6 million, a \$55.4 million, or 64%, decline from the comparable year-ago period. This reduction reflected the same factors impacting second quarter year-over-year performance.

| 2 | 1 | |
|-----|---|--|
| - 1 | ι | |
| | | |

Non-Interest Income

Table 7 reflects non-interest income detail for each of the past five quarters, and the first six-months of 2004 and 2003:

Table 7 - Non-Interest Income

| | 20 | 04 | | 2003 | | 2Q04 vs. 2Q03 | |
|---|------------|------------|------------|-----------|------------|---------------|----------|
| (in thousands) | Second | First | Fourth | Third | Second | Amount | % |
| Service charges on deposit accounts | \$ 43,596 | \$ 41,837 | \$ 44,763 | \$ 42,294 | \$ 40,914 | \$ 2,682 | 6.6% |
| Trust services | 16,708 | 16,323 | 15,793 | 15,365 | 15,580 | 1,128 | 7.2 |
| Brokerage and insurance | 13,523 | 15,197 | 14,344 | 13,807 | 14,196 | (673) | (4.7) |
| Mortgage banking | 23,322 | (4,296) | 9,677 | 30,193 | 7,185 | 16,137 | N.M. |
| Bank owned life insurance | 11,309 | 10,485 | 10,410 | 10,438 | 11,043 | 266 | 2.4 |
| Gain on sale of automobile loans | 4,890 | 9,004 | 16,288 | _ | 13,496 | (8,606) | (63.8) |
| Gain on sale of branch offices | _ | _ | — | 13,112 | _ | — | — |
| Other service charges and fees | 10,645 | 9,513 | 9,237 | 10,499 | 11,372 | (727) | (6.4) |
| Securities gains (losses) | (9,230) | 15,090 | 1,280 | (4,107) | 6,887 | (16,117) | N.M. |
| Other | 24,659 | 25,619 | 19,411 | 23,543 | 27,704 | (3,045) | (11.0) |
| | | | | | · | | <u> </u> |
| Sub-total before operating lease income | 139,422 | 138,772 | 141,203 | 155,144 | 148,377 | (8,955) | (6.0) |
| Operating lease income | 78,706 | 88,867 | 105,307 | 117,624 | 128,574 | (49,868) | (38.8) |
| | | | | | | | |
| Total Non-Interest Income | \$ 218,128 | \$ 227,639 | \$ 246,510 | \$272,768 | \$ 276,951 | \$ (58,823) | (21.2)% |
| | | | | | | | |

| (in thousands) | Six | Six Months Ending June 30, | | | YTD 2004 vs. 2003 | | |
|---|---------|----------------------------|-----------|------|-------------------|----------|--|
| | 2004 | | 2003 | | Amount | | |
| Service charges on deposit accounts | \$ 85,4 | 133 § | 80,783 | \$ | 4,650 | 5.8% | |
| Trust services | 33, |)31 | 30,491 | | 2,540 | 8.3 | |
| Brokerage and insurance | 28, | /20 | 29,693 | | (973) | (3.3) | |
| Mortgage banking | 19, |)26 | 18,310 | | 716 | 3.9 | |
| Bank owned life insurance | 21, | 794 | 22,180 | | (386) | (1.7) | |
| Gain on sale of automobile loans | 13, | 894 | 23,751 | | (9,857) | (41.5) | |
| Other service charges and fees | 20, | 58 | 21,710 | | (1,552) | (7.1) | |
| Securities gains (losses) | 5, | 360 | 8,085 | | (2,225) | (27.5) | |
| Other | 50,2 | 278 | 48,105 | | 2,173 | 4.5 | |
| | | | | | | | |
| Sub-total before operating lease income | 278, | 194 | 283,108 | | (4,914) | (1.7) | |
| Operating lease income | 167, | 573 | 266,767 | (| 99,194) | (37.2) | |
| | | | , | | | <u> </u> | |
| Total Non-Interest Income | \$ 445, | 767 \$ | 5 549,875 | \$(1 | 04,108) | (18.9)% | |
| | | _ | | | | | |

N.M. - Not Meaningful.

2004 Second Quarter versus 2003 Second Quarter

Non-interest income decreased \$58.8 million, or 21%, from the year-ago quarter. Comparisons with prior-period results are heavily influenced by the decline in operating leases and related operating lease income. These trends are expected to continue as all automobile leases originated since April 2002 are direct financing leases with income reflected in net interest income, not non-interest income. Reflecting the run-off of the operating lease portfolio, operating lease income declined \$49.9 million, or 39%, from the 2003 second quarter. Excluding operating lease income, non-interest income decreased \$9.0 million, or 6%, from the year-ago quarter with the primary drivers being:

- \$16.1 million decline in investment securities gains with current quarter securities losses of \$9.2 million, including \$10.2 million of losses related to investment securities used to offset MSR temporary valuation changes.
- \$8.6 million, or 64%, decrease in gains on the sale of automobile loans, as the current quarter included \$4.9 million of pre-tax gains, compared with \$13.5 million of such pre-tax gains in the year-ago quarter. The higher relative gain in the year-ago quarter reflected the higher average rate of the loans sold in that quarter.
 - \$3.0 million, or 11%, decline in other income due to lower fees collected at the end of leases, partially offset by higher letter of credit fees.

Partially offset by:

- \$16.1 million increase in mortgage banking income. This reflected a \$16.8 million change in MSR temporary impairment valuations, as the current quarter included a
 \$10.4 million recovery of previously recorded MSR temporary impairment compared with
 \$6.4 million of MSR temporary impairment recognized in the year-ago
 quarter. The valuation changes included
 \$4.5 million of MSR hedge losses in the current quarter.
- \$2.7 million, or 7%, increase in service charges on deposit accounts.
- \$1.1 million, or 7%, increase in trust services income.

2004 Second Quarter versus 2004 First Quarter

Compared with the 2004 first quarter, non-interest income declined \$9.5 million, or 4%. This comparison is also heavily influenced by the decline in operating lease income for the reasons noted above. Reflecting the run-off of the operating lease portfolio, operating lease income declined \$10.2 million, or 11%, from the 2004 first quarter. Excluding operating lease income, non-interest income increased \$0.7 million from the 2004 first quarter with the primary drivers being:

- \$27.6 million increase in mortgage banking income. This reflected a \$20.5 million change in MSR temporary impairment valuations, as the current quarter included a \$10.4 million recovery of previously recorded MSR temporary impairment compared with \$10.1 million of MSR temporary impairment recognized in the first quarter. This reversal in MSR temporary impairment valuation between quarters reflected an upward movement in mortgage interest rates in the second quarter. The MSR temporary impairment valuation reserve at June 30, 2004, was \$1.4 million. Reflecting the rise of interest rates during the quarter, the value of MSRs as a percent of mortgage serviced for others was 1.21%, up from 0.93% at March 31, 2004. Excluding the MSR temporary impairment valuation change between quarters, mortgage banking income increased \$7.1 million, primarily reflecting higher mortgage originations and sales and including \$4.5 million of MSR hedge losses in the current quarter.
- \$1.8 million, or 4%, increase in service charges on deposit accounts.
- \$1.1 million, or 12%, increase in other service charges and fees.

Partially offset by:

- \$24.3 million decline in securities gains (losses), with the current quarter reflecting \$9.2 million in securities losses, compared with \$15.1 million of securities gains in the 2004 first quarter. Investment securities are used as an on balance sheet hedge to MSR temporary valuation changes.
- \$4.1 million decrease in gains on the sale of automobile loans as the current quarter reflected \$4.9 million of such gains, compared with \$9.0 million of gains in the
 first quarter.
- \$1.7 million, or 11%, decline in brokerage and insurance income primarily due to lower annuity sales.

2004 First Six Months versus 2003 First Six Months

Non-interest income for the first six months of 2004 declined \$104.1 million, or 19%, from the comparable year-ago period. Comparisons with prior-period results are heavily influenced by the decline in operating leases and related operating lease income (see above discussion). Reflecting the run-off of the operating lease portfolio, operating lease income for the first six months of 2004 declined \$99.2 million, or 37%, from the comparable year-ago period. Excluding operating lease income, non-interest income for the first six months of 2004 declined \$4.9 million, or 2%, from the year-ago period with the primary drivers being:

- \$9.9 million, or 42%, decrease in gains on the sale of automobile loans, as the current six-month period included \$13.9 million of pre-tax gains, compared with \$23.8 million of such pre-tax gains in the year-ago period.
- \$2.2 million decline in investment securities gains with current six-month period securities gains of \$5.9 million, compared with \$8.1 million of gains in the year-ago six-month period.

Partially offset by:

- \$4.7 million, or 6%, increase in service charges on deposit accounts, primarily reflecting higher NSF and overdraft fees.
- \$2.5 million, or 8%, increase in trust services income.
- \$2.2 million, or 4%, increase in other income.

Non-Interest Expense

Table 8 reflects non-interest expense detail for each of the last five quarters and the first six-month period for 2004 and 2003:

| Table 8 - Non-Interest Expense | | | | | | | | |
|--|------------|------------|------------|------------|------------|-------------|---------------|--|
| | 20 | 2004 | | 2003 | | | 2Q04 vs. 2Q03 | |
| (in thousands) | Second | First | Fourth | Third | Second | Amount | % | |
| Personnel costs | \$ 119,715 | \$ 121,624 | \$ 115,762 | \$ 113,170 | \$105,242 | \$ 14,473 | 13.8% | |
| | 17,563 | | 15,957 | 17,478 | 16,104 | 1,459 | 9.1 | |
| Outside data processing and other services | , | 18,462 | , | , | | , | | |
| Equipment | 16,228 | 16,086 | 16,840 | 16,328 | 16,341 | (113) | (0.7) | |
| Net occupancy | 16,258 | 16,763 | 14,925 | 15,570 | 15,377 | 881 | 5.7 | |
| Professional services | 7,836 | 7,299 | 12,175 | 11,116 | 9,872 | (2,036) | (20.6) | |
| Marketing | 8,069 | 7,839 | 6,895 | 5,515 | 8,454 | (385) | (4.6) | |
| Telecommunications | 4,638 | 5,194 | 5,272 | 5,612 | 5,394 | (756) | (14.0) | |
| Printing and supplies | 3,098 | 3,016 | 3,417 | 3,658 | 2,253 | 845 | 37.5 | |
| Amortization of intangible assets | 204 | 204 | 204 | 204 | 204 | _ | | |
| Loss on early extinguishment of debt | _ | _ | 15,250 | _ | | | _ | |
| Restructuring reserve releases | _ | _ | (351) | _ | (5,315) | 5,315 | N.M. | |
| Other | 25,981 | 18,457 | 25,510 | 18,397 | 20,168 | 5,813 | 28.8 | |
| | 210 500 | 014.044 | 001.056 | 207.049 | 104.004 | 25.406 | 12.1 | |
| Sub-total before operating lease expense | 219,590 | 214,944 | 231,856 | 207,048 | 194,094 | 25,496 | 13.1 | |
| Operating lease expense | 62,563 | 70,710 | 85,609 | 93,134 | 102,939 | (40,376) | (39.2) | |
| Total Non-Interest Expense | \$ 282,153 | \$ 285,654 | \$ 317,465 | \$ 300,182 | \$ 297,033 | \$ (14,880) | (5.0)% | |
| | | | | | | | | |

| | Six Months Er | YTD 2004 vs. 2003 | | | |
|--|---------------|-------------------|------------|--------|--|
| (in thousands) | 2004 | 2003 | Amount | % | |
| | | | | | |
| Personnel costs | \$ 241,339 | \$ 218,331 | \$ 23,008 | 10.5% | |
| Outside data processing and other services | 36,025 | 32,683 | 3,342 | 10.2 | |
| Equipment | 32,314 | 32,753 | (439) | (1.3) | |
| Net occupancy | 33,021 | 31,986 | 1,035 | 3.2 | |
| Professional services | 15,135 | 19,157 | (4,022) | (21.0) | |
| Marketing | 15,908 | 15,080 | 828 | 5.5 | |
| Telecommunications | 9,832 | 11,095 | (1,263) | (11.4) | |
| Printing and supplies | 6,114 | 5,934 | 180 | 3.0 | |
| Amortization of intangible assets | 408 | 408 | _ | | |
| Restructuring reserve releases | _ | (6,315) | 6,315 | N.M. | |
| Other | 44,438 | 36,873 | 7,565 | 20.5 | |
| | | | | | |
| Sub-total before operating lease expense | 434,534 | 397,985 | 36,549 | 9.2 | |
| Operating lease expense | 133,273 | 214,527 | (81,254) | (37.9) | |
| | | | | | |
| Total Non-Interest Expense | \$ 567,807 | \$ 612,512 | \$(44,705) | (7.3)% | |

N.M.- Not Meaningful.

Non-interest expense decreased \$14.9 million, or 5%, from the year-ago quarter. Comparisons with prior-period results are influenced by the decline in operating lease expense as the operating lease portfolio continues to run-off (see above operating lease income discussion). Operating lease expense declined \$40.4 million, or 39%, from the 2003 second quarter. Excluding operating lease expense, non-interest expense increased \$25.5 million, or 13%, from the year-ago quarter with the primary drivers being:

\$14.5 million, or 14%, increase in personnel costs, primarily reflecting a \$4.3 million increase in pension costs and a \$4.2 million increase in health care expense.

- \$5.8 million, or 29%, increase in other expense, primarily attributable to costs recorded in the current quarter related to investments in partnerships generating tax benefits, of which \$3.2 million related to previous periods.
- \$5.3 million in restructuring reserve releases that lowered expenses in the year-ago quarter.

²⁰⁰⁴ Second Quarter versus 2003 Second Quarter
• \$1.5 million, or 9%, increase in outside data processing and other services.

Partially offset by:

\$2.0 million, or 21%, decline in professional services expense.

2004 Second Quarter versus 2004 First Quarter

Compared with the 2004 first quarter, non-interest expense declined \$3.5 million, or 1%. Comparisons with prior-period results are also heavily influenced by the decline in operating lease expense. Operating lease expense declined \$8.1 million, or 12%, from the 2004 first quarter. Excluding operating lease expense, non-interest expense increased \$4.6 million, or 2%, from the first quarter with the primary drivers being:

\$7.5 million increase in other expense primarily related to \$5.8 million of costs consisting of investments in partnerships generating tax benefits.

Partially offset by:

- \$1.9 million, or 2%, decrease in personnel costs, primarily reflecting lower employment taxes.
- \$0.9 million, or 5%, decline in outside data processing and other services.

2004 First Six Months versus 2003 First Six Months

Non-interest expense for the first six months of 2004 declined \$44.7 million, or 7%, from the comparable year-ago period. Comparisons with prior-period results are influenced by the decline in operating lease expense as the operating lease portfolio continues to run-off (see above operating lease income discussion). Operating lease expense declined \$81.3 million, or 38%, from the 2003 six-month period. Excluding operating lease expense, non-interest expense for the first six months of 2004 increased \$36.5 million, or 9%, from the year-ago period with the primary drivers being:

- \$23.0 million, or 11%, increase in personnel costs, primarily reflecting a \$14.7 million, or 37%, increase in benefits expense and an \$8.8 million, or 6%, increase in salaries.
- \$7.6 million, or 21%, increase in other expense, primarily attributable to \$5.8 million of costs related to investments in partnerships generating tax benefits in the current six-month period.
- \$6.3 million in restructuring reserve releases that lowered expenses in the year-ago six-month period.
- \$3.3 million, or 10%, increase in outside data processing and other services.

Partially offset by:

\$4.0 million, or 21%, decline in professional services expense.

Operating Lease Assets

Table 9 reflects operating lease assets performance detail for each of the last five quarters, and the first six-month period for 2004 and 2003:

Table 9 - Operating Lease Assets Performance

| | 200 | 4 | | 2003 | 2003 2Q0 | | Q03 |
|---|-----------|-----------|-----------|-----------|------------|-------------|---------|
| | Second | First | Fourth | Third | Second | Amount | % |
| Balance Sheet (in millions) | | | | | | | |
| Average operating lease assets outstanding | \$ 977 | \$ 1,166 | \$ 1,355 | \$ 1,565 | \$ 1,802 | \$ (825) | (45.8)% |
| Income Statement (in thousands) | | | | | | | |
| Net rental income | 72,402 | \$ 83,517 | \$ 98,223 | \$109,645 | \$ 120,502 | \$ (48,100) | (39.9)% |
| Fees | 4,838 | 3,543 | 5,204 | 5,372 | 5,414 | (576) | (10.6) |
| Recoveries - early terminations | 1,466 | 1,807 | 1,880 | 2,607 | 2,658 | (1,192) | (44.8) |
| Total Operating Lease Income | 78,706 | 88.867 | 105,307 | 117,624 | 128,574 | (49,868) | (38.8) |
| Total Operating Lease Income | 78,700 | 88,807 | 105,507 | 117,024 | 120,574 | (4),808) | (30.0) |
| Depreciation and residual losses at termination | 57,412 | 63,932 | 76,768 | 83,112 | 91,387 | (33,975) | (37.2) |
| Losses - early termination | 5,151 | 6,778 | 8,841 | 10,022 | 11,552 | (6,401) | (55.4) |
| Total Operating Lease Expense | 62,563 | 70,710 | 85,609 | 93,134 | 102,939 | (40,376) | (39.2) |
| | | | | | | | |
| Net Earnings Contribution | \$ 16,143 | \$ 18,157 | \$ 19,698 | \$ 24,490 | \$ 25,635 | \$ (9,492) | (37.0)% |
| Earnings ratios (1) | | | | | | | |
| Net rental income | 29.6% | 28.7% | 29.0% | 28.0% | 26.7% | 2.9% | 10.9% |
| Depreciation and residual losses at termination | 23.5% | 21.9% | 22.7% | 21.2% | 20.3% | 3.2% | 15.8% |

(1) As a percent of average operating lease assets, quartlery amounts annualized.

| | Six Months En | Six Months Ended June 30, | | . 2003 |
|---|---------------|---------------------------|-------------|---------|
| | 2004 | 2003 | Amount | % |
| Balance Sheet (in millions) | | | | |
| Average operating lease assets outstanding | \$ 1,070 | \$ 1,937 | \$ (867) | (44.8)% |
| Income Statement (in thousands) | | | | |
| Net rental income | 155,919 | \$ 250,776 | \$ (94,857) | (37.8) |
| Fees | 8,381 | 11,047 | (2,666) | (24.1) |
| Recoveries - early terminations | 3,273 | 4,944 | (1,671) | (33.8) |
| | | | | |
| Total Operating Lease Income | 167,573 | 266,767 | (99,194) | (37.2) |
| | | | | |
| Depreciation and residual losses at termination | 121,344 | 190,670 | (69,326) | (36.4) |
| Losses - early termination | 11,929 | 23,857 | (11,928) | (50.0) |
| | | | | |
| Total Operating Lease Expense | 133,273 | 214,527 | (81,254) | (37.9) |
| | | | | |
| Net Earnings Contribution | \$ 34,300 | \$ 52,240 | \$ (17,940) | (34.3)% |
| | | | | |
| Earnings ratios (1) | | | | |
| Net rental income | 29.1% | 25.9% | 3.3% | 12.6% |
| Depreciation and residual losses at termination | 22.7% | 19.7% | 3.0% | 15.2% |

(1) As a percent of average operating lease assets, quartlery amounts annualized.



Operating lease assets represent automobile leases originated before May 2002. This operating lease portfolio will run-off over time since all automobile lease originations after April 2002 have been recorded as direct financing leases and are reported in the automobile loan and lease category in earning assets. As a result, the non-interest income and non-interest expenses associated with the operating lease portfolio will also decline over time.

2004 Second Quarter versus 2003 Second Quarter and 2004 First Quarter

Average operating lease assets in the 2004 second quarter were \$1.0 billion, down 46% from the year-ago quarter and 12% from the 2004 first quarter.

Operating lease income, which totaled \$78.7 million in the 2004 second quarter, represented 36% of non-interest income in the quarter. Operating lease income was down \$49.9 million, or 39%, from the year-ago quarter and \$10.2 million, or 11%, from the 2004 first quarter, reflecting the declines in average operating leases. As no new operating leases have been originated after April 2002, the operating lease asset balances will continue to decline through both depreciation and lease terminations. Net earnings contribution was down 37% and 11%, respectively, from the year-ago and 2004 first quarter. Fees declined 11% from the year-ago quarter, but increased \$1.3 million, or 37%, from the first quarter, reflecting the recognition of deferred fees resulting from higher than expected prepayments of operating lease assets. Recoveries from early terminations declined 45% from the year-ago quarter and 19% from the first quarter.

Operating lease expense totaled \$62.6 million, down \$40.4 million, or 39%, from the year-ago quarter and down \$8.1 million, or 12%, from the 2004 first quarter. These declines also reflected the fact that this portfolio is decreasing over time as no new operating leases are being originated. Losses on early terminations, included in operating lease expense, declined \$6.4 million, or 55%, from the year-ago quarter, and \$1.6 million, or 24%, from the prior quarter.

Losses on operating lease assets consist of residual losses at termination and losses on early terminations. Residual losses arise if the ultimate value or sales proceeds from the automobile are less then Black Book value, which represents the insured amount under the company's residual value insurance policies. This situation may occur due to excess wear-and-tear or excess mileage not collected from the lessee. Losses on early terminations occur when a lessee, due to credit or other reasons, turns in the automobile before the end of the lease term. A loss is realized if the automobile is sold for a value less than the net book value at the date of turn-in. Such losses are not covered by the residual value insurance policies. To the extent the company is successful in collecting any deficiency from the lessee, amounts received are recorded as recoveries from early terminations.

The ratio of operating lease asset credit losses, net of recoveries, to average operating lease assets outstanding was an annualized 1.51% in the current quarter, 1.97% in the year-ago quarter, and 1.78% in the 2004 first quarter.

On a quarterly basis, Management evaluates the amount of residual value losses that it anticipates will result from the estimated fair value of a leased vehicle being less than the residual value inherent in the lease. Fair value includes estimated net proceeds from the sale of the leased vehicle plus expected residual value insurance proceeds and amounts expected to be collected from the lessee for excess mileage and other items that are billable under terms of the lease contract. When estimating the amount of expected insurance proceeds, Management takes into consideration policy caps that exist in two of the residual value insurance policies and whether it expects aggregate claims under such policies to exceed these caps. Management currently anticipates that aggregate claims for losses under the policy that insurance policy cap of \$120 million. Residual value losses exceeding the insurance policy cap are reflected in higher depreciation expense over the remaining life of the affected automobile lease. Also as part of its quarterly analysis, Management evaluates automobile leases individually for impairment.

All automobile leases covered under the two policies containing caps are accounted for using the operating lease method of accounting. Accordingly, residual value losses are provided for as additional depreciation expense over the remaining term to maturity of the underlying lease so that the carrying amount of the operating lease asset at the end of the lease term does not exceed its estimated fair value. At June 30, 2004, estimated future residual value losses on operating leases outstanding were expected to be between \$15 million to \$19 million. As of June 30, 2004, \$11.7 million of this amount had been provided for through additional depreciation expense. Any remaining losses above \$11.7 million will be provided for through additional depreciation expense over future periods.

2004 First Six Months versus 2003 First Six Months

Average operating lease assets in the first six months of 2004 were \$1.1 billion, down 45% from the comparable year-ago period.

Operating lease income, which totaled \$167.6 million in the first six months of 2004, was down \$99.2 million, or 37%, from the comparable year-ago period, reflecting the decline in average operating leases. As no new operating leases have been originated after April 2002, the operating lease asset balances will continue to decline through both depreciation and lease terminations. Net rental income for the first six months of 2004 was down 38%, with fees declining 24%. Recoveries from early terminations declined 34% from the comparable year-ago six-month period.

Operating lease expense for the first six months of 2004 totaled \$133.3 million, down \$81.3 million, or 38%, from the year-ago period. This decline also reflected the fact that this portfolio is decreasing over time as no new operating leases are being originated. Losses on early terminations for the first six months of 2004, included in operating lease expense, declined \$11.9 million, or 50%, from the comparable year-ago period.

The ratio of operating lease asset credit losses, net of recoveries, to average operating lease assets outstanding for the first six months was an annualized 1.62% in the current six-month period, down from 1.95% in the comparable year-ago period.

Provision for Income Taxes

The provision for income taxes in the second quarter of 2004 was \$43.4 million and represented an effective tax rate on income before taxes of 28.3%. The provision for income taxes increased \$6.7 million from the year-ago quarter, primarily due to higher pre-tax income. The effective tax rates in the second quarter of 2003 and first quarter of 2004 were 27.5% and 25.1%, respectively. For the first six months of 2004, provision for income taxes was \$78.3 million and represented an effective tax rate on income before taxes of 26.8%. This increased \$11.0 million from the same period in 2003, in which the effective tax rate was 26.3%, reflecting higher pre-tax income.

Each quarter, taxes for the full year are estimated and year-to-date tax accrual adjustments are made. Revisions to the full year estimate of accrued taxes occur periodically due to changes in the tax rates, audit resolution with taxing authorities, and newly enacted statutory, judicial, and regulatory guidance. These changes, when they occur, affect accrued taxes and can result in fluctuations in the quarterly effective tax rate.

The cost of investments in partnerships, along with the related tax credit is recognized in the financial statements as a component of income taxes under the effective yield method. The cost of the investment in partnerships is reported in non-interest expense. During the second quarter 2004, \$5.8 million of costs related to such investments in partnerships were recorded in non-interest expense.

In accordance with FAS 109, Accounting for Income Taxes, no deferred income taxes are to be recorded when a company intends to permanently reinvest their earnings from a foreign activity. As of June 30, 2004, the company intended to permanently reinvest the earnings from its foreign asset securitization activities of approximately \$79.4 million.

Management expects the 2004 effective tax rate to remain below 30% as the level of tax-exempt income, general business credits, and asset securitization activities remain consistent with prior years.

CREDIT RISK

Credit risk is the risk of loss due to adverse changes in a borrower's ability to meet its financial obligations under agreed upon terms. The company is subject to credit risk in lending, trading, and investment activities. The nature and degree of credit risk is a function of the types of transactions, the structure of those transactions, and the parties involved. The majority of the company's credit risk is associated with lending activities, as the acceptance and management of credit risk is central to profitable lending. Credit risk represents a limited portion of the total risks associated with the investment portfolio and is incidental to trading activities. Credit risk is management. There are very specific and differing methodologies for managing credit risk for commercial credits compared with consumer credits (see Credit Risk Management section of the company's 2003 Form 10-K for a complete discussion).

Loan and Lease Composition

Table 10 reflects period-end loan and lease portfolio mix by type of loan or lease, as well as by business segment:

Table 10 - Loans and Lease Portfolio Composition

| | June : 2004 | · · | March 2004 | | Decembre 200 | | Septemb 2003 | | June 3 200 | |
|--|----------------|-------------|--------------------|----------------|-----------------|--------|-----------------|--------|----------------|--------|
| (in millions) | Amount | % | Amount | % | Amount | % | Amount | % | Amount | % |
| Ву Туре | | | | | | | | | | |
| Commercial and industrial | \$ 5,277 | 23.3% | \$ 5,480 | 25.9% | \$ 5,314 | 25.2% | \$ 5,433 | 25.7% | \$ 5,532 | 28.9% |
| Commercial real estate | 4,514 | 19.9 | 4,272 | 20.2 | 4,172 | 19.8 | 4,047 | 19.1 | 3,951 | 20.7 |
| Total Commercial | 9,791 | 43.2 | 9,752 | 46.0 | 9,486 | 45.0 | 9,480 | 44.8 | 9,483 | 49.6 |
| Consumer | | | | | | | | | | |
| Automobile loans | 1,815 | 8.0 | 2,267 | 10.7 | 2,992 | 14.2 | 3,709 | 17.5 | 2,367 | 12.4 |
| Automobile leases | 2,185 | 9.6 | 2,267 | 9.7 | 1,902 | 9.0 | 1,688 | 8.0 | 1,481 | 7.9 |
| Home equity | 4,315 | 9.0 19.0 | 3,980 | 18.8 | 3,792 | 9.0 | 3,590 | 17.0 | 3,436 | 18.0 |
| Residential mortgage | 3,284 | 19.0 | 2,757 | 13.0 | 2,531 | 12.0 | 2,326 | 11.0 | 1,915 | 10.0 |
| Other loans | 3,284 386 | 14.5 | 372 | 13.0 | 2,331 | 12.0 | 2,320 | 11.0 | 378 | 2.1 |
| Other loans | 380 | 1./ | 572 | 1.8 | 572 | 1.8 | 580 | 1./ | 578 | 2.1 |
| Total Consumer | 11,984 | 52.9 | 11,442 | 54.0 | 11,589 | 55.0 | 11,693 | 55.2 | 9,577 | 50.4 |
| Total Loans and Leases | \$ 21,776 | 96.1% | \$ 21,194 | 100.0% | \$ 21,075 | 100.0% | \$ 21,173 | 100.0% | \$ 19,060 | 100.0% |
| | . | | ¢ 1000 | | * 4.00.4 | | ¢ 5.207 | | A A A A | |
| Total automobile loans and leases | \$ 3,999 | 17.6 | \$ 4,333 | | \$ 4,894 | | \$ 5,397 | | \$ 3,848 | |
| Operating lease assets | 889 | 3.9 | 1,071 | | 1,260 | | 1,455 | | 1,673 | |
| Securitized loans | | _ | 28 | | 37 | | 49 | | 1,076 | |
| Total Automobile Exposure ⁽²⁾ | \$ 4,888 | 21.6% | \$ 5,432 | 24.4% | \$ 6,191 | 27.7% | \$ 6,901 | 30.4% | \$ 6,597 | 30.2% |
| | | | | | | | | | | |
| Total Credit Exposure | \$ 22,664 | 100.0% | \$ 22,293 | 100.0% | \$ 22,372 | 100.0% | \$ 22,677 | 100.0% | \$ 21,809 | 100.0% |
| | | | | — | | — | | | | |
| By Business Segment ⁽³⁾ | | | | | | | | | | |
| Regional Banking | | | * * * * * * | 22 1 21 | | | . | 10.00/ | . | 10 |
| Central Ohio | \$ 5,652 | 24.9% | \$ 4,988 | 22.4% | \$ 4,652 | 20.8% | \$ 4,491 | 19.8% | \$ 4,080 | 18.7% |
| Northern Ohio | 2,694 | 11.9 | 2,681 | 12.0 | 2,579 | 11.5 | 2,639 | 11.6 | 2,712 | 12.4 |
| Southern Ohio/Kentucky | 1,759 | 7.8 | 1,703 | 7.6 | 1,677 | 7.5 | 1,623 | 7.2 | 1,547 | 7.1 |
| West Michigan | 2,216 | 9.8 | 2,155 | 9.7 | 2,077 | 9.3 | 2,028 | 8.9 | 1,967 | 9.0 |
| East Michigan | 1,359 | 6.0 | 1,341 | 6.0 | 1,268 | 5.7 | 1,306 | 5.8 | 1,225 | 5.6 |
| West Virginia | 811 811 | 3.6 | 808 | 3.6 | 802 731 | 3.6 | 802 741 | 3.5 | 796 729 | 3.6 |
| Indiana | 811 | 3.6 | 753 | 3.4 | /31 | 3.3 | /41 | 3.3 | 129 | 3.3 |
| Total Regional Banking | 15,302 | 67.5 | 14,429 | 64.7 | 13,786 | 61.7 | 13,630 | 60.1 | 13,056 | 59.9 |
| Dealer Sales | 5,840 | 25.8 | 6,399 | 28.7 | 7,095 | 31.6 | 7,598 | 33.5 | 7,373 | 33.8 |
| Private Financial Group | 1,381 | 6.1 | 1,322 | 5.9 | 1,296 | 5.8 | 1,260 | 5.6 | 1,181 | 5.4 |
| Treasury / Other | 1,501 | 0.6 | 143 | 0.7 | 195 | 0.9 | 189 | 0.8 | 199 | 0.9 |
| Total Credit Exposure | \$ 22,664 | 100.0% | \$ 22,293 | 100.0% | \$ 22,372 | 100.0% | \$ 22,677 | 100.0% | \$ 21,809 | 100.0% |
| | | | | | | | | | | |

(1) Effective June 30, 2004, \$282 million of commercial and industrial loans were reclassified to commercial real estate to conform to the classification of these loans with the presentation of similar loans.

⁽²⁾ Total Loans and leases, operating lease assets and securitized loans.

⁽³⁾ Prior period amounts have been reclassified to conform to the current period business segment structure.

During 2004, the composition of the loan and lease portfolio changed such that lower credit risk home equity loans and residential mortgages represented 19% and 14%, respectively, of total credit exposure at June 30, 2004, up from 18% and 10%, respectively, a year earlier. Conversely, C&I loans have declined from 29% a year ago to 23% at June 30, 2004, reflecting, in part, strategies to exit large, individual commercial credits, including out-of-footprint shared national credits.

At the beginning of the 2004 second quarter, the criteria for categorizing commercial loans as either C&I loans or CRE loans was clarified. The new criteria are based on the purpose of the loan. Previously, the categorization was based on the nature of the collateral securing, or partially securing, the loan. As a result of this change, \$282 million in C&I loans were reclassified to CRE loans effective June 30, 2004. This change had no impact on total commercial loans, underlying credit quality, or 2004 second quarter reported income. Under this new methodology, as new loans are originated or existing loans renewed, loans secured by owner-occupied real estate will be categorized as C&I loans (previously CRE loans) and unsecured loans for the purpose of developing real estate will be categorized as CRE loans (previously C&I loans).

The company also has a portfolio of automobile operating lease assets. Although these assets are reflected on the balance sheet, they are not part of total loans and leases or earning assets. In addition, prior to June 30, 2004, there was a small pool of securitized automobile loans, which represented off-balance sheet securitized automobile loan assets. Both of these asset classes represent automobile financing credit exposure, despite not being components of total loans and leases. As such, operating lease assets and securitized loans are added to the on-balance sheet automobile loans and leases to determine a total automobile financing exposure, which Management finds helpful in evaluating the overall credit risk for the company.

On June 30, 2004, \$512 million of automobile loans were sold and \$102 million of automobile loans were transferred to loans held for sale. Combined, these transactions resulted in second quarter net pre-tax gains on the sale of automobile loans of \$4.9 million: \$5.1 million associated with the \$512 million sale, partially offset by a \$0.2 million lower of cost or market write-down on the loans held for sale. On a combined basis, these transactions increased the total automobile loans sold since the beginning of 2003 to \$3.6 billion. These sales represented a continuation of a strategy to reduce exposure to automobile financing to approximately 20% of total credit exposure (see Table 10). At June 30, 2004, this exposure was \$4.9 billion, down from \$6.2 billion at year-end, and represented 22% of total credit exposure, down from 24% at the end of the last quarter and from 33% at the end of 2002.

Net Loan and Lease Charge-offs

Table 11 reflects net loan and lease charge-off detail for each of the last five quarters, and the first six-month period for 2003 and 2004:

Table 11 - Net Loan and Lease Charge-offs

Net Charge-offs by Loan and Lease Type

| | 20 | 004 | 2003 | | | |
|------------------------------------|------------|-----------|-----------|-----------|-----------|--|
| (in thousands) | Second | First | Fourth | Third | Second | |
| Commercial and industrial | \$ (2,803) | \$ 5,956 | \$ 31,186 | \$ 12,222 | \$ 26,546 | |
| Commercial real estate | 2,940 | 1,637 | 5,743 | 3,621 | 607 | |
| Total Commercial | 137 | 7,593 | 36,929 | 15,843 | 27,153 | |
| Consumer | | | | | | |
| Automobile loans | 5,604 | 13,422 | 11,346 | 10,773 | 7,524 | |
| Automobile direct financing leases | 2,159 | 3,159 | 1,936 | 1,450 | 1,422 | |
| | | | | | | |
| Automobile loans and leases | 7,763 | 16,581 | 13,282 | 12,223 | 8,946 | |
| | | | | | | |
| Home equity | 3,019 | 3,116 | 3,464 | 3,416 | 3,671 | |
| Residential mortgage | 302 | 316 | 174 | 246 | 267 | |
| Other loans | 1,294 | 1,021 | 1,294 | 1,046 | 1,019 | |
| | | · | | | | |
| Total Consumer | 12,378 | 21,034 | 18,214 | 16,931 | 13,903 | |
| Total Net Charge-offs | \$ 12,515 | \$ 28,627 | \$ 55,143 | \$ 32,774 | \$ 41,056 | |
| | | | | | | |

Net Charge-offs - Annualized Percentages

| | 2004 | | | | |
|---|---------|-------|--------|----------|--------|
| | Second | First | Fourth | Third | Second |
| Commercial and industrial | (0.20)% | 0.44% | 2.32% | \$ 0.91% | 1.89% |
| Commercial real estate | 0.28 | 0.16 | 0.56 | 0.36 | 0.06 |
| Total Commercial | 0.01 | 0.32 | 1.55 | 0.68 | 1.14 |
| Consumer | | | | | |
| Automobile loans | 0.96 | 1.77 | 1.29 | 1.20 | 1.06 |
| Automobile direct financing leases | 0.40 | 0.64 | 0.43 | 0.36 | 0.44 |
| Automobile loans and leases | 0.69 | 1.32 | 1.00 | 0.94 | 0.87 |
| Home equity | 0.29 | 0.32 | 0.38 | 0.39 | 0.44 |
| Residential mortgage | 0.04 | 0.05 | 0.03 | 0.05 | 0.06 |
| Other loans | 1.30 | 1.15 | 1.34 | 1.14 | 1.08 |
| | | | | | |
| Total Consumer | 0.41 | 0.70 | 0.61 | 0.61 | 0.57 |
| Net Charge-offs as a % of Average Loans | 0.23% | 0.53% | 1.03% | 0.64% | 0.85% |

Table 11 - Net Loan and Lease Charge-offs, Continued

Net Charge-offs by Loan and Lease Type

| | Six Months E | Ending June 30, |
|------------------------------------|--------------|-----------------|
| (in thousands) | 2004 | 2003 |
| Commercial and industrial | \$ 3,153 | \$ 41,450 |
| Commercial real estate | 4,577 | 1,153 |
| Total Commercial | 7,730 | 42,603 |
| Consumer | | |
| Automobile loans | 19,026 | 18,147 |
| Automobile direct financing leases | 5,318 | 2,342 |
| Automobile loans and leases | 24,344 | 20,489 |
| Home equity | 6,135 | 7,724 |
| Residential mortgage | 618 | 412 |
| Other loans | 2,315 | 2,664 |
| | | |
| Total Consumer | 33,412 | 31,289 |
| Total Net Charge-offs | \$ 41,142 | \$ 73,892 |
| 0 | | |

Net Charge-offs - Annualized Percentages

| 2004 0.12% 0.22 | 2003 1.47% 0.06 |
|-----------------------|-----------------------|
| | |
| 0.22 | 0.06 |
| | |
| 0.16 | 0.90 |
| | |
| 1.42 | 1.23 |
| 0.52 | 0.40 |
| 1.02 | 1.63 |
| 1.02 | 1.03 |
| 0.31 | 0.47 |
| 0.04 | 0.04 |
| 1.23 | 1.39 |
| 0.56 | 0.65 |
| 0.38% | 0.77% |
| | 0.04 1.23 0.56 |

Six Months Ending June 30

2004 Second Quarter versus 2003 Second Quarter and 2004 First Quarter

Total net charge-offs for the 2004 second quarter were \$12.5 million, or an annualized 0.23% of average total loans and leases. This was a reduction from \$41.1 million, or 0.85%, in the year-ago quarter and from \$28.6 million, or 0.53% of average total loans and leases, in the prior quarter. Total net charge-offs in the current and 2004 first quarter were influenced by one-time events.

Current quarter total net charge-offs were reduced by a \$9.7 million one-time recovery on a C&I loan charged-off in the fourth quarter of 2002. This recovery lowered total commercial (C&I and CRE) net charge-offs by an annualized 39 basis points and total loan and lease net charge-offs by 18 basis points. Excluding the impact of this recovery, current quarter total net charge-offs would have been \$22.2 million, or an annualized 0.41% of average total loans and leases. As previously reported, total net charge-offs in the 2004 first quarter included a one-time \$4.7 million cumulative increase in automobile loan and lease charge-offs would have represented 0.45% of average total loans and leases.

Total commercial (C&I and CRE) net charge-offs in the second quarter were only \$0.1 million. Adjusting for the \$9.7 million recovery noted above (39 basis point impact), total commercial net charge-offs would have been \$9.8 million, or an annualized 0.40% of related loans, down from \$27.2 million, or 1.14% of related loans, in the year-ago quarter, and up from \$7.6 million, or 0.32%, in the 2004 first quarter.

Total consumer net charge-offs in the current quarter were \$12.4 million, or an annualized 0.41% of related loans. This compared with \$13.9 million, or 0.57%, in the year-ago quarter, and \$21.0 million, or 0.70% of related loans, in the 2004 first quarter. First quarter net consumer charge-offs included 15 basis points related to the one-time \$4.7 million cumulative adjustment noted above.

Total automobile loan and lease net charge-offs in the 2004 second quarter were \$7.8 million, or 0.69% of average automobile loans and leases. This compared with \$8.9 million of net charge-offs, or 0.87%, in the year-ago quarter and \$16.6 million, or 1.32%, in the first quarter. The first quarter net charge-offs included 37 basis points from the one-time \$4.7 million cumulative adjustment.

2004 First Six Months versus 2003 First Six Months

Total net charge-offs for the first six months of 2004 were \$41.1 million, or an annualized 0.38% of average total loans and leases. This was a reduction from \$73.8 million, or 0.77%, in the comparable year-ago period.

Total commercial (C&I and CRE) net charge-offs in the first six months of 2004 were only \$7.7 million, or 0.16%, down from \$42.6 million, or 0.90%, in the comparable year-ago period. The decline from the year-ago period reflected a \$9.7 million C&I recovery in the 2004 first six-month period.

Total consumer net charge-offs in the first six months of 2004 were \$33.4 million, or 0.56% of related loans. This compared with \$31.2 million, or 0.65%, in the comparable year-ago period.

Total automobile loan and lease net charge-offs in the first six months of 2004 were \$24.3 million, or 1.02% of average automobile loans and leases. This compared with \$20.5 million, or 1.00% of average automobile loans and leases, in the year-ago six-month period.

For the second half of 2004, total loan and lease net charge-offs are expected to be in the 0.40%-0.45% range, including third and fourth quarter seasonality associated with automobile loan and lease net charge-offs.

Three Months Ended

Non-performing Assets and Past Due Loans and Leases

Table 12 reflects period-end NPAs and past due loans and leases detail for each of the last five quarters:

Table 12 - Non-Performing Assets and Past Due Loans and Leases

| | Three Months Ended | | | | | | | |
|---|--------------------|-------------------|----------------------|-----------------------|------------------|--|--|--|
| (in thousands) | June 30, 2004 | March 31, 2004 | December 31, 2003 | September 30, 2003 | June 30, 2003 | | | |
| Non-accrual loans and leases | | | | | | | | |
| Commercial and industrial | \$ 32,044 | \$ 45,056 | \$ 43,387 | \$ 82,413 | \$ 86,021 | | | |
| Commercial real estate | 15,782 | 20,019 | 22,399 | 30,545 | 22,398 | | | |
| Residential mortgage | 13,952 | 12,052 | 9,695 | 8,923 | 11,735 | | | |
| Total non-performing loans and leases (NPLs) | 61,778 | 77,127 | 75,481 | 121,881 | 120,154 | | | |
| Other real estate, net | 12,918 | 14,567 | 11,905 | 15,196 | 13,568 | | | |
| Total Non-performing Assets (NPAs) | \$ 74,696 | \$ 91,694 | \$ 87,386 | \$ 137,077 | \$ 133,722 | | | |
| Accruing loans and leases past due 90 days or more | \$ 51,490 | \$ 59,697 | \$ 55,913 | \$ 66,060 | \$ 55,287 | | | |
| NPLs as a % of total loans and leases | 0.28% | 0.36% | 0.36% | 0.58% | 0.63% | | | |
| NPLs as a % of total loans and leases and other real estate | 0.34% | 0.43% | 0.41% | 0.65% | 0.70% | | | |
| Allowance for loan and lease losses as a % of: | | | | | | | | |
| NPLs | 464% | 383% | 397% | 276% | 256% | | | |
| NPAs | 384% | 322% | 343% | 245% | 230% | | | |
| Allowance for loan and lease losses plus allowance for unfunded commitments and letters of credit: | | | | | | | | |
| NPLs | 515% | 425% | 444% | 304% | 284% | | | |
| NPAs | 426% | 357% | 384% | 270% | 255% | | | |
| Accruing loans and leases past due 90 days or more to total loans and leases | 0.24% | 0.28% | 0.27% | 0.31% | 0.29% | | | |



NPAs were \$74.7 million at June 30, 2004, down \$59.0 million, or 44%, from the prior year, and down \$17.0 million, or 19%, from March 31, 2004. NPAs as a percent of total loans and leases and other real estate were 0.34% at June 30, 2004, down from 0.70% a year-ago and from 0.43% at March 31, 2004. NPAs at June 30, 2004, included \$23.3 million of lower-risk residential real estate related assets, which represented 31% of total NPAs. This compared with \$20.7 million, or 15%, at the end of the year-ago quarter.

The over 90-day delinquent, but still accruing, ratio was 0.24% at June 30, 2004, down from 0.29% a year ago and 0.28% at March 31, 2004.

Table 13 - Non-Performing Asset Activity

| | Three Months Ended | | | | | | | |
|----------------------------|--------------------|-------------------|----------------------|-----------------------|------------------|--|--|--|
| (in thousands) | June 30, 2004 | March 31, 2004 | December 31, 2003 | September 30, 2003 | June 30, 2003 | | | |
| Beginning of Period | \$ 91,694 | \$ 87,386 | \$ 137,077 | \$ 133,722 | \$ 140,725 | | | |
| New non-performing assets | 25,727 | 27,208 | 38,367 | 52,213 | 83,104 | | | |
| Returns to accruing status | (1,493) | (54) | (454) | (319) | (9,866) | | | |
| Loans and lease losses | (12,872) | (10,463) | (39,657) | (22,090) | (30,204) | | | |
| Payments | (13,571) | (10,717) | (22,710) | (18,905) | (26,831) | | | |
| Sales | (14,789) | (1,666) | (25,237) | (7,544) | (23,206) | | | |
| | | · | · | | | | | |
| End of Period | \$ 74,696 | \$ 91,694 | \$ 87,386 | \$ 137,077 | \$133,722 | | | |
| | | | | | | | | |

Table 13 reflects NPA activity and shows that during the 2004 second quarter the level of new NPAs continued to decline and totaled \$25.7 million, down 69% from the year-ago quarter, and 5% less than in the first quarter. Also, impacting the decline in period-end NPAs was the sale of \$14.8 million of NPAs during the 2004 second quarter.

Allowances for Credit Losses (ACL)

The allowances for credit losses (ACL) include the allowance for loan and lease losses (ALLL) and the allowance for unfunded loan commitments and letters of credit (AULC). Table 14 reflects activity in the ALLL and AULC for the past five quarters:

Table 14 - Allowances for Credit Losses

| | Three Months Ended | | | | | | | |
|--|--------------------|-------------------|----------------------|-----------------------|------------------|--|--|--|
| (in thousands) | June 30, 2004 | March 31, 2004 | December 31, 2003 | September 30, 2003 | June 30, 2003 | | | |
| Allowance for Loan and Lease Losses, Beginning of Period | \$ 295,377 | \$ 299,732 | \$ 336,398 | \$ 307,667 | \$ 303,636 | | | |
| Loan and lease losses | (30,845) | (37,167) | (68,023) | (43,261) | (49,985) | | | |
| Recoveries of loans previously charged off | 18,330 | 8,540 | 12,880 | 10,487 | 8,929 | | | |
| Net loan and lease losses | (12,515) | (28,627) | (55,143) | (32,774) | (41,056) | | | |
| Provision for credit losses | 5,027 | 25,596 | 26,341 | 51,615 | 49,193 | | | |
| Net change in allowance for unfunded loan commitments and lettes | | | | | | | | |
| of credit | 896 | 3,433 | (1,785) | (457) | 101 | | | |
| Allowance of assets sold and securitized ⁽¹⁾ | (1,850) | (4,757) | (6,079) | 10,347 | (4,207) | | | |
| Allowance for Loan and Lease Losses, End of Period | \$ 286,935 | \$ 295,377 | \$ 299,732 | \$ 336,398 | \$ 307,667 | | | |
| Allowance for Unfunded Loan Commitments and Letters of | | | | | | | | |
| Credit, Beginning of Period | \$ 32,089 | \$ 35,522 | \$ 33,737 | \$ 33,280 | \$ 33,381 | | | |
| Net change | (896) | (3,433) | 1,785 | 457 | (101) | | | |
| Allowance for Unfunded Loan Commitments and Letters of | | . | ÷ | | | | | |
| Credit, End of Period | \$ 31,193 | \$ 32,089 | \$ 35,522 | \$ 33,737 | \$ 33,280 | | | |
| Total Allowances for Credit Losses | \$ 318,128 | \$ 327,466 | \$ 335,254 | \$ 370,135 | \$ 340,947 | | | |
| Allowances for Credit Losses as a % of total loans and leases | 1.46% | 1.55% | 1.59% | 1.75% | 1.79% | | | |
| The ALL as a % of total loans and leases | 1.32% | 1.39% | 1.42% | 1.59% | 1.61% | | | |
| Components: | 1.52 /0 | 1.5970 | 1.4270 | 1.5570 | 1.0170 | | | |
| Transaction reserve | 0.86% | 0.91% | 0.88% | 0.98% | 1.05% | | | |
| Economic reserve | 0.36 | 0.38 | 0.40 | 0.47 | 0.45 | | | |
| Specific reserve | 0.10 | 0.10 | 0.14 | 0.14 | 0.11 | | | |
| 1 | | | | | | | | |

(1) The third quarter 2003 includes the reserve for loan losses associated with automobile loans contained in one of Huntington's securitizations trusts consolidated as a result of the adoption of FASB Interpretation No. 46 on July 1, 2003.

The June 30, 2004, ALLL was \$286.9 million, down from \$307.7 million a year ago and from \$295.4 million at March 31, 2004. These declines primarily reflected improving credit quality and the change in the mix of the loan portfolio to lower-risk residential mortgages and home equity loans. Expressed as a percent of period-end loans and leases, the ALLL at June 30, 2004, was 1.32%, down from 1.61% a year-ago and from 1.39% at March 31, 2004. The ALLL as a percent of NPAs was 384% at June 30, 2004, up from 230% a year ago and from 322% at March 31, 2004.

The June 30, 2004, AULC was \$31.2 million, down slightly from \$33.3 million at the end of the year-ago quarter, and from \$32.1 million at March 31, 2004.

On a combined basis, the ACL as a percent of total loans and leases was 1.46% at June 30, 2004, compared with 1.79% a year ago and 1.55% at the end of last quarter. Similarly, the ACL as a percent of NPAs was 426% at June 30, 2004, compared with 255% a year earlier and 357% at March 31, 2004.

The ALLL consists of three components, the transaction reserve, the economic reserve, and specific reserves (see the Credit Risk discussion in company's 2003 Form 10-K for additional discussion).

<u>Transaction reserve</u> – This ALLL component is based on historical portfolio performance information. Specifically, the probability-of-default and the loss-in-event-of-default are assigned an expected risk factor based on the type and structure of each credit. Reserve factors are then calculated and applied at an individual loan level for all products.

Specific reserve – This ALLL component represents the sum of credit-by-credit reserve decisions for individual C&I and CRE loans when it is determined that the related expected risk factor is insufficient to cover the estimated losses embedded in the specified credit facility.

Economic reserve – This ALLL component reflects anticipated losses impacted by changes in the economic environment. As previously reported, effective January 1, 2004, the company adopted a significantly more quantitative approach to the calculation of the economic reserve component. In order to quantify the economic reserve, the company identified four statistically significant indicators of loss volatility over the seven-year period from 1996 through 2003. The four variables as identified by the regression model are: (1) the US Index of Leading Economic Indicators, (2) the US Corporate Profits Index, (3) the US Unemployment Index, and (4) the University of Michigan Current Consumer Confidence Index.

This methodology permits the decomposition of the total ALLL ratio into these three components and provides increased insight into the rationale for increases or decreases in the overall ALLL ratio. As shown in Table 14, the ALLL ratio at June 30, 2004 was 1.32%, of which 0.86% represented the transaction reserve, 0.36% the economic reserve, and 0.10% specific reserves. This analysis shows that of the 7 basis point decline in the ALLL ratio from 1.39% at March 31, 2004, the transaction reserve accounted for 5 basis points. The sale of lower-performing commercial credits during the 2004 second quarter, along with the release of their inherent higher than average transaction reserve component, as well as the change in the overall loan and lease portfolio toward higher credit quality loans, contributed to this decline. The remaining 2 basis points of decline in the ALLL ratio represented lower relative economic reserves, reflecting an improved economic outlook. Specific reserves were unchanged at 10 basis points. This more quantitative methodology for determining the ALLL will be more responsive to changes in the portfolio mix, the economic environment, and to individual credit situations, with the result being an ALLL ratio that exhibits greater quarterly fluctuations.

MARKET RISK

Market risk is the potential for losses in the fair value of the company's assets and liabilities due to changes in interest rates, exchange rates, and equity prices. The company incurs market risk in the normal course of business. Market risk arises when the company extends fixed-rate loans, purchases fixed-rate securities, originates fixed-rate certificates of deposit (CDs), obtains funding through fixed-rate borrowings, and leases automobiles and equipment based on expected lease residual values. Market risk arising from changes in interest rates, which affects the market values of fixed-rate assets and liabilities, is interest rate risk. Market risk arising from the possibility that the uninsured residual value of leased assets will be different at the end of the lease term than was estimated at the lease's inception is residual value risk. From time to time, the company also has small exposures to trading risk and foreign exchange risk. At June 30, 2004, the company had \$20.6 million of trading assets, primarily in its broker/dealer businesses.

Interest Rate Risk

Interest rate risk is the primary market risk incurred by the company. It results from timing differences in the repricing and maturity of assets and liabilities and changes in relationships between market interest rates and the yields on assets and rates on liabilities, including the impact of embedded options.

Management seeks to minimize the impact of changing interest rates on the company's net interest income and the fair value of assets and liabilities. The board of directors establishes broad policies regarding interest rate and market risk and liquidity risk. The asset and liability committee (ALCO) establishes specific operating limits within the parameters of the board of directors' policies. ALCO regularly monitors position concentrations and the level of interest rate sensitivity to ensure compliance with board of directors approved risk tolerances (see Interest Rate Risk discussion in the company's 2003 Form 10-K for a complete discussion.)



Interest rate risk modeling is performed monthly. Two broad approaches to modeling interest rate risk are employed: income simulation and economic value analysis. An income simulation analysis is used to measure the sensitivity of forecasted net interest income to changes in market rates over a one-year horizon. The economic value analysis (Economic Value of Equity or EVE) is calculated by subjecting the period-end balance sheet to changes in interest rates and measuring the impact of the changes in the value of the assets and liabilities.

The simulations for evaluating short-term interest rate risk exposure are scenarios that model gradual 100 and 200 basis point increasing and decreasing parallel shifts in interest rates over the next twelve-month period beyond the interest rate change implied by the current yield curve. The table below shows the results of the scenarios as of June 30, 2004, and March 31, 2004. All of the positions were well within the board of directors' policy limits.

Net Interest Income at Risk (%)

| Basis point change scenario Board Policy Limits | -200 - 4.0% | -100 -2.0% | +100 -2.0% | +200 -4.0% |
|--|-----------------------|---------------|---------------|---------------|
| June 30, 2004 | N.M. | -0.3% | -0.0% | -0.1% |
| March 31, 2004 | N.M. | -0.5% | -0.1% | -0.3% |
| December 31, 2003 | N.M. | -0.3% | -0.2% | -0.5% |

N.M., not meaningful.

The primary simulations for EVE risk assume an immediate and parallel increase in rates of +/- 100 and +/- 200 basis points beyond any interest rate change implied by the current yield curve. The table below outlines the results compared to the previous quarter and policy limits.

Economic Value of Equity at Risk (%)

| Basis point change scenario Board Policy Limits | -200 -12.0% | -100 -5.0% | +100 -5.0% | +200 -12.0% |
|--|----------------|---------------|---------------|----------------|
| June 30, 2004 | N.M. | +1.5% | -2.8% | -6.2% |
| March 31, 2004 | N.M. | +0.1% | -2.7% | -6.2% |
| December 31, 2003 | N.M. | +1.8% | -3.5% | -7.9% |

N.M., not meaningful.

LIQUIDITY RISK

The objective of effective liquidity management is to ensure that cash flow needs can be met on a timely basis at a reasonable cost under both normal operating conditions and unforeseen or unpredictable circumstances. The liquidity of the Bank is available to originate loans and leases and to repay deposit and other liabilities as they become due or are demanded by customers. Liquidity risk arises from the possibility that funds may not be available to satisfy current or future commitments based on external macro market issues, investor perception of financial strength, and events unrelated to the company such as war, terrorism, or financial institution market specific issues (see Liquidity discussion in the company's 2003 Form 10-K for a complete discussion.)

The primary source of funding is core deposits from retail and commercial customers (see Table 15). As of June 30, 2004, core deposits totaled \$16.5 billion, and represented 85% of total deposits. This compared with \$15.9 billion, or 87% of total deposits, a year earlier. Most of the growth in core deposits was attributable to growth in interest bearing and non-interest bearing demand deposits as retail CDs declined.

Table 15 - Deposit Liabilities

| | June 3 2004 | | March 2004 | | Decembo 2003 | | Septemb 2003 | | June 200 | |
|---|----------------|--------|---------------|--------|-----------------|--------|-----------------|--------|-----------|--------|
| (in millions) | Amount | % | Amount | % | Amount | % | Amount | % | Amount | % |
| Ву Туре | | | | | | | | | | |
| Demand deposits | | | | | | | | | | |
| Non-interest bearing | \$ 3,327 | 17.1% | \$ 2,918 | 15.4% | \$ 2,987 | 16.2% | \$ 3,003 | 15.9% | \$ 3,110 | 16.9% |
| Interest bearing | 7,124 | 36.6 | 6,866 | 36.2 | 6,411 | 34.7 | 6,425 | 34.1 | 6,332 | 34.5 |
| Savings deposits | 3,011 | 15.5 | 3,002 | 15.8 | 2,960 | 16.0 | 3,000 | 15.9 | 3,085 | 16.8 |
| Retail certificates of deposit | 2,412 | 12.4 | 2,395 | 12.6 | 2,462 | 13.3 | 2,484 | 13.2 | 2,739 | 14.9 |
| Other domestic time deposits | 595 | 3.1 | 608 | 3.2 | 631 | 3.4 | 638 | 3.4 | 664 | 3.6 |
| Total Core Deposits | 16,470 | 84.7 | 15,789 | 83.2 | 15,451 | 83.6 | 15,550 | 82.5 | 15,930 | 86.7 |
| Domestic time deposits of \$100,000 or more | 808 | 4.2 | 791 | 4.2 | 789 | 4.3 | 844 | 4.5 | 826 | 4.5 |
| Brokered time deposits and negotiable CDs | 1,679 | 8.6 | 1,942 | 10.2 | 1,772 | 9.6 | 1,837 | 9.8 | 1,224 | 6.7 |
| Foreign time deposits | 508 | 2.5 | 467 | 2.4 | 475 | 2.5 | 603 | 3.2 | 391 | 2.1 |
| Total Deposits | \$ 19,465 | 100.0% | \$ 18,989 | 100.0% | \$ 18,487 | 100.0% | \$ 18,834 | 100.0% | \$ 18,371 | 100.0% |
| By Business Segment ⁽²⁾ | | | | | | | | | | |
| Central Ohio | \$ 4,386 | 22.5% | \$ 4,378 | 23.1% | \$ 4,184 | 22.6% | \$ 4,189 | 22.3% | \$ 5.045 | 27.5% |
| Northern Ohio | 3,774 | 19.4 | 3,517 | 18.5 | 3,505 | 19.0 | 3,531 | 18.8 | 3,529 | 19.2 |
| Southern Ohio/Kentucky | 1,559 | 8.0 | 1,476 | 7.8 | 1,442 | 7.8 | 1,437 | 7.6 | 1,413 | 7.7 |
| West Michigan | 2,599 | 13.4 | 2,609 | 13.7 | 2,457 | 13.3 | 2,529 | 13.4 | 2,582 | 14.1 |
| East Michigan | 2,081 | 10.7 | 2,030 | 10.7 | 1,988 | 10.8 | 2,000 | 10.6 | 2,078 | 11.3 |
| West Virginia | 1,369 | 7.0 | 1,292 | 6.8 | 1,315 | 7.1 | 1,324 | 7.0 | 1,339 | 7.3 |
| Indiana | 668 | 3.4 | 637 | 3.4 | 648 | 3.5 | 661 | 3.5 | 640 | 3.4 |
| Total Regional Banking | 16,435 | 84.4 | 15,939 | 84.0 | 15,539 | 84.1 | 15,671 | 83.2 | 16,626 | 90.5 |
| Dealer Sales | 71 | 0.4 | 77 | 0.4 | 77 | 0.4 | 65 | 0.4 | 68 | 0.4 |
| Private Financial Group | 1,016 | 5.2 | 1,057 | 5.6 | 1,164 | 6.3 | 1,117 | 5.9 | 1,029 | 5.6 |
| Treasury/Other ⁽¹⁾ | 1,943 | 10.0 | 1,916 | 10.0 | 1,707 | 9.2 | 1,981 | 10.5 | 648 | 3.5 |
| Total Deposits | \$ 19,465 | 100.0% | \$ 18,989 | 100.0% | \$ 18,487 | 100.0% | \$ 18,834 | 100.0% | \$ 18,371 | 100.0% |

⁽¹⁾ Comprised largely of brokered deposits and negotiable CDs.
 ⁽²⁾ Prior period amounts have been reclassified to conform to the

Prior period amounts have been reclassified to conform to the current period business segment structure.

Liquidity policies and limits are established by the board of directors, with operating limits set by ALCO. Two primary liquidity measures are the ratio of loans and operating lease assets to deposits and the percentage of assets funded with non-core, or wholesale, liabilities. The limits set by the board for these two liquidity measures are 135% and 40%, respectively. At June 30, 2004, the actual ratio of loans and operating leases to deposits was 116%, while the percentage of assets funded with non-core or wholesale liabilities was 35%. In addition, guidelines are established by ALCO to ensure diversification of wholesale funding by type, source, and maturity and provide sufficient balance sheet liquidity to cover 100% of wholesale funds maturing within a six-month time period. A contingency funding plan is in place, which includes forecasted sources and uses of funds under various scenarios in order to prepare for unexpected liquidity shortages, including the implications of any rating agency changes. ALCO meets monthly to identify and monitor liquidity issues, provide policy guidance, and oversee adherence to, and the maintenance of, an evolving contingency funding plan.

Credit ratings by the three major credit rating agencies are an important component of the company's liquidity profile. Among other factors, the credit ratings are based on the financial strength, credit quality and concentrations in the loan portfolio, the level and volatility of earnings, capital adequacy, the quality of management, the liquidity of the balance sheet, the availability of a significant base of core retail and commercial deposits, and the company's ability to access a broad array of wholesale funding sources. Adverse changes in these factors could result in a negative change in credit ratings and impact not only the ability to raise funds in the capital markets, but also the cost of these funds. In addition, certain financial on- and off-balance sheet arrangements contain credit rating triggers that could increase funding needs if a negative rating change occurs. Letter of credit commitments for marketable securities, interest rate swap collateral agreements, and certain asset securitization transactions contain credit rating provisions.

As a result of the formal SEC investigation announced June 26, 2003, Standard and Poor's rating agency placed the company's debt ratings on "CreditWatch Negative". On April 15, 2004, Standard and Poor's removed the company's debt rating from "CreditWatch Negative" and revised the outlook to "Stable" from "Negative." Credit ratings are as follows:

| | Senior Unsecured Notes | Subordinated Notes | Short Term | Outlook |
|------------------------------------|------------------------------|-----------------------|---------------|----------|
| | | | | |
| Huntington Bancshares Incorporated | | | | |
| Moody's Investor Service | A2 | A3 | P1 | Negative |
| Standard and Poor's | A- | BBB+ | A2 | Stable |
| Fitch Ratings | А | A- | F1 | Stable |
| The Huntington National Bank | | | | |
| Moody's Investor Service | A1 | A2 | P1 | Negative |
| Standard and Poor's | А | A- | A1 | Stable |
| Fitch Ratings | А | A- | F1 | Stable |

Management believes that sufficient liquidity exists to meet the funding needs of the Bank and the parent company.

OFF-BALANCE SHEET ARRANGEMENTS

Like other financial organizations, Huntington has various commitments in the ordinary course of business that, under GAAP, are not recorded in the financial statements. Specifically, Huntington makes various commitments to extend credit to customers, to sell loans, and to maintain obligations under operating-type non-cancelable leases for its facilities. Derivatives and other off-balance sheet arrangements are discussed under the "Market Risk" section of the company's 2003 Form 10-K.

Standby letters of credit are conditional commitments issued to guarantee the performance of a customer to a third party. These guarantees are primarily issued to support public and private borrowing arrangements, including commercial paper, bond financing, and similar transactions. There were \$962 million of outstanding standby letters of credit at June 30, 2004. Non-interest income was recognized from the issuance of these standby letters of credit of \$2.8 million for the period ended June 30, 2004. The carrying amount of deferred revenue related to standby letters of credit at June 30, 2004, was \$3.2 million. Standby letters of credit are included in the determination of the amount of risk-based capital that the company and the Bank are required to hold.

CAPITAL

Capital is managed both at the parent and the Bank levels. Capital levels are maintained based on regulatory capital requirements and the economic capital required to support credit, market, and operation risks inherent in the company's business and to provide the flexibility needed for future growth and new business opportunities. Management places significant emphasis on the maintenance of a strong capital position, which promotes investor confidence, provides access to the national markets under favorable terms, and enhances business growth and acquisition opportunities. The importance of managing capital is also recognized, and Management continually strives to maintain an appropriate balance between capital adequacy and providing attractive returns to shareholders.

Shareholders' equity totaled \$2.4 billion at June 30, 2004. This balance represented a \$111 million increase from December 31, 2003. The growth in shareholders' equity resulted from the retention of net income after dividends to shareholders of \$134 million and stock option exercises of \$7 million. The growth was offset by a decrease in accumulated other comprehensive income primarily resulted from a decline in the market value of securities available for sale, partially offset by an increase in the market value of cash flow hedges at June 30, 2004, compared with December 31, 2003.

At June 30, 2004, the company had unused authority to repurchase up to 7.5 million shares. On September 4, 2001, options totaling 3.2 million shares were granted to all employees, the vesting of which occurs on September 4, 2006, or earlier should Huntington's common stock closing price for five consecutive trading days be at or

above \$25.00. Huntington's common stock price closed at \$25.07 on August 2, 2004, thus increasing the possibility that the remaining 2.2 million options outstanding at June 30, 2003, may vest in the near future. Should that occur, there is also the possibility that a portion of the current stock repurchase authorization would be used to help mitigate the dilutive earnings impact resulting from the issuance of these shares. These, and any additional purchases including the 2.5 million shares associated with the Unizan merger, will be made from time-to-time in the open market or through privately negotiated transactions depending on market conditions.

On July 16, 2004, the board of directors declared a quarterly cash dividend on its common stock of \$0.20 per common share, up 14.3% from the current quarterly dividend of \$0.175 per common share. The dividend is payable October 1, 2004, to shareholders of record on September 17, 2004.

Table 16 - Quarterly Common Stock Summary

| | 20 | 04 | | 2003 | |
|--|-----------|-----------|-----------|-----------|-----------|
| | Second | First | Fourth | Third | Second |
| Common Stock Price | | | | | |
| High ⁽¹⁾ | \$ 23.120 | \$ 23.780 | \$ 22.550 | \$ 20.890 | \$ 21.540 |
| Low ⁽¹⁾ | 20.890 | 21.000 | 19.850 | 19.220 | 18.030 |
| Close | 22.980 | 22.030 | 22.500 | 19.850 | 19.510 |
| Average closing price | 22.050 | 22.501 | 21.584 | 20.199 | 19.790 |
| Book value per share | \$ 10.40 | \$ 10.31 | \$ 9.93 | \$ 9.79 | \$ 9.63 |
| Dividends | | | | | |
| Cash dividends declared | \$ 0.175 | \$ 0.175 | \$ 0.175 | \$ 0.175 | \$ 0.160 |
| Common shares outstanding (000s) | | | | | |
| Average — Basic | 229,429 | 229,227 | 228,902 | 228,715 | 228,633 |
| Average — Diluted | 232,659 | 232,915 | 231,986 | 230,966 | 230,572 |
| Ending | 229,476 | 229,410 | 229,008 | 228,870 | 228,660 |
| Common Share Repurchase Program (000s) | | | | | |
| Number of Shares Repurchased | _ | _ | _ | _ | |

⁽¹⁾ High and low stock prices are intra-day quotes obtained from NASDAQ.

Average equity to average assets in the 2004 second quarter was 7.42%, down from 7.66% a year earlier, but up from 7.39% for the first quarter of 2004 (see Table 17). At June 30, 2004, the tangible equity to assets ratio was 6.95%, down from 7.06% a year ago, and from 6.97% March 31, 2004. The decline from the year-ago period primarily reflected the impact of the 2003 third quarter adoption of FIN 46, which resulted in the consolidation of \$1.0 billion of securitized automobile loans, partially offset by earnings-related growth in capital. The increase from March 31, 2004, primarily reflected growth in retained earnings.

Table 17 - Capital Adequacy

| | Three Months Ended | | | | | | | | | | |
|---|--------------------|------------------|----------------------|-----------------------|------------------|--|--|--|--|--|--|
| (in millions) | June 30, 2004 | March 31 2004 | December 31, 2003 | September 30, 2003 | June 30, 2003 | | | | | | |
| Total Risk-Adjusted Assets | \$ 28,413 | \$ 28,236 | \$ 28,164 | \$ 27,949 | \$ 27,456 | | | | | | |
| Tier 1 Risk-Based Capital Ratio | 9.00% | 8.74% | 8.53% | 8.40% | 8.32% | | | | | | |
| Total Risk-Based Capital Ratio Tier 1 Leverage Ratio | 12.58% 8.22% | 12.38% 8.08% | 11.95% 7.98% | 11.19% 7.94% | 11.11% 8.25% | | | | | | |
| Tangible Equity / Assets Ratio | 6.95% | 6.97% | 6.79% | 6.77% | 7.06% | | | | | | |
| Tangible Equity / Risk-Weighted Assets Ratio | 7.64% | 7.61% | 7.30% | 7.24% | 7.23% | | | | | | |
| Average equity / average assets | 7.42% | 7.39% | 7.32% | 7.49% | 7.66% | | | | | | |

At June 30, 2004, the tangible equity to risk-weighted assets ratio was 7.64%, up significantly from 7.23% in the year-ago quarter, and up from 7.61% at March 31, 2004. The increase in the tangible equity to risk-weighted assets ratio reflected primarily the positive impact resulting from reducing the overall risk profile of earning assets throughout this period, most notably a less risky loan portfolio mix, as well as growth in low risk investment securities.

The Federal Reserve Board, which supervises and regulates the company, sets minimum capital requirements for each of these regulatory capital ratios. In the calculation of these risk-based capital ratios, risk weightings are assigned to certain asset and off-balance sheet items such as interest rate swaps, loan commitments, and securitizations. Huntington's Tier 1 Risk-based Capital, Total Risk-based Capital, Tier 1 Leverage ratios, and risk-adjusted assets for the recent five quarters are well in excess of minimum levels established for "well capitalized" institutions of 6.00%, 10.00%, and 5.00%, respectively. At June 30, 2004, the company had regulatory capital ratios in excess of "well capitalized" regulatory minimums.

The Bank is primarily supervised and regulated by the Office of the Comptroller of the Currency, which establishes regulatory capital guidelines for banks similar to those established for bank holding companies by the Federal Reserve Board. At June 30, 2004, the Bank had regulatory capital ratios in excess of "well capitalized" regulatory minimums.

LINES OF BUSINESS DISCUSSION

Huntington has three distinct lines of business: Regional Banking, Dealer Sales, and the Private Financial Group (PFG). A fourth segment includes the company's Treasury functions and capital markets activities and other unallocated assets, liabilities, revenue, and expense. Lines of business results are determined based upon the company's management reporting system, which assigns balance sheet and income statement items to each of the business segments. A description of each segment and discussion of financial results is provided below.

Management uses earnings on an operating basis, rather than on a GAAP basis, to measure underlying performance trends for each business segment. Operating earnings represent GAAP earnings adjusted to exclude the impact of the significant items discussed in Note 8 to the Condensed Consolidated Financial Statements. Analyzing earnings on an operating basis is very helpful in assessing underlying performance trends, a critical factor used by Management to determine the success of strategies and future earnings capabilities.

Regional Banking

Regional Banking provides products and services to retail, business banking, and commercial customers. These products and services are offered in seven operating regions within the five states of Ohio, Michigan, West Virginia, Indiana, and Kentucky through the company's traditional banking network. Each region is further divided into Retail and Commercial Banking units. Retail products and services include home equity loans and lines of credit, first mortgage loans, direct installment loans, business loans, personal and business deposit products, as well as sales of investment and insurance services. Retail products and services 57% and 81% of total Regional Banking loans and deposits, respectively. These products and services are delivered to customers through banking offices, ATMs, Direct Bank—Huntington's customer service center, and Web Bank at huntington.com. Commercial banking serves middle-market and commercial banking relationships, which use a variety of banking products and services including, commercial loans, international trade, cash management, leasing, interest rate protection products, capital market alternatives, 401(k) plans, and mezzanine investment capabilities.

2004 Second Quarter versus 2003 Second Quarter

Regional banking contributed \$60.9 million of the company's net operating earnings in the second quarter of 2004, up \$35.9 million from the second quarter of 2003. This increase primarily reflected a \$44.5 million reduction in provision for credit losses. Also benefiting comparisons with the year-ago quarter was a \$15.3 million, or 7%, growth in revenue, reflecting a \$4.7 million, or 3%, increase in net interest income and \$10.7 million, or 15%, increase in non-interest income. The benefits of lower provision for credit losses and higher revenue from the year-ago quarter were partially offset by expense growth of \$4.5 million, or 3%. The ROA in the 2004 second quarter was 1.52%, up from 0.69% in the year-ago quarter. The ROE was 23.9% in the current quarter, up from 9.8% in the 2003 second quarter.

Compared with the year-ago quarter, 2004 second quarter net interest income increased \$4.7 million, or 3%, reflecting a 14% increase in average total loans and 5% increase in average total deposits, partially offset by a 28 basis point decline in net interest margin to 4.15% from 4.43%.

Average total loans increased \$1.8 billion, or 14%, primarily reflecting a \$1.0 billion, or 66%, increase in average residential mortgages, a \$0.7 billion, or 24%, increase in home equity loans and lines of credit, as well as a \$0.3 billion, or 10%, increase in average CRE loans. The growth in home equity, residential mortgages, and CRE loans reflected the favorable impact of low interest rates on demand for real estate-related financing. In addition, the launch of a new Consumer First Mortgage, targeted at high convenience, low loan-to-value ratio borrowers, accounted for \$0.3 billion of the \$0.7 billion increase in average home equity loans and lines of credit. Total average C&I loans declined \$0.3 billion, or 7%, from the year-ago quarter. The decline in C&I loans was due in part to weak demand, as well as the impact from continued strategies to lower exposure to large individual commercial credits, including shared national credits. Small Business C&I and CRE loans (included in total average C&I and CRE loans) increased \$0.2 billion, or 11%, due to specific strategies that focus on this business segment.

Average total deposits increased \$0.8 billion, or 5%. This reflected strong growth in average interest bearing demand deposits, up \$1.0 billion, or 19%, and a \$0.1 billion, or 5%, increase in non-interest bearing deposits, which was partially offset by a \$0.5 billion, or 12%, decline in domestic time deposits. Of the \$0.8 billion increase in average total deposits, Corporate Banking accounted for \$0.6 billion and Small Business \$0.4 billion, with average total deposits in Retail Banking declining \$0.1 billion.

The company continued to focus on customer service and delivery channel optimization. From the year-ago quarter, eight banking offices were opened while nine were closed. Progress was made in improving the Retail Banking 90-day cross sell ratio, from 1.9 products or services to 2.2, a 19% improvement. Further, the online banking penetration of retail households with on-line banking increased to 35% from 26% a year earlier, with a 37% increase in the number of online customers.

The 28 basis points, or an effective 6%, decline in the net interest margin to 4.15% from 4.43% reflected a combination of factors. This included a shift in the loan portfolio mix to lower-margin, but higher credit quality, consumer residential real estate-related loans. In addition, interest rates offered on deposits have been near historical lows throughout this period, and as rates continued to fall from a year-ago, it was increasingly difficult to make commensurate reductions in deposit rates compared with reductions in loan yields.

Provision for credit losses for the second quarter of 2004 represented a credit of \$3.9 million, or \$44.5 million less than in the year-ago quarter. This decline reflected the overall improvement in credit quality, as well as the shift to lower-rate, lower-risk residential mortgages and home equity loans and lines. Net charge-offs were only \$1.8 million, or an annualized 0.05% of average loans and leases, down from \$31.6 million, or 0.97%, in the year-ago quarter. The 2004 second quarter net charge-offs were reduced by a \$9.7 million C&I recovery on a single credit charged-off in the 2002 fourth quarter (see Credit Risk for additional discussion regarding charge-offs and allowance for loan loss reserve methodologies).

Non-interest income increased \$10.7 million, or 15%, from the year-ago quarter, reflecting a combination of factors including higher mortgage banking income, an increase in deposit service charges, growth in brokerage and insurance income, and an increase in other income, partially offset by a decline in other service charges and fees.

Mortgage banking income increased \$6.2 million, or 89%, from the year-ago quarter largely as a result of a change in reporting methodology. In 2004, MSR impairment and recovery is reflected in the Treasury/Other segment, whereas in the year-ago quarter MSR impairment of \$6.4 million was reflected in Regional Banking. Deposit service charges increased \$2.7 million, or 7%, reflecting higher personal NSF and overdraft fees. The \$1.6 million, or 16%, increase in other income was due primarily to the full year impact of the 2003 adoption of FIN 45 for standby letters of credit, which recognizes fees over the life of the related guarantee, rather than at the time of quarterly or annual billings. Brokerage and insurance income increased \$0.7 million, or 18%, due to higher credit protection insurance fees. The \$0.7 million, or 7%, decline in other service charges and fees reflected lower interchange fees on debit card transactions.

Non-interest expense increased \$4.5 million, or 3%, from the year-ago quarter. Higher personnel costs contributed \$2.0 million to this increase due to an increase in the number of employees, as well as higher benefit costs. The \$2.3 million, or 3%, increase in other expenses reflected higher occupancy, printing and supplies, depreciation, and outside servicing expenses, partially offset by lower marketing, telecommunications, legal and professional, and transportation expenses.

2004 Second Quarter versus 2004 First Quarter

Regional Banking earnings in the 2004 second quarter increased \$12.8 million, or 27%, from the 2004 first quarter. This reflected a \$14.4 million, or 6%, increase in revenue as non-interest income increased \$10.4 million, or 14%, and net interest income increased \$4.0 million, or 3%. Also contributing to the increase in earnings was a \$6.1 million reduction in provision for credit losses. The benefits of higher revenue and lower provision for credit losses were partially offset by a slight increase in non-interest expense. The ROA in the 2004 second quarter was 1.52%, up from 1.27% in the 2004 first quarter, with the ROE of 23.9%, up from 19.2% in the prior quarter.

Net interest income increased \$4.0 million, or 3%, from the prior quarter, reflecting growth in average total loans and deposits, partially offset by a decline in the net interest margin to 4.15% from 4.25%.

Average total loans increased \$0.7 billion, or 5%. Consumer loans increased 10%, reflecting strong growth in residential mortgages and home equity loans and lines of credit. Average C&I loans increased 2% with CRE loans increasing 1%. The growth in average C&I loans is encouraging and may reflect the positive impact of a recovering economy. Total average deposits increased \$0.7 billion, or 5%, reflecting strong growth in interest bearing and non-interest bearing demand deposits, up 10% and 7%, respectively.

From the end of the 2004 first quarter, the number of demand deposit account (DDA) households increased 1%, and the 90-day cross sell ratio of 2.2 products was unchanged. On-line banking penetration of retail households increased to 35% from 34%, and the number of active online users increased 5%.

The \$6.1 million decline in provision for credit losses from the first quarter reflected the combination of improved credit quality performance and improved economic outlook, as well as the continued lowering of the overall credit risk profile of the loan portfolio through the growth in residential real estate-related loans. Net charge offs were \$1.8 million, or an annualized 0.05% of average loans and leases in the current quarter, down from \$11.6 million, or 0.33%, in the first quarter, primarily attributable to the \$9.7 million C&I recovery in the current quarter.

Non-interest expense increased \$0.7 million from the first quarter of 2004. This increase was largely due to a \$1.9 million increase in other expenses, as personnel expense declined \$1.4 million. The increase in other expenses reflected higher other real estate owned expenses, low income housing investment amortization, printing and supplies costs, and operating losses. Personnel expense declined due to higher FASB 91 deferred salary costs associated with higher loan production.

2004 First Six Months versus 2003 First Six Months

Regional banking contributed \$108.9 million of the company's net operating earnings in the first six months of 2004, up \$48.8 million from the comparable year-ago period. This increase primarily reflected a \$65.9 million reduction in provision for credit losses. Also benefiting comparisons with the year-ago period was a \$20.3 million, or 5%, growth in revenue, reflecting a \$9.3 million, or 3%, increase in net interest income and \$11.1 million, or 8%, increase in non-interest income. The benefits of lower provision for credit losses and higher revenue from the year-ago period were partially offset by expense growth of \$11.3 million, or 4%. The ROA in the first six months of 2004 was 1.39%, up from 0.85% in the year-ago period. The ROE was 21.5% in the first six months, up from 12.1% in the comparable 2003 period.

Net interest income in the first six months of 2004 increased \$9.3 million, or 3%, reflecting a 12% increase in average total loans and 4% increase in average total deposits, partially offset by a 27 basis point decline in net interest margin to 4.19% from 4.45%.

Average total loans increased \$1.5 billion, or 12%, primarily reflecting a \$0.8 billion, or 57%, increase in average residential mortgages, a \$0.7 billion, or 22%, increase home equity loans and lines of credit, as well as a \$0.4 billion, or 11%, increase in average CRE loans. Total average C&I loans declined \$0.4 million, or 8%, from the year-ago six-month period. The growth in home equity, residential mortgages, and CRE loans, as well as the decline in C&I loans reflected the same factors noted above in the year-ago quarter. Small Business C&I and CRE loans (included in total average C&I and CRE loans) increased \$0.2 billion, or 11%, due to specific strategies that focus on this business segment.

Average total deposits increased \$0.6 billion, or 4%. This reflected strong growth in average interest bearing demand deposits, up \$0.9 billion, or 18%, partially offset by a \$0.5 billion, or 13%, decline in domestic time deposits. Of the \$0.6 billion increase in average total deposits, Corporate Banking accounted for \$0.6 billion and Small Business \$0.3 billion, with average total Consumer deposits declining \$0.3 billion.

The 27 basis points, or an effective 6%, decline in the net interest margin to 4.19% from 4.45% reflected the same factors discussed above in the 2004 second quarter versus 2003 second quarter performance.

Provision for credit losses for the first six months of 2004 represented a credit of \$1.8 million, or \$65.9 million less than in the year-ago period. This decline reflected the overall improvement in credit quality, as well as the shift to lower risk residential mortgages and home equity loans and lines. Net charge-offs were only \$13.4 million, or an annualized 0.19% of average loans and leases, down from \$51.9 million, or 0.81% in the comparable year-ago period. The 2004 first-half net charge-offs were reduced by a \$9.7 million C&I recovery (see Credit Risk for additional discussion regarding charge-offs and allowance for loan loss reserve methodologies).

Non-interest income for the first six months of 2004 increased \$11.1 million, or 8%, from the comparable year-ago period, reflecting a combination of factors including higher other income, an increase in deposit service charges, and higher mortgage banking revenue, partially offset by a decline in other service charges and fees.

The \$5.8 million, or 34%, increase in other income from the comparable year-ago six-month period was due primarily to the full year impact of the FIN 45 adjustment discussed above, and \$2.3 million of other fees. Deposit service charges increased \$4.7 million, or 6%, reflecting higher personal NSF and overdraft fees. Mortgage banking income increased \$1.3 million, or 7%, from the year-ago period largely as a result of a change in reporting methodology. In 2004, MSR impairment and recovery is reflected in the Treasury/Other segment, whereas in the comparable year-ago six month period MSR impairment of \$6.4 million was reflected in Regional Banking. This was partially offset by changes in origination income, marketing income and MSR amortization, as a result of changes in the interest rate environment. The \$1.6 million, or 7%, decline in other service charges and fees reflected lower interchange fees on debit card transactions.

Non-interest expense for the first six months of 2004 increased \$11.3 million, or 4%, from the year-ago period. Higher personnel costs contributed \$5.7 million to this increase due to a 3% increase in the number of employees, as well as higher benefit costs. The \$5.3 million, or 3%, increase in other expenses reflected higher marketing, depreciation, printing and supplies, and charge card processing expenses, partially offset by lower telecommunication, legal and professional, and transportation expenses.

Table 18 - Regional Banking⁽¹⁾

| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$ | Table 18 - Regional Banking ⁽¹⁾ | 2004 | | 2004 | | 2003 | 2Q04 vs. 2 | 2Q03 | 2004 | | 2003 | 1H04 vs. 1 | H03 |
|---|--|-------------|----|---------------------------------------|----|---------------------------------------|-------------|--------|---------------------------------------|----|-----------|---------------------------------------|------------|
| Interest freemer Before Securities Calls Data 155,083 (1,099) 151,062 (1,099) 152,083 (1,099) 151,062 (1,099) 152,083 (1,099) 152,083 (1,109) 152,083 (1,10 | | Second | _ | First | | Second | Amount | % | 6 Months | | 6 Months | Amount | % |
| Net Intervel Income \$155,003 \$150,002 \$150,003 \$11,117 \$150,003 \$11,117 \$150,003 \$11,117 \$150,003 \$11,117 \$150,003 \$11,117 \$150,003 \$11,113 \$150,003 \$11,113 | · | | | | | | | | | | | | |
| Phonishion for could losses (3,949) 2,105 40.525 (44,474) NM (1,544) 64,078 (65,922) NM Net Interest Income After Provision for Credit Losses 159,032 144,957 (109,993) 49,139 44,7% 307,989 222,754 75,235 32,25 Operating lasse income 327 49 — 327 NM 376 — 376 — 376 — 376 — 376 — 376 — 376 MM 376 — 376 MM 376 — 376 MM 376 — 376 6.48 6.75 137.96 148.95 6.75 148.96 6.75 148.96 177.99 16.44% 23.060 177.187 5.811 33.36 11.137 7.85 7.95 7.90 10.085 14.9% 154.526 143.39 11.137 7.97 9.068 10.073 13.3 144.348 141.359 11.137 7.97 9.068 10.065 12.9% | <i>,</i> | \$ 155 092 | ¢ | 151.062 | ¢ | 150 / 19 | \$ 1665 | 3 10/2 | \$ 306 145 | ¢ | 296 822 | \$ 0.212 | 2 10/ |
| for Call Loses 159,022 148,857 109,893 41,78 307,089 222,74 72,255 32,35 Operating lease income 327 49 327 NM 376 376 NM Struce charges not posed accounts 235 40,010 39,070 2,650 NM Struce charges not posed 376 NM Struce charges not posed 376 NM Struce charges not posed 376 376 NM Struce charges not posed 376 376 376 376 376 NM Struce charges not posed 378 (31) -57 376 | | . , | ф | .) | φ | , | . , | | | ¢ | | , ., | |
| Operating lease income 327 49 537 NM 376 176 NM Service darges on deposit accounts 42.57 40.70 9.700 2.639 6.75 83,00 73.21 4.488 6.00 Service darges on deposit accounts 42.57 3.20 6.33 6.997 6.230 80.04 19.422 21.498 6.01 Other service darges and fees 10.225 (1.105 9.700 1.494 23.000 17.941 1.319 7.4 Other service darges and fees 10.225 11.105 9.700 1.645 14.994 154.526 143.389 11.137 7.88 Securities gams - NM NM 319 NM Total Non-interest lacome 82.475 72.051 71.709 10.685 14.994 154.556 143.389 11.137 7.898 Total Non-interest lacome 82.475 72.051 71.709 10.685 14.99 | Net Interest Income After Provision | | _ | | _ | | | | | | | | · <u> </u> |
| Schole charge on deposit acounts 42.97 40,003 39,707 2,650 6.7% 82,000 78,371 7,489 6.09 Biolengage building 4.515 3.585 3.542 6.75 82,001 75,95 83,01 7,348 6,73 7,843 5,28 6,77 Ottor struce charge and frees 11,295 91,024 17,197 5,813 3,78 7,843 5,81 3,78 7,843 6,78 3,78 7,817 4,88 6,77 5,813 3,78 7,848 6,71 3,78 7,848 6,71 3,389 17,179 10,685 14,975 15,256 14,339 11,137 7,88 Securitie gains | for Credit Losses | 159,032 | | 148,957 | _ | 109,893 | 49,139 | 44.7% | 307,989 | _ | 232,754 | 75,235 | 32.3% |
| Brokenge and insurance income 44.15 3.856 3.842 67.3 17.3% 8.371 7.443 5.28 6.77 Moragge bunking 13.227 6.033 6.997 6.230 80.0% 19.262 21.139 7.43 13.19 7.43 Other verve charges and fees 11.295 7.10 9.403 1.1295 7.00 9.042 21.499 (1.357) 7.27 Other verve charges and fees 11.295 71.090 10.685 14.9% 154.556 143.389 11.137 7.88 Securities gains — — — M.M — — M.M From locats 61.728 63.156 59.76 14.9% 154.556 143.59 11.137 7.88 Operating lease expense 12.75 14.4 — 22.78 10.797 104.430 5.901 4.59 Operating lease expense 17.34 63.15 5.535 2.27 10.972 10.43.99 4.92 Tass 53.67 73 | | | | | | | | | | | | | |
| Trans scrives. 225 220 280 (15) 19.0% 517 548 (10) 5.7.7 Other scrives charges and fees 10.529 9.413 11.295 (730) 6.5% 19.200 17.04 1.310 7.49 Other scrives charges and fees 10.257 7.2051 7.709 10.685 14.9% 154.526 143.890 11.137 7.88 Securities gains — — — — MM MM< | | · · · · · · | | | | | | | , | | | | |
| | 6 | , | | | | · · · · · · · · · · · · · · · · · · · | | | , | | | | |
| Other 11,205 11,705 9,705 1.590 16.4% 23,000 17,187 5,813 33,89 Total Non-Interest Income Before Securities Gains 82,475 72,051 71,790 10,685 14,9% 154,526 143,389 11,137 7,88 Securities gains - - - - - - NM - - - NM Total Non-Interest Income 82,475 72,051 71,790 10,685 14,9% 154,526 143,389 11,137 7,88 Order 85,887 33,525 32,779 16,478 109 - 319 - 13,99 4.07 Total Non-Interest Expense 147,690 147,092 143,319 4.521 3.2% 20,473 28,363 11,399 4.07 Total Non-Interest Expense 147,667 73,374 25,301 NM 167,583 92,520 75,063 81,19 Net Income Fore Provision for Income Taxe exprision for Income Taxe expret Provision for Income Taxe expret income 13,0 | | | | | | | | | | | | . , | 7.4% |
| Total Non-Interest Income Before Securities Gains 82,475 72,051 71,709 10,685 14,9% 154,526 143,389 11,137 7.88 Total Non-Interest Income 82,475 72,051 71,709 10,685 14,9% 154,526 143,389 11,137 7.88 Total Non-Interest Income 82,475 72,051 71,709 10,685 14,9% 154,526 143,389 11,137 7.88 Presoning loss: synence 275 44 - 275 NM 10 310 NM Presoning loss: synence 82,857 83,892 83,558 2,279 12,484 119,93 5,691 44,80 5,299 3,29 144,400 5,299 3,28 2,5531 11,342 19,356 NM 167,583 92,520 75,063 81,19 Provision for income Taxes 93,667 73,916 38,364 55,003 NM 167,583 92,520 75,063 81,19 Net Income - Operating '' \$60,854 \$48,045 \$24,937 \$ | Other service charges and fees | 10,529 | | 9,413 | | 11,259 | (730) | -6.5% | 19,942 | | 21,499 | (1,557) | -7.2% |
| Before Securities Gains 82.475 72.051 71.790 10.685 14.9% 154.526 14.3389 11.137 7.89 Total Non-Interest Income 82.475 72.051 71.790 10.685 14.9% 154.526 143.389 11.137 7.89 Operating lease expense 275 44 - 275 NM 319 - 319 NM Deter 85.837 83.958 2.279 2.240.32 2.8.20 11.99 4.69 4.69 4.64 3.19 14.39 14.19 14.39 14.19 14.39 14.19 14.39 14.19 14.39 14.19 <td>Other</td> <td>11,295</td> <td></td> <td>11,705</td> <td>_</td> <td>9,705</td> <td>1,590</td> <td>16.4%</td> <td>23,000</td> <td></td> <td>17,187</td> <td>5,813</td> <td>33.8%</td> | Other | 11,295 | | 11,705 | _ | 9,705 | 1,590 | 16.4% | 23,000 | | 17,187 | 5,813 | 33.8% |
| Securities gains - - - - NM - - - NM Total Non-Interest Income 82,475 72,051 71,790 10,685 14,9% 154,526 143,389 11,137 7.89 Operating lease expense 275 64,1 - 275 NM 319 - 319 NM Presonnel costs 61,725 63,156 9,761 1,967 3,3554 2,239 164,430 5,299 3,29 Total Non-Interest Expense 147,840 147,002 143,319 4,521 3,2% 294,992 283,623 11,399 4,69 Income Hofor Provision for Income Taxes 93,667 73,916 38,364 55,303 NM 167,583 92,520 75,063 81,19 Net Income Operating (h) \$ 60,884 \$ 48,045 \$ 24,937 \$ 35,947 NM \$ 108,929 \$ 60,138 \$ 48,791 81,19 Revence - Fully Taxable Equivalent (FTE) 933,33 151,311 150,799 4,6465 3,1 | Total Non-Interest Income | | | | | | | | | | | | |
| Total Non-Interest Income 82,475 72,051 71,790 10,685 14.0% 154,526 143,389 11,137 7.89 Operating lease expense 275 44 - 275 NM 319 - 319 NM Operating lease expense 61,728 63,156 99,701 1.967 3.3% 12,4844 110,103 5,091 4.89 Other 85,837 83,836 2.279 106,728 106,430 5,299 3.22 11,309 4.09 Income Edefor Provision for Income 173,840 147,092 143,319 4,521 3.276 104,932 2.83,623 11,309 4.09 Net increst income taxes ⁽²⁾ 32,67 73,916 38,364 55,303 NM 107,853 92,520 75,063 81,19 Net increst income taxes ⁽²⁾ 32,673 24,095 5,5947 NM 51,062,85 9,113 3,19 Net increst income (FTE) 152,333 15,1042 \$ 150,418 \$ 4,665 3,1% \$ 30,6145 | Before Securities Gains | 82,475 | | 72,051 | | 71,790 | 10,685 | 14.9% | 154,526 | | 143,389 | 11,137 | 7.8% |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$ | Securities gains | | | — | | — | _ | NM | | | — | _ | NM |
| Personal costs 61,728 63,155 99,761 1.967 3.3% 124,884 119,193 5,501 4.89 Other 88,887 83,802 83,558 2,279 2,7% 109,729 164,400 5,209 3,22 Total Non-Interest Expense 147,840 147,002 143,310 4,521 3,2% 204,032 283,623 11,300 4,00 Total Non-Interest Expense 147,840 147,002 143,310 4,521 3,2% 204,032 283,623 11,300 4,00 Total Non-Interest Expense 147,840 144,002 143,310 4,521 3,2% 204,032 28,623 11,300 4,00 Total Non-Interest Expense 100,000 32,382 2,520 7,50,63 81,19 Provision for income taxes (*) 33,64 \$ 4,8045 \$ 24,937 \$ 35,947 NM \$ 108,929 \$ 60,138 \$ 4,87,91 81,19 Revenue - Fully Taxable Equivalent If Taxable Equivalent 113,201 113,21 113,21 110,20 14,521 </td <td>Total Non-Interest Income</td> <td>82,475</td> <td></td> <td>72,051</td> <td></td> <td>71,790</td> <td>10,685</td> <td>14.9%</td> <td>154,526</td> <td></td> <td>143,389</td> <td>11,137</td> <td>7.8%</td> | Total Non-Interest Income | 82,475 | | 72,051 | | 71,790 | 10,685 | 14.9% | 154,526 | | 143,389 | 11,137 | 7.8% |
| Personal costs 61,728 63,155 99,761 1.967 3.3% 124,884 119,193 5,501 4.89 Other 88,887 83,802 83,558 2,279 2,7% 109,729 164,400 5,209 3,22 Total Non-Interest Expense 147,840 147,002 143,310 4,521 3,2% 204,032 283,623 11,300 4,00 Total Non-Interest Expense 147,840 147,002 143,310 4,521 3,2% 204,032 283,623 11,300 4,00 Total Non-Interest Expense 147,840 144,002 143,310 4,521 3,2% 204,032 28,623 11,300 4,00 Total Non-Interest Expense 100,000 32,382 2,520 7,50,63 81,19 Provision for income taxes (*) 33,64 \$ 4,8045 \$ 24,937 \$ 35,947 NM \$ 108,929 \$ 60,138 \$ 4,87,91 81,19 Revenue - Fully Taxable Equivalent If Taxable Equivalent 113,201 113,21 113,21 110,20 14,521 </td <td>Operating lease expense</td> <td>275</td> <td>_</td> <td>44</td> <td>_</td> <td></td> <td>275</td> <td>NM</td> <td>319</td> <td>-</td> <td></td> <td>319</td> <td>NM</td> | Operating lease expense | 275 | _ | 44 | _ | | 275 | NM | 319 | - | | 319 | NM |
| Other 85,837 83,802 83,558 2,279 2,7% 160,729 164,430 5,209 3,29 Total Non-Interest Expense 147,840 147,092 143,319 4,521 3,2% 294,932 283,663 11,309 4,00 Income Before Provision for Income Taxes 93,667 73,916 38,364 55,303 NM 167,583 92,520 75,063 81,19 Provision for income taxes ¹⁰ 32,783 25,871 13,427 19,356 NM \$66,64 32,382 26,272 81,19 Net increst income \$155,083 \$151,062 \$150,418 \$4,665 3,1% \$106,145 \$26,0632 \$9,313 311 Tax equivalent adjustment ⁽²⁾ 250 233,62 \$222,519 \$154,083 \$151,062 \$150,043 \$46,071 \$40,664 297,474 9,170 3,19 Yea taiterest income (FTE) 1553,33 151,311 150,729 4,604 3,1% \$440,863 \$20,308 4,60 Stat Revenue Excluding Securities \$23 | | | | | | | | | | | | | 4.8% |
| Income Before Provision for Income Taxes 93,667 73,916 38,364 55,303 NM 167,583 92,520 75,063 81,19 Provision for income taxes © 32,783 25,871 13,427 19,356 NM 58,654 32,382 26,272 81,19 Net Income - Fully Taxable Equivalent (FTE) 5 60,884 \$48,045 \$24,937 \$35,947 NM \$108,929 \$60,138 \$48,791 81,19 Net incerest income \$155,083 \$15,1062 \$150,418 \$4,665 3,1% \$306,145 \$296,832 \$9,313 3,19 Tax equivalent digitament ¹⁰ 250 249 311 (61) -19,6% 499 642 (14)<-223 | | , | | | | | | | , | | | | 3.2% |
| Taxes 93,667 73,916 38,364 55,303 NM 167,583 92,520 75,063 81,19 Provision for income taxes © 32,783 25,871 13,427 19,356 NM 58,654 32,382 26,272 81,19 Net Income - Operating(¹⁰) \$60,884 \$48,045 \$24,937 \$35,947 NM \$108,929 \$60,138 \$48,791 81,19 Revenue - Fully Taxable Equivalent (FTE) 55,083 \$15,1062 \$150,418 \$4,665 3,1% \$306,145 \$296,832 \$9,313 3,19 Tax equivalent Ligitisment ¹⁰ 250 249 311 (61) -19,6% 499 642 (143)<-223,93 | Total Non-Interest Expense | 147,840 | _ | 147,092 | _ | 143,319 | 4,521 | 3.2% | 294,932 | _ | 283,623 | 11,309 | 4.0% |
| Taxes 93,667 73,916 38,364 55,303 NM 167,583 92,520 75,063 81,19 Provision for income taxes ^(D) 32,783 25,871 13,427 19,356 NM 58,654 32,382 26,272 81,19 Net Income - Operating ⁽¹⁾ S 60,884 \$ 48,045 \$ 24,937 \$ 35,947 NM \$108,929 \$ 60,138 \$ 48,791 81,19 Revenue - Fully Taxable Equivalent (FTE) 55,083 \$ 151,062 \$ 150,418 \$ 4,664 3.1% \$ 306,644 297,474 9,170 3.19 Net interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.89 Non-interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.89 Total Revenue Excluding Securities 5 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 </td <td>Income Before Provision for Income</td> <td></td> <td></td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td></td> <td></td> <td></td> | Income Before Provision for Income | | | | - | | | | | - | | | |
| Provision for income taxes ⁽¹⁾ 32,783 25,871 13,427 19,356 NM 58,654 32,382 26,272 81,19 Net Income - Operating ⁽¹⁾ \$ 60,884 \$ 48,045 \$ 24,937 \$ 35,947 NM \$ 108,929 \$ 60,138 \$ 48,791 81,19 Revenue - Fully Taxable Equivalent (FTE) NM \$ 108,929 \$ 60,138 \$ 48,791 81,19 Revenue - Fully Taxable Equivalent (FTE) \$ 155,083 \$ 151,002 \$ 150,418 \$ 4,665 3.1% \$ 306,145 \$ 296,832 \$ 9,313 3.1% Net interest income \$ 125,083 \$ 151,011 150,729 4,664 3.1% 306,644 297,474 9,170 318 Non-interest income \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 Total Revenue Excluding Securities Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 SELECTED AVERAGE BALANCES (in millions) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 < | | 93,667 | | 73,916 | | 38,364 | 55,303 | NM | 167,583 | | 92,520 | 75,063 | 81.1% |
| Revenue - Fully Taxable Equivalent (FTE) S 155,083 S 151,062 S 150,418 S 4,665 3.1% S 306,145 S 296,832 S 9,313 3.19 Net interest income S 155,083 S 151,062 S 150,418 S 4,665 3.1% S 306,145 S 296,832 S 9,313 3.19 Tax equivalent adjustment' 250 249 311 (61) -19.6% 499 642 (143) -22.39 Net interest income 82,476 72.051 71,790 10,686 14.9% 154,527 143,389 11,138 7.89 Total Revenue (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 Total Revenue Excluding Securities Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 SELECTED AVERAGE BALANCES (in millions) Lams: C& CR CR 2,567 2,315 256 1.0% 2,578 2,300 | Provision for income taxes (2) | 32,783 | | · · · · · · · · · · · · · · · · · · · | | · · · · · · · · · · · · · · · · · · · | · · · · · · | NM | · · · · · · · · · · · · · · · · · · · | | · · · · · | · · · · · · · · · · · · · · · · · · · | 81.1% |
| (FTE) S 515,083 S 151,062 S 150,012 S 311 (61) -19.6% 499 642 9,313 3,1% Tax equivalent adjustment ^(D) 250 249 311 (61) -19.6% 499 642 (143) -22.3% Net interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Total Revenue (FTE) 5237,809 S 223,362 S 222,519 S 15,290 6.9% 5461,171 S 440,863 S 20,308 4.69 SELECTED AVERACE gains (FTE) 5237,809 S 223,362 S 222,519 S 15,290 6.9% 5461,171 S 440,863 S 20,308 4.69 SELECTED AVERACE gains (FTE) S 4,287 S 4,290 S (303) -6.6% 5 4,631 S (378) -8.29 Conturecial 2,591 </td <td>Net Income - Operating⁽¹⁾</td> <td>\$ 60,884</td> <td>\$</td> <td>48,045</td> <td>\$</td> <td>24,937</td> <td>\$ 35,947</td> <td>NM</td> <td>\$ 108,929</td> <td>\$</td> <td>60,138</td> <td>\$ 48,791</td> <td>81.1%</td> | Net Income - Operating ⁽¹⁾ | \$ 60,884 | \$ | 48,045 | \$ | 24,937 | \$ 35,947 | NM | \$ 108,929 | \$ | 60,138 | \$ 48,791 | 81.1% |
| (FTE) S 515,083 S 151,062 S 150,012 S 311 (61) -19.6% 499 642 9,313 3,1% Tax equivalent adjustment ^(D) 250 249 311 (61) -19.6% 499 642 (143) -22.3% Net interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Total Revenue (FTE) 5237,809 S 223,362 S 222,519 S 15,290 6.9% 5461,171 S 440,863 S 20,308 4.69 SELECTED AVERACE gains (FTE) 5237,809 S 223,362 S 222,519 S 15,290 6.9% 5461,171 S 440,863 S 20,308 4.69 SELECTED AVERACE gains (FTE) S 4,287 S 4,290 S (303) -6.6% 5 4,631 S (378) -8.29 Conturecial 2,591 </td <td>Revenue - Fully Taxable Equivalent</td> <td></td> <td>-</td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td></td> <td>_</td> <td></td> <td></td> <td></td> | Revenue - Fully Taxable Equivalent | | - | | - | | | | | _ | | | |
| Tax equivalent adjustment ⁽²⁾ 250 249 311 (61) -19.6% 499 642 (143) -22.3% Net interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Non-interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Total Revenue (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% Total Revenue Excluding Securities Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% SELECTED AVERAGE BALANCES (in millions) Loans: \$ 223,562 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% Cost CEE \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% CRE CALINCES (in millions) Loans: \$ 2,597 \$ 2,2567 2,335 2,66 | (FTE) | ¢ 155 002 | ¢ | 151.062 | ¢ | 150 419 | ¢ 1665 | 2 10/ | £ 206 145 | ¢ | 206 822 | ¢ 0.212 | 2 10/ |
| Non-interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Total Revenue (FTE) 5.237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% Total Revenue Excluding Securities Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% SELECTED AVERAGE BALANCES (in millions) 5 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 42,253 \$ 4,631 \$ (378) -8.2% Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.3% Construction 1,297 2,567 2,335 256 11.0% 2,300 27.8% 12,103 Construction 1,297 3,169 3,109< | | . , | \$ | , | \$ | , | . , | | . , | Э | | . , | -22.3% |
| Non-interest income 82,476 72,051 71,790 10,686 14.9% 154,527 143,389 11,138 7.8% Total Revenue (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% Total Revenue Excluding Securities Gains (FTE) \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% SELECTED AVERAGE BALANCES (in millions) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% Cell \$ 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 4,253 \$ 4,631 \$ (378) -8.2% Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.3% Construction 1,297 2,567 2,335 256 11.0% 2,370 3,053 667 2.138 1,480 838 56.6% Mome equity loans & lines of credit | Net interest income (FTE) | 155.333 | | 151 311 | _ | 150 729 | 4 604 | 3.1% | 306.644 | - | 297 474 | 9 1 7 0 | 3 1% |
| Total Revenue Excluding Securities Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.69 SELECTED AVERAGE BALANCES (in millions) Loans: 5 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 4,253 \$ 4,631 \$ (378) -8.29 CRE Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.39 Construction 1,297 2,567 2,335 2.56 11.0% 2,578 2,300 2.78 12.19 Consumer 6 credit 5 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines 3,848 3,593 3,109 739 23.8% 3,720 3,053 66.7 2,18 Other loans 317 2.78 315 2 0.6% 297 320 (23) -7.29 Total Consumer 6.644 6.039 4.920 1 | | , | | | | · · · · · · · · · · · · · · · · · · · | / | | , | | | · · · · · · | 7.8% |
| Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% SELECTED AVERAGE BALANCES (in millions) Summer is an intervent of the second secon | Total Revenue (FTE) | \$ 237,809 | \$ | 223,362 | \$ | 222,519 | \$ 15,290 | 6.9% | \$ 461,171 | \$ | 440,863 | \$ 20,308 | 4.6% |
| Gains (FTE) \$ 237,809 \$ 223,362 \$ 222,519 \$ 15,290 6.9% \$ 461,171 \$ 440,863 \$ 20,308 4.6% SELECTED AVERAGE BALANCES (in millions) Summer is an intervent of the second secon | Total Revenue Excluding Securities | | - | | - | | | | | _ | | | |
| BALANCES (in millions) Loans: C&I \$ 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 4,253 \$ 4,631 \$ (378) -8.2% CRE Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.3% Construction 1,297 2,567 2,335 256 11.0% 2,578 2,300 278 1219 Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Other loans 317 278 315 2 0.6% 297 320 (23) -7.29 Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.59 Total Loans \$ 14,819 <td></td> <td>\$ 237,809</td> <td>\$</td> <td>223,362</td> <td>\$</td> <td>222,519</td> <td>\$ 15,290</td> <td>6.9%</td> <td>\$ 461,171</td> <td>\$</td> <td>440,863</td> <td>\$ 20,308</td> <td>4.6%</td> | | \$ 237,809 | \$ | 223,362 | \$ | 222,519 | \$ 15,290 | 6.9% | \$ 461,171 | \$ | 440,863 | \$ 20,308 | 4.6% |
| Loans: C&I \$ 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 4,253 \$ 4,631 \$ (378) -8.29 CRE Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.39 Commercial 2,591 2,567 2,335 256 11.0% 2,578 2,300 278 12.19 Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% of credit 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Loans </td <td></td> | | | | | | | | | | | | | |
| C&I \$ 4,287 \$ 4,219 \$ 4,590 \$ (303) -6.6% \$ 4,253 \$ 4,631 \$ (378) -8.29 CRE Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.39 Commercial 2,591 2,567 2,335 256 11.0% 2,578 2,300 278 12.19 Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines of credit 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Other loans 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 838 56.6% Deposits: 7 14,115 13,056 1,763 13.5% 14,463 12,962 1,501 11.6% Domestic time deposits 2,982 2,783 2,85 | | | | | | | | | | | | | |
| Construction 1,297 1,290 1,211 86 7.1% 1,292 1,171 121 10.3% Commercial 2,591 2,567 2,335 256 11.0% 2,578 2,300 278 12.1% Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines of credit 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Residential mortgage 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.6% Other loans 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Consumer 6,644 6,039 4,920 1,724 35.0% 5 13.5% \$ 14,463 \$ 12,96 | C&I | \$ 4,287 | \$ | 4,219 | \$ | 4,590 | \$ (303) | -6.6% | \$ 4,253 | \$ | 4,631 | \$ (378) | -8.2% |
| Commercial 2,591 2,567 2,335 256 11.0% 2,578 2,300 278 12.19 Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines of credit 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Residential mortgage Other loans 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.69 Other loans 317 278 315 2 0.6% 297 320 (23) -7.29 Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.59 Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: | | 1.297 | | 1 290 | | 1 211 | 86 | 7 1% | 1.292 | | 1 171 | 121 | 10.3% |
| Consumer Auto loans - indirect 5 5 7 (2) -28.6% 5 7 (2) -28.6% Home equity loans & lines of credit 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Residential mortgage 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.6% Other loans 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: | | | | | | | | | | | | | 12.1% |
| Home equity loans & lines of credit 3,848 3,593 3,109 739 23.8% 3,720 3,053 667 21.8% Residential mortgage 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.6% Other loans 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: | Consumer | , | | | | | | | , | | | | |
| of credit 5,848 5,393 5,109 739 23.876 5,720 5,033 607 21.87 Residential mortgage 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.6% Other loans 317 278 315 2 0.6% 297 320 (23) -7.29 Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: Non-interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing deposits \$ 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 | | 5 | | 5 | | 7 | (2) | -28.6% | 5 | | 7 | (2) | -28.6% |
| of credit Residential mortgage 2,474 2,163 1,489 985 66.2% 2,318 1,480 838 56.69 Other loans 317 278 315 2 0.6% 297 320 (23) -7.29 Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.59 Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.69 Deposits: | | 3,848 | | 3,593 | | 3,109 | 739 | 23.8% | 3,720 | | 3,053 | 667 | 21.8% |
| Other loans 317 278 315 2 0.6% 297 320 (23) -7.2% Total Consumer 6,644 6,039 4,920 1,724 35.0% 6,340 4,860 1,480 30.5% Total Consumer § 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: Non-interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | | 2 474 | | 2 162 | | 1 /00 | 095 | 66 20/ | | | | 820 | 56 60/ |
| Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: Non-interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | 00 | , | | | | | | | , | | | | -7.2% |
| Total Loans \$ 14,819 \$ 14,115 \$ 13,056 \$ 1,763 13.5% \$ 14,463 \$ 12,962 \$ 1,501 11.6% Deposits: Non-interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | Total Consumer | 6.644 | _ | 6.039 | - | 4.920 | 1.724 | 35.0% | 6.340 | _ | 4.860 | 1.480 | 30.5% |
| Deposits: \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | | | ¢ | | ¢ | | | | | \$ | | | |
| Non-interest bearing deposits \$ 2,982 \$ 2,783 \$ 2,850 \$ 132 4.6% \$ 2,882 \$ 2,787 \$ 95 3.4% Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | LUAIIS | J 14,019 | ф | 14,113 | ф | 15,050 | φ 1,705 | 13.3% | J 14,403 | ¢ | 12,902 | \$ 1,501 | 11.0% |
| Interest bearing demand deposits 6,454 5,854 5,432 1,022 18.8% 6,154 5,222 932 17.8% Savings deposits 2,790 2,773 2,749 41 1.5% 2,781 2,737 44 1.6% Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | | \$ 2 082 | ¢ | 2 783 | ¢ | 2 850 | \$ 132 | 4 6% | \$ 7.887 | ¢ | 2 787 | \$ 05 | 3 /10/ |
| Savings deposits2,7902,7732,749411.5%2,7812,737441.6%Domestic time deposits3,6893,7284,184(495)-11.8%3,7094,249(540)-12.7% | 0 1 | | Φ | | Ф | | | | | Φ | | | |
| Domestic time deposits 3,689 3,728 4,184 (495) -11.8% 3,709 4,249 (540) -12.7% | | | | | | | , | | , | | | | 1.6% |
| Foreign time deposits 420 423 325 95 29.2% 421 314 107 34.1% | Domestic time deposits | 3,689 | | 3,728 | | 4,184 | (495) | -11.8% | 3,709 | | 4,249 | (540) | -12.7% |
| | Foreign time deposits | 420 | | 423 | | 325 | 95 | 29.2% | 421 | | 314 | 107 | 34.1% |

| Total Deposits | \$ 16,335 | \$ 15,561 | \$ 15,540 | \$ 795 | 5.1% | \$ 15,947 | \$ 15,309 | \$ 638 | 4.2% |
|--|-----------|--------------|--------------|-----------|------|-----------|--------------|-----------|------|
| | | | | | — | | | | — |
| Operating basis, see page 50 for Calculated assuming a 35% tax NM, not a meaningful value. | | | | | | | | | |

Table 18 - Regional Banking⁽¹⁾

| Table 18 - Regional Banking ⁽¹⁾ | | | | | | | | | |
|--|------------|----------|-----------|-------------|--------|--------------|-----------|-------------|--------|
| | 2004 | 2004 | 2003 | 2Q04 vs. 2 | Q03 | 2004 | 2003 | 1H04 vs. 1 | H03 |
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| PERFORMANCE METRICS | | | | | | | | | |
| Return on average assets | 1.52% | 1.27% | 0.69% | 0.83% | | 1.39% | 0.85% | 0.55% | |
| Return on average equity | 23.9% | 19.2% | 9.8% | 14.1% | | 21.5% | 12.1% | 9.5% | |
| Net interest margin | 4.15% | 4.25% | 4.43% | -0.28% | | 4.19% | 4.45% | -0.27% | |
| Efficiency ratio | 62.2% | 65.9% | 64.4% | -2.2% | | 64.0% | 64.3% | -0.4% | |
| CREDIT QUALITY | | | | , | | | | | |
| Net Charge-offs by Loan Type(in | | | | | | | | | |
| thousands) | | | | | | | | | |
| C&I | \$ (3,656) | \$ 5,939 | \$26,213 | \$ (29,869) | NM | \$ 2,283 | \$ 40,853 | \$ (38,570) | -94.4% |
| CRE | 941 | 1,636 | 605 | 336 | 55.5% | 2,577 | 1,151 | 1,426 | NM |
| | | | | | | | | | |
| Total commercial | (2,715) | 7,575 | 26,818 | (29,533) | NM | 4,860 | 42,004 | (37,144) | -88.4% |
| Consumer | | | | | | | | | |
| Auto loans | 40 | (21) | 34 | 6 | 17.6% | 19 | 23 | (4) | -17.4% |
| Home equity loans & lines of credit | 3,019 | 2,956 | 3,609 | (590) | -16.3% | 5,975 | 7,463 | (1,488) | -19.9% |
| Residential mortgage | 302 | 316 | 267 | 35 | 13.1% | 618 | 391 | 227 | 58.1% |
| Other loans | 1,196 | 778 | 819 | 377 | 46.0% | 1,974 | 2,055 | (81) | -3.9% |
| | <u> </u> | | | | | | | | |
| Total consumer | 4,557 | 4,029 | 4,729 | (172) | -3.6% | 8,586 | 9,932 | (1,346) | -13.6% |
| Total Net Charge-offs | \$ 1,842 | \$11,604 | \$ 31,547 | \$ (29,705) | -94.2% | \$13,446 | \$ 51,936 | \$ (38,490) | -74.1% |
| | | | | | | | | | |
| <u>Net Charge-offs - annualized</u> | | | | | | | | | |
| percentages | | | | | | | | | |
| C&I | -0.34% | 0.57% | 2.29% | -2.63% | | 0.11% | 1.78% | -1.67% | |
| CRE | 0.10% | 0.17% | 0.07% | 0.03% | | 0.13% | 0.07% | 0.07% | |
| | | | | | | | | | |
| Total commercial | -0.13% | 0.38% | 1.32% | -1.45% | | 0.12% | 1.05% | -0.93% | |
| Consumer | | | | | | | | | |
| Auto loans | 3.22% | -1.69% | 1.95% | 1.27% | | 0.76% | 0.66% | 0.10% | |
| Home equity loans & lines of credit | 0.32% | 0.33% | 0.47% | -0.15% | | 0.32% | 0.49% | -0.17% | |
| Residential mortgage | 0.05% | 0.06% | 0.07% | -0.02% | | 0.05% | 0.05% | 0.00% | |
| Other loans | 1.52% | 1.13% | 1.04% | 0.48% | | 1.33% | 1.30% | 0.04% | |
| T . 1 | | | | 0.110/ | | | 0.410/ | 0.140/ | |
| Total consumer | 0.28% | 0.27% | 0.39% | -0.11% | | 0.27% | 0.41% | -0.14% | |
| Total Net Charge-offs | 0.05% | 0.33% | 0.97% | -0.92% | | 0.19% | 0.81% | -0.62% | |
| Ū | | | | | | | | | — |
| Non-Performing Assets (NPA) (in millions) | | | | | | | | | |
| C&I | \$ 30 | \$ 42 | \$ 84 | \$ (54) | -64.3% | \$ 30 | \$ 84 | \$ (54) | -64.3% |
| CRE | 7 | 9 | 22 | (15) | -68.2% | 7 | 22 | (15) | -68.2% |
| Residential mortgage | 11 | 11 | 11 | | 0.0% | 11 | 11 | | 0.0% |
| itesiaenna morgage | | | | | 0.070 | | | | |
| Total Non-accrual Loans | 48 | 62 | 117 | (69) | -59.0% | 48 | 117 | (69) | -59.0% |
| Renegotiated loans | | | | (0) | NM | | | (0) | NM |
| Reliegotiated loans | | | | | | | | | |
| Total Non-nonforming Loong (NDL) | 40 | () | 117 | (60) | -59.0% | 40 | 117 | (60) | 50.00/ |
| Total Non-performing Loans (NPL) Other real estate, net (OREO) | 48 | 62 | 117 | (69) | | 48 | 117 | (69) | -59.0% |
| Other real estate, het (OREO) | 14 | 16 | 14 | | 0.0% | 14 | 14 | | 0.0% |
| Total Non-performing Assets | \$ 62 | \$ 78 | \$ 131 | \$ (69) | -52.7% | \$ 62 | \$ 131 | \$ (69) | -52.7% |
| | | | | | | | | | |
| Accruing loans past due 90 days or | | | | | | | | | |
| more <i>(eop)</i> | \$ 41 | \$ 47 | \$ 40 | \$ 1 | 2.5% | \$ 41 | \$ 40 | \$ 1 | 2.5% |
| Allowance for Loan and Lease Losses | | | | | | | | | |
| (ALLL) (eop) | \$ 186 | \$ 158 | \$ 199 | \$ (13) | -6.5% | \$ 186 | \$ 199 | \$ (13) | -6.5% |
| ALLL as a % of total loans and leases | 1.22% | 1.10% | 1.52% | -0.30% | | 1.22% | 1.52% | -0.30% | |
| ALLL as a % of NPLs | 387.5% | 254.8% | 170.1% | 217.4% | | 387.5% | 170.1% | 217.4% | |
| ALLL + OREO as a% of NPAs | 322.6% | 223.1% | 162.6% | 160.0% | | 322.6% | 162.6% | 160.0% | |
| NPLs as a % of total loans and leases | 0.31% | 0.43% | 0.90% | -0.59% | | 0.31% | 0.90% | -0.59% | |
| NPAs as a % of total loans and leases + | | | | | | | | | |
| OREO | 0.40% | 0.54% | 1.00% | -0.60% | | 0.40% | 1.00% | -0.60% | |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 NM, not a meaningful value.
 eop, end of period.

Table 18 - Regional Banking⁽¹⁾

| Table 16 - Regional Danking | | 2004 | | 2004 | | 2003 | | 2Q04 vs. 2 | Q03 | | 2004 | | 2003 | | 1H04 vs. 1 | H03 |
|---|----------|---------|----|---------|----|---------|----|------------|--------|----------|----------|----------|---------|----------|------------|--------|
| | | Second | | First | | Second | А | mount | % | 6 | Months | 6 | Months | A | mount | % |
| SUPPLEMENTAL DATA | | | | | | | | | | | | | | _ | | |
| # employees - full-time equivalent (eop) | | 4,953 | | 4,823 | | 4,834 | | 119 | 2.5% | | 4,953 | | 4,834 | | 119 | 2.5% |
| Retail Banking | |) | | , | | , | | | | | <i>y</i> | | , | | | |
| Average loans (in millions) | \$ | 4,545 | \$ | 4,236 | \$ | 3,694 | \$ | 851 | 23.0% | \$ | 4,385 | \$ | 3,645 | \$ | 740 | 20.3% |
| Average deposits (in millions) | \$ | 11,025 | \$ | 10,680 | \$ | 11,174 | \$ | (149) | -1.3% | \$ | 10,851 | \$ | 11,116 | \$ | (265) | -2.4% |
| # employees - full-time equivalent (eop) | | 3,510 | | 3,392 | | 3,378 | | 132 | 3.9% | | 3,510 | | 3,378 | | 132 | 3.9% |
| # banking offices (eop) | | 335 | | 332 | | 336 | | (1) | -0.3% | | 335 | | 336 | | (1) | -0.3% |
| # ATMs (eop) | | 700 | | 684 | | 855 | | (155) | -18.1% | | 700 | | 855 | | (155) | -18.1% |
| # DDA households (eop) | | 494,960 | | 491,949 | | 493,219 | | 1,741 | 0.4% | 4 | 494,960 | | 493,219 | | 1,741 | 0.4% |
| # New 90-day cross sell (average) ⁽³⁾ | | 2.2 | | 2.2 | | 1.9 | | 0.4 | 18.9% | | 2.2 | | 1.8 | | 0.4 | 22.2% |
| # on-line customers (eop) | | 185,454 | | 179,681 | | 136,700 | 4 | 48,754 | 35.7% | 1 | 185,454 | | 136,700 | 4 | 48,754 | 35.7% |
| % on-line retail household penetration | | | | | | | | | | | | | | | | |
| (eop) | | 35% | | 34% | | 26% | | 9% | | | 35% | | 26% | | 9% | |
| Small Business | | | | | | | | | | | | | | | | |
| Average loans (in millions) | \$ | 1,865 | \$ | 1,824 | \$ | 1,684 | \$ | 181 | 10.7% | \$ | 1,845 | \$ | 1,660 | \$ | 185 | 11.1% |
| Average deposits (in millions) | \$ | 1,980 | \$ | 1,844 | \$ | 1,620 | \$ | 360 | 22.2% | \$ | 1,912 | \$ | 1,585 | \$ | 327 | 20.6% |
| # employees - full-time equivalent (eop) | | 260 | | 260 | | 252 | | 8 | 3.2% | | 260 | | 252 | | 8 | 3.2% |
| # customers (eop) | | 64,558 | | 63,440 | | 68,756 | | (4,198) | -6.1% | | 64,558 | | 68,756 | | (4,198) | -6.1% |
| # New 90-day cross sell (average) (4) | | 2.2 | | 2.1 | | 1.8 | | 0.4 | 20.2% | | 2.1 | | 1.8 | | 0.3 | 16.7% |
| Corporate Banking | | | | | | | | | | | | | | | | |
| Average loans (in millions) | \$ | 6,377 | \$ | 6,316 | \$ | 6,505 | \$ | (128) | -2.0% | \$ | 6,347 | \$ | 6,495 | \$ | (148) | -2.3% |
| Average deposits (in millions) | \$ | 3,110 | \$ | 2,888 | \$ | 2,518 | \$ | 592 | 23.5% | \$ | 2,999 | \$ | 2,411 | \$ | 588 | 24.4% |
| # employees - full-time equivalent (eop) | | 574 | | 567 | | 590 | | (16) | -2.7% | | 574 | | 590 | | (16) | -2.7% |
| # customers (eop) | | 5,684 | | 5,527 | | 7,278 | | (1,594) | -21.9% | | 5,684 | | 7,278 | | (1,594) | -21.9% |
| Mortgage Banking | | | | | | | | | | | | | | | | |
| Average loans (in millions) | \$ | 2,032 | \$ | 1,739 | \$ | 1,173 | \$ | 859 | 73.2% | \$ | 1,886 | \$ | 1,162 | \$ | 724 | 62.3% |
| Average deposits (in millions) | \$ | 220 | \$ | 149 | \$ | 228 | \$ | (8) | -3.5% | \$ | 185 | \$ | 197 | \$ | (12) | -6.1% |
| # employees - full-time equivalent (eop) | | 609 | | 604 | | 614 | | (5) | -0.8% | | 609 | | 614 | | (5) | -0.8% |
| Closed loan volume (in millions) | \$ | 1,330 | \$ | 860 | \$ | 1,702 | \$ | (372) | -21.9% | \$ | 2,190 | \$ | 2,969 | \$ | (779) | -26.2% |
| Portfolio closed loan volume (in millions) | \$ | 863 | \$ | 533 | \$ | 362 | \$ | 501 | NM | \$ | 1,396 | \$ | 731 | \$ | 665 | 91.0% |
| Agency delivery volume (in millions) | \$ | 502 | \$ | 342 | \$ | 1,128 | \$ | (626) | -55.5% | \$ | 844 | \$ | 2,038 | \$ | (1,194) | -58.6% |
| Total servicing portfolio (eop and in | <i>•</i> | 0.507 | ¢ | 0.442 | ¢ | 7.462 | ¢ | | 21.10/ | <i>•</i> | 0.000 | • | 5.460 | • | | 21.10/ |
| millions) | \$ | 9,786 | \$ | 9,442 | \$ | 7,463 | \$ | 2,323 | 31.1% | \$ | 9,786 | \$ | 7,463 | \$ | 2,323 | 31.1% |
| Portfolio serviced for others (eop and in millions) | \$ | 6,537 | \$ | 6,523 | \$ | 5,069 | \$ | 1,468 | 29.0% | \$ | 6,537 | \$ | 5,069 | \$ | 1,468 | 29.0% |
| Mortage servicing rights (eop and in | P | 0,007 | φ | 0,525 | φ | 5,007 | ψ | 1,700 | 27.070 | φ | 0,007 | φ | 5,007 | φ | 1,700 | 29.070 |
| millions) | \$ | 79.2 | \$ | 60.4 | \$ | 36.7 | \$ | 42.5 | NM | \$ | 79.2 | \$ | 36.7 | \$ | 42.5 | NM |
| | | | | | | | | | | | | | | | | |

Operating basis, see page 50 for definition.
 Total cross-sell on new relationships at 90 days (out of 16 core products).
 Total cross-sell on new relationships at 90 days (out of 18 products).
 NM, not a meaningful value.

N/A, not available.

eop, end of period.

Dealer Sales

Dealer Sales serves over 3,500 automotive dealerships within Huntington's primary banking markets, as well as in Arizona, Florida, Georgia, Pennsylvania, and Tennessee. The segment finances the purchase of automobiles by customers of the automotive dealerships, purchases automobiles from dealers and simultaneously leases the automobiles under long-term direct financing leases, finances dealership floor plan inventories, real estate, or working capital needs, and provides other banking services to the automotive dealerships and their owners.

The accounting for automobile leases significantly impacts the presentation of Dealer Sales' financial results. All automobile leases originated prior to May 2002 are accounted for as operating leases, with leases originated since April 2002 accounted for as direct financing leases. For automobile leases originated prior to May 2002, the related financial results are reported as operating lease income and operating lease expense, components of non-interest income and non-interest expense, respectively, whereas the cost of funding these leases is included in interest expense. Credit losses associated with these leases are also reflected in operating lease expense. With no new operating lease being originated, this portfolio, and related operating leases, where the income and operating lease expense, will decrease over time and eventually become immaterial. In contrast, all new leases since April 2002 are originated as direct financing leases, where the income and funding are included in net interest income. The net interest margin increased during the three and six-month periods ended June 30, 2003, as compared with the same periods in 2004 as the declining operating lease portfolio resulted in less assessed interest expense. Direct financing lease credit losses are charged against an allowance for credit losses with provision for credit losses recorded to maintain an appropriate allowance level.

Residual values on leased automobiles are evaluated periodically for impairment. Residual value losses arise if the market value at the end of the lease term is less than the residual value embedded in the original lease contract. Residual value insurance covers the difference between the recorded residual value and the fair value of the automobile at the end of the lease term. Impairment of the residual values of direct financing leases is recognized by writing the leases down to fair value with a charge to non-interest income. Impairment of residual values of operating leases is evaluated under Statement No. 144. Under that Statement, when the future cash flows from the operating lease, including the expected realizable fair value of the automobile or equipment at the end of the lease, is less than the book value of the lease, an immediate impairment writedown is recognized. Otherwise, reductions in the expected residual value result in additional depreciation of the leased asset over the remaining term of the lease. Upon disposition, a gain or loss is recorded for any difference between the net book value of the lease and the proceeds.

Second Quarter 2004 versus Second Quarter 2003 Performance

Dealer Sales contributed \$21.0 million of the company's net operating earnings in the second quarter of 2004, up \$3.6 million, or 20%, from the 2003 second quarter. This increase was primarily due to higher net income from loan and lease assets (net interest income plus operating lease income less operating lease expense) and a lower provision for credit losses resulting from a decrease in charge-offs. The ROA and ROE for the second quarter of 2004 were 1.27% and 20.5%, respectively, up from 0.99% and 16.1%, respectively, in the 2003 second quarter.

Net interest income was \$37.8 million, up \$16.0 million, or 74%, from \$21.7 million in the year-ago quarter. This significant increase reflected a \$0.5 billion, or 10%, increase in average total loans and leases, as well as a 104 basis point increase in the net interest margin to 2.77% from 1.73% a year ago. The increase in average total loans and leases, and the net interest margin was driven by rapid growth in direct financing leases as average automobile loans declined due to loan sales.

Average automobile direct financing leases increased \$833 million, reflecting the fact that this is still a relatively young portfolio with relatively fewer maturities and pay-offs than a more mature portfolio. Direct financing lease originations totaled \$246 million in the second quarter of 2004, down 37% from \$389 million in the 2003 second quarter. The growth in average direct financing lease balances contrasts with the \$825 million decline in average operating lease assets, which consists of all leases originated prior to May 2002 with balances running off through maturities and pay-offs.

Average automobile loans declined \$491 million compared with the same period a year ago, reflecting sales of automobile loans. Automobile loan originations declined \$213 million, or 33%, to \$431 million in the 2004 second quarter compared with \$644 million in the 2003 second quarter, reflecting a softer market for new and used vehicle sales, as well as a decision to maintain high quality originations. The impact of the loan sales and lower production levels was offset in part by the consolidation in July 2003 of \$1.0 billion of previously securitized automobile loans related to the adoption of FIN 46.

Also contributing to the growth in average total loans and leases was a 20% increase in C&I loans, including dealer floor plan loans.

The net interest margin continued to be favorably impacted by the run-off of operating lease assets and the fact that all of the funding cost associated with these assets is reflected in interest expense, whereas the income is reflected in non-interest income. In contrast, the net interest margin was negatively impacted by growth in lower yielding direct financing lease balances, loan sales, and a lower net interest margin associated with securitized loans that are now recorded on balance sheet as a result of the adoption of FIN 46.

The provision for credit losses decreased to \$8.3 million from \$9.2 million for the year-ago quarter, primarily reflecting a \$1.3 million, or 14%, decline in charge-offs. The annualized net charge-off ratio for automobile loans was 0.96% in the second quarter of 2004, down from 1.06% in the 2003 second quarter, while the charge-off ratio for direct financing leases was 0.41%, down from 0.44% in the year-ago quarter.

Non-interest income decreased \$51.2 million, or 37%, driven by a \$50.2 million, or 39%, decline in operating lease income as that portfolio continued to run-off. Other non-interest income decreased \$0.9 million, or 9%, primarily due to declines in securitization income that resulted from the adoption of FIN 46 in the third quarter of 2003. Non-interest expense declined \$39.7 million, or 32%, primarily reflecting a \$40.7 million, or 39%, decline in operating lease expense. Other non-interest expense increased slightly as lower residual value insurance costs were offset by a loss accrual for expenses associated with pending litigation. In contrast, personnel costs increased \$0.8 million, or 16%, primarily due to higher benefit costs and less benefit from deferring loan origination costs, reflecting the decline in loan and lease production.

Second Quarter 2004 versus First Quarter 2004 Performance

Dealer Sales' net operating earnings of \$21.0 million in the second quarter of 2004 were up \$9.1 million, or 7%, from the first quarter of 2004. The primary contributor to this increase was a lower provision for credit losses. The ROA and ROE in the 2004 second quarter were 1.27% and 20.5%, respectively, up from 0.65% and 10.7%, respectively, in the 2004 first quarter.

Net interest income was \$37.8 million for second quarter of 2004, up \$2.8 million, or 8%, from the previous quarter. This increase reflected a 41 basis point increase in the net interest margin to 2.77% from 2.36%, partially offset by an 8% decline in average total loans and leases.

Average automobile loans decreased \$704 million, or 23%, primarily as a result of the sale loans. Also contributing to the decline from the 2004 first quarter was a 12% decrease in automobile loan originations in the second quarter of 2004 compared with the first quarter.

Average automobile direct financing leases increased \$151 million, or 8%, reflecting the fact that this continues to be a relatively young portfolio. However, direct financing lease originations declined 11% compared with the first quarter of 2004. The \$151 million, or 8%, growth in average direct financing lease balances contrasts with the \$189 million, or 16%, decline in average operating lease assets.

During the second quarter of 2004, C&I loans, including dealer floor plan loans, increased \$52 million, or 7%, from the prior quarter, consistent with seasonal patterns for usage of available credit lines.

The provision for credit losses decreased 62%, reflecting both lower charge-offs and lower provision expense for growth in loans and direct financing leases. Charge-offs in the 2004 second quarter represented an annualized 0.58%, down from 1.14% in the first quarter. The first quarter of 2004 included a \$4.7 million one-time charge to correct for the classification of claims received under policies purchased to protect the company from loss in the event repossessed vehicles had physical damage.

Non-interest income decreased \$8.3 million, or 9%, primarily due to a \$10.4 million, or 12%, decline in operating lease income as that portfolio continues to run-off. This decrease was partially offset by increases in servicing fee income, securitization income, and fee income from lease terminations. Non-interest expense declined \$5.6 million, or 6%, primarily reflecting an \$8.4 million, or 12%, decline in operating lease expense, offset in part by a loss accrual in other expense associated with pending litigation.

2004 First Six Months versus 2003 First Six Months

Dealer Sales contributed \$33.0 million of the company's net operating earnings for the first six months of 2004, down slightly from \$33.5 million for the same period a year ago. Earnings in 2004 compared with 2003 were primarily impacted by a higher provision for credit losses, which was driven by increased charge-offs and the provision for credit losses associated with loan and lease growth, offset by higher net income from loan and lease assets. The ROA and ROE for the first six months of 2004 were 0.95% and 15.3%, respectively, compared with 0.94% and 15.3%, respectively, for the 2003 six-month period.



Net interest income was \$72.7 million, up \$29.1 million, or 67%, from \$43.6 million in the year-ago period. This significant increase reflected an \$806 million, or 17%, increase in average total loans and leases, as well as a 79 basis point increase in the net interest margin to 2.55% from 1.76% a year ago. The significant increase in average total loans and leases, as well as the net interest margin, was driven by rapid growth in direct financing leases as average automobile loans declined.

Average automobile direct financing leases increased \$911 million, or 79%, reflecting the fact that this is still a relatively young portfolio with fewer maturities and payoffs than a more mature portfolio. Direct financing lease originations totaled \$522 million in the first six months of 2004, down 25% from \$699 million in the first six months of 2003. The growth in average direct financing lease balances contrasts with the \$867 million, or 45%, decline in average operating lease assets, which consists of all leases originated prior to May 2002 with balances running off through maturities and pay-offs.

Average automobile loans declined \$263 million compared with the same period a year ago, reflecting the impact of loan sales, as well as a \$436 million, or 32%, decline in originations, reflecting a softer market for new and used vehicle sales, as well as a decision to maintain high quality loan originations. The impact of the loan sales and lower production levels was offset in part by the consolidation in July 2003 of \$1.0 billion of previously securitized loans related to the adoption of FIN 46.

Also contributing to the growth in average total loans and leases was a \$121 million, or 18%, increase in C&I loans, including dealer floor plan loans.

The net interest margin continued to be favorably impacted by the run-off of the operating lease assets and the fact that all of the funding cost associated with these assets is reflected in interest expense, whereas the income is reflected in non-interest income. In contrast, the net interest margin was negatively impacted by growth in lower yielding direct financing lease balances, loan sales, and a lower net interest margin associated with securitized loans that are now recorded on balance sheet as a result of the adoption of FIN 46.

The provision for credit losses increased from \$20.6 million from the year-ago period to \$29.9 million, reflecting growth in direct financing leases, an increase in provision allocation factors and a \$3.6 million increase in charge-offs. Though net charge-offs increased \$3.6 million from the year-ago six-month period, as an annualized percent of average total loans and leases, total net charge-offs were unchanged at 0.87%.

Non-interest income decreased \$103.0 million, or 36%, driven by a \$99.6 million, or 37%, decline in operating lease income as that portfolio continued to run-off. Other non-interest income decreased \$2.5 million, or 14%, primarily due to declines in securitization income that resulted from the adoption of FIN 46 in the third quarter of 2003. Non-interest expense declined \$82.5 million, or 32%, primarily reflecting an \$81.6 million, or 38%, decline in operating lease expense. Other non-interest expense declined \$2.4 million, or 7%, primarily due to lower residual value insurance costs. In contrast, personnel costs increased \$1.4 million, or 14%, primarily due to higher benefit costs and less benefit from deferring loan origination costs, reflecting the decline in loan and lease production.

| | 2004 | 2004 | 2003 | 2Q04 vs. | 2Q03 | 2004 | 2003 | 1H04 vs. 1 | 1H03 |
|--|---------------|------------|------------|-----------------|---------------|----------------|------------|------------------|---------------|
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| INCOME STATEMENT (in thousands) | | | | | _ | | | | |
| Net Interest Income | \$ 37,757 | \$ 34,951 | \$ 21,727 | \$ 16,030 | 73.8% | \$ 72,708 | \$ 43,642 | \$ 29,066 | 66.6% |
| Provision for credit losses | 8,261 | 21,655 | 9,191 | (930) | -10.1% | 29,916 | 20,576 | 9,340 | 45.4% |
| Net Interest Income After Provision for Credit Losses | 29,496 | 13,296 | 12,536 | 16,960 | NM | 42,792 | 23,066 | 19,726 | 85.5% |
| Operating lease income | 78,379 | 88,818 | 128,574 | (50,195) | -39.0% | 167,197 | 266,767 | (99,570) | -37.3% |
| Service charges on deposit accounts | 230 | 199 | 231 | (1) | -0.4% | 429 | 438 | (9) | -2.1% |
| Brokerage and insurance income | 335 | 510 | 851 | (516) | -60.6% | 845 | 2,193 | (1,348) | -61.5% |
| Trust services Mortgage banking | | _ | 6 | (6) | -100.0% NM | _ | 6 2 | (6) | -100.0% |
| Other service charges and fees | | | | | NM | _ | 2 | (2) | -100.0% NM |
| Other | 9,218 | 6,926 | 10,145 | (927) | -9.1% | 16,144 | 18,662 | (2,518) | -13.5% |
| | | | | | | | | | |
| Total Non-Interest Income Before Securities Gains Securities gains | 88,162 469 | 96,453 | 139,807 | (51,645) 469 | -36.9% NM | 184,615 469 | 288,068 | (103,453) 469 | -35.9% NM |
| Securites gains | 407 | | | 407 | | 407 | | | |
| Total Non-Interest Income | 88,631 | 96,453 | 139,807 | (51,176) | -36.6% | 185,084 | 288,068 | (102,984) | -35.7% |
| Operating lease expense | 62,288 | 70,666 | 102,939 | (40,651) | -39.5% | 132,954 | 214,527 | (81,573) | -38.0% |
| Personnel costs | 5,666 | 5,901 | 4,882 | 784 | 16.1% | 11,567 | 10,122 | 1,445 | 14.3% |
| Other | 17,814 | 14,802 | 17,643 | 171 | 1.0% | 32,616 | 34,984 | (2,368) | -6.8% |
| Total Non-Interest Expense | 85,768 | 91,369 | 125,464 | (39,696) | -31.6% | 177,137 | 259,633 | (82,496) | -31.8% |
| Income Before Provision for Income Taxes | 32,359 | 18,380 | 26,879 | 5,480 | 20.4% | 50,739 | 51,501 | (762) | -1.5% |
| Provision for income taxes ⁽²⁾ | 11,326 | 6,433 | 9,408 | 1,918 | 20.4% | 17,759 | 18,026 | (267) | -1.5% |
| | | | | | | | | | |
| Net Income - Operating ⁽¹⁾ | \$ 21,033 | \$ 11,947 | \$ 17,471 | \$ 3,562 | 20.4% | \$ 32,980 | \$ 33,475 | \$ (495) | -1.5% |
| Revenue - Fully Taxable Equivalent (FTE) | | | | | | | | | |
| Net Interest Income | \$ 37,757 | \$ 34,951 | \$ 21,727 | \$ 16,030 | 73.8% | \$ 72,708 | \$ 43,642 | \$ 29,066 | 66.6% |
| Tax Equivalent Adjustment ⁽²⁾ | — | _ | _ | _ | NM | — | _ | — | NM |
| | | | | | — | | | | — |
| Net Interest Income (FTE) | 37,757 | 34,951 | 21,727 | 16,030 | 73.8% | 72,708 | 43,642 | 29,066 | 66.6% |
| Non-Interest Income | 88,631 | 96,453 | 139,807 | (51,176) | -36.6% | 185,084 | 288,068 | (102,984) | -35.7% |
| Total Revenue (FTE) | \$ 126,388 | \$ 131,404 | \$ 161,534 | \$ (35,146) | -21.8% | \$ 257,792 | \$ 331,710 | \$ (73,918) | -22.3% |
| Total Revenue Excluding Securities Gains (FTE) | \$ 125,919 | \$ 131,404 | \$ 161,534 | \$ (35,615) | -22.0% | \$ 257,323 | \$ 331,710 | \$ (74,387) | -22.4% |
| | | | | | | | | | |
| SELECTED AVERAGE BALANCES (in millions) Loans: | | | | | | | | | |
| C&I | \$ 808 | \$ 756 | \$ 673 | \$ 135 | 20.1% | \$ 782 | \$ 661 | \$ 121 | 18.3% |
| CRE Construction | 3 | 8 | 6 | (3) | -50.0% | 6 | 4 | 2 | 50.0% |
| Commercial | 80 | 81 | 64 | 16 | 25.0% | 81 | 62 | 19 | 30.6% |
| Consumer | | | | | | | | | |
| Auto leases – indirect | 2,139 | 1,988 | 1,306 | 833 | 63.8% | 2,064 | 1,153 | 911 | 79.0% |
| Auto loans – indirect | 2,332 | 3,036 | 2,823 | (491) | -17.4% | 2,684 | 2,947 | (263) | -8.9% |
| Home equity loans & lines of credit Other loans | 0 74 | 0 70 | 0 57 | 0 17 | NM 29.8% | 0 72 | 0 56 | 0 16 | NM 28.6% |
| Otter toalis | /4 | /0 | | 17 | 27.8 /0 | 12 | | 10 | 28.070 |
| Total Consumer | 4,545 | 5,094 | 4,186 | 359 | 8.6% | 4,820 | 4,156 | 664 | 16.0% |
| Total Loans | \$ 5,436 | \$ 5,939 | \$ 4,929 | \$ 507 | 10.3% | \$ 5,689 | \$ 4,883 | \$ 806 | 16.5% |
| | e 077 | § 1100 | ¢ 1.000 | ¢ (835) | 45 00/ | 6 1.070 | ¢ 1.027 | \$ (9(7) | 44.007 |
| Operating lease assets | \$ 977 | \$ 1,166 | \$ 1,802 | \$ (825) | -45.8% | \$ 1,070 | \$ 1,937 | \$ (867) | -44.8% |
| Deposits: Non-interest bearing deposits | \$ 67 | \$ 65 | \$ 55 | \$ 12 | 21.8% | \$ 66 | \$ 55 | \$ 11 | 20.0% |
| Interest bearing demand deposits | 2 | 2 | ¢ 55 | 1 | 100.0% | 2 | ¢ 55 | 1 | 100.0% |
| Foreign time deposits | 4 | 4 | 4 | 0 | 0.0% | 4 | 4 | 0 | 0.0% |
| | | | | | | | | | |
| Total Deposits | \$ 73 | \$ 71 | \$ 60 | \$ 13 | 21.7% | \$ 72 | \$ 60 | \$ 12 | 20.0% |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 Calculated assuming a 35% tax rate.
 NM, not a meaningful value.

| Table | 19 - | Dealer | Sales(1) |
|-------|------|--------|----------|
| | | | |

| Table 19 - Dealer Sales ⁽¹⁾ | 2004 | 2004 | 2003 | 2Q04 vs. 2 | 2Q03 | 2004 | 2003 | 1H04 vs. 1H03 | |
|---|-------------|-----------|----------|------------|---------|-------------|-----------|---------------|---------|
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| PERFORMANCE METRICS | | | | | | | | | |
| Return on average assets | 1.27% | 0.65% | 0.99% | 0.28% | | 0.95% | 0.94% | 0.00% | |
| Return on average equity | 20.5% | 10.7% | 16.1% | 4.4% | | 15.3% | 15.3% | 0.1% | |
| Net interest margin | 2.77% | 2.36% | 1.73% | 1.04% | | 2.55% | 1.76% | 0.79% | |
| Efficiency ratio | 68.1% | 69.5% | 77.7% | -9.6% | | 68.8% | 78.3% | -9.4% | |
| | 0011 /0 | 0,10,70 | ,,,,,, | 21070 | | 001070 | 101070 | 2.1.70 | |
| CREDIT QUALITY Net Charge-offs by Loan Type (in thousands) | | | | | | | | | |
| C&I | \$ 36 | \$ 1 | \$ — | \$ 36 | NM | \$ 37 | \$ (38) | \$ 75 | NM |
| CRE | — | | _ | | NM | | | | NM |
| | | | | | | | | | |
| Total commercial | 36 | 1 | | 36 | NM | 37 | (38) | 75 | NM |
| Consumer | | | | | | | () | | |
| Auto leases | 2,159 | 3,159 | 1,422 | 737 | 51.8% | 5,318 | 2,342 | 2,976 | NM |
| Auto loans | 5,564 | 13,443 | 7,490 | (1,926) | -25.7% | 19,007 | 18,124 | 883 | 4.9% |
| Home equity loans & lines of credit | | | | (1,520) | NM | | 10,121 | | NM |
| Other loans | 38 | 211 | 177 | (139) | -78.5% | 249 | 575 | (326) | -56.7% |
| | | | 177 | (157) | -78.570 | | | (320) | -50.770 |
| Total consumer | 7,761 | 16,813 | 9,089 | (1,328) | -14.6% | 24,574 | 21,041 | 3,533 | 16.8% |
| Total Net Charge-offs | \$ 7,797 | \$ 16,814 | \$ 9,089 | \$(1,292) | -14.2% | \$ 24,611 | \$ 21,003 | \$ 3,608 | 17.2% |
| Not Charge offer annualized noncentages | | | | | | | | | |
| <u>Net Charge-offs—annualized percentages</u> | 0.030/ | 0.000/ | 0.000/ | 0.020/ | | 0.010/ | 0.010/ | 0.020/ | |
| C&I | 0.02% | 0.00% | 0.00% | 0.02% | | 0.01% | -0.01% | 0.02% | |
| CRE | 0.00% | 0.00% | 0.00% | 0.00% | | 0.00% | 0.00% | 0.00% | |
| Total commercial | 0.02% | 0.00% | 0.00% | 0.02% | | 0.01% | -0.01% | 0.02% | |
| Consumer | | | | | | | | | |
| Auto leases | 0.41% | 0.64% | 0.44% | -0.03% | | 0.52% | 0.41% | 0.11% | |
| Auto loans | 0.96% | 1.78% | 1.06% | -0.10% | | 1.42% | 1.24% | 0.18% | |
| Home equity loans & lines of credit | NM | NM | NM | NM | | NM | NM | NM | |
| Other loans | 0.21% | 1.21% | 1.25% | -1.04% | | 0.69% | 2.07% | -1.38% | |
| Total consumer | 0.69% | 1.33% | 0.87% | -0.18% | | 1.02% | 1.02% | 0.00% | |
| | 0.500/ | 1.1.40/ | 0.740/ | 0.1.01 | | 0.050/ | 0.050/ | | |
| Total Net Charge-offs | 0.58% | 1.14% | 0.74% | -0.16% | | 0.87% | 0.87% | 0.00% | |
| Non-performing Assets (NPA) (in millions) | | | | | | | | | |
| C&I | \$ — | \$ — | \$ — | \$ — | NM | \$ — | \$ — | \$ — | NM |
| CRE | | _ | | | NM | | | | NM |
| | | | | | | | | | |
| Total Non-accrual Loans | 0 | 0 | 0 | 0 | NM | 0 | 0 | 0 | NM |
| Renegotiated loans | _ | _ | _ | _ | NM | _ | _ | _ | NM |
| | | | | | | | | | |
| Total Non-performing Loans (NPL) | 0 | 0 | 0 | | NM | 0 | 0 | _ | NM |
| Other real estate, net (OREO) | | _ | | _ | NM | | | _ | NM |
| Total Non-performing Assets | \$ — | \$ — | \$ — | \$ — | NM | \$ — | \$ — | \$ — | NM |
| | | | | | | | | | |
| Accruing loans past due 90 days or more | \$8 | \$9 | \$ 11 | \$ (3) | -27.3% | \$ 8 | \$ 11 | \$ (3) | -27.3% |
| Allowance for Loan and Lease Losses (ALLL) (eop) | \$ 50 | \$ 51 | \$ 63 | \$ (13) | -20.6% | \$ 50 | \$ 63 | \$ (13) | -20.6% |
| ALLL as a % of total loans and leases | 1.01% | 0.96% | 1.36% | -0.35% | -20.070 | 1.01% | 1.36% | -0.35% | -20.070 |
| | | | | | | | | | |
| ALLL as a % of NPLs | NM | NM | NM | NM | | NM | NM | NM | |
| ALLL + OREO as a % of NPAs | NM | NM | NM | NM | | NM | NM | NM | |
| NPLs as a % of total loans and leases | | 0.00% | 0.00% | 0.00% | | 0.00% | 0.00% | 0.00% | |
| NPAs as a % of total loans and leases + OREO | 0.00% | 0.00% | 0.00% | 0.00% | | 0.00% | 0.00% | 0.00% | |

Operating basis, see page 50 for definition.
 NM, not a meaningful value.
 eop, end of period.

Table 19 - Dealer Sales⁽¹⁾

| | 2004 | 2004 | 2003 | 2Q04 vs. 2 | 2Q03 | 2004 | 2003 | 1H04 vs. | 1H03 |
|--|----------|---------|----------|------------|--------|----------|------------|----------|--------|
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| SUPPLEMENTAL DATA | | | | | | | | | |
| # employees - full-time equivalent (eop) | 409 | 433 | 441 | (32) | -7.3% | 409 | 441 | (32) | -7.3% |
| Automobile loans | | | | | | | | | |
| Production (in millions) | \$ 431.2 | \$487.9 | \$ 644.2 | (213) | -33.1% | \$ 919.1 | \$ 1,355.5 | (436) | -32.2% |
| % Production new vehicles | 52.0% | 52.7% | 55.1% | -3.1% | | 52.4% | 53.6% | -1.2% | |
| Average term (in months) | 65.1 | 64.7 | 64.3 | 0.8 | | 64.9 | 64.2 | 0.7 | |
| Automobile leases | | | | | | | | | |
| Production (in millions) | \$ 246.4 | \$275.4 | \$388.7 | (142) | -36.6% | \$ 521.8 | \$ 698.9 | (177) | -25.3% |
| % Production new vehicles | 99.1% | 98.9% | 96.9% | 2.2% | | 99.0% | 95.7% | 3.3% | |
| Average term (in months) | 54.6 | 53.4 | 52.4 | 2.2 | | 54.0 | 53.0 | 0.9 | |
| Average residual % | 41.0% | 42.3% | 43.7% | -2.7% | | 41.7% | 42.9% | -1.2% | |

(1) Operating basis, see page 50 for definition. eop, end of period.

Private Financial Group

The Private Financial Group (PFG) provides products and services designed to meet the needs of the company's higher net worth customers. Revenue is derived through trust, asset management, investment advisory, brokerage, insurance, and private banking products and services. The trust division provides fiduciary services to more than 11,000 accounts with assets totaling \$39.2 billion, including \$8.7 billion managed by PFG. In addition, PFG has over \$560 million in assets managed by Haberer Registered Investment Advisor, which provides investment management services to nearly 400 customers.

PFG provides investment management and custodial services to the company's 29 proprietary mutual funds, including ten variable annuity funds, which represented nearly \$3 billion in total assets under management at June 30, 2004. The Huntington Investment Company offers brokerage and investment advisory services to both Regional Banking and PFG customers through more than 100 licensed investment sales representatives and 700 licensed personal bankers. This customer base has over \$4 billion in mutual fund and annuity assets. PFG's insurance entities provide a complete array of insurance products including individual life insurance products ranging from basic term life insurance, to estate planning, group life and health insurance, property and casualty insurance, mortgage title insurance, and reinsurance for payment protection products. Income and related expenses from the sale of brokerage and insurance products is shared with the line of business that generated the sale or provided the customer referral, most notably Regional Banking.

2004 Second Quarter versus 2003 Second Quarter

The Private Financial Group (PFG) contributed \$6.3 million of the company's net operating earnings in the second quarter of 2004, down \$1.7 million, or 21%, from the second quarter of 2003. The decline primarily reflected higher non-interest expense and provision for credit losses, partially offset by an increase in net interest income. The ROA in the 2004 second quarter was 1.67%, down from 2.45% in the year-ago quarter. The ROE was 21.3% in the current quarter, down from 30.3% in the 2003 second quarter.

Net interest income increased \$1.4 million, or 14%, from the year-ago quarter as average loan balances increased \$200 million, or 17%, and average deposit balances increased \$87 million, or 9%. Most of the loan growth occurred in residential real estate loans and home equity loans and lines, largely due to the favorable mortgage rate environment. Total C&I and CRE loans grew 8% from the prior-year quarter. Loan growth reflected initiatives to add relationship managers in several markets targeted for growth opportunities, most notably East and West Michigan and Central and Southern Ohio. Most of the deposit growth was attributable to growth in commercial non-interest bearing and interest bearing demand deposit accounts aided by favorable rate offerings. The net interest margin declined 9 basis points from the year-ago quarter to 3.14%, reflecting growth in lower-margin loans partially offset by deposit growth.

Provision for credit losses increased \$1.1 million from the year-ago quarter. The increase was mainly due to the fact that the year-ago quarter included a provision credit resulting from a large, previously non-performing loan being brought current, thus freeing-up reserves. The annualized total net charge-off ratio increased to 0.27% in the second quarter 2004 from 0.15% in the year-ago quarter, primarily reflecting timing differences as evidenced by the fact that the annualized net charge-offs ratio for the first six months of 2004 was unchanged from the comparable year-ago period.

Non-interest income, net of fees shared with other business units, was down \$0.2 million from the year-ago quarter, reflecting declines in brokerage and insurance revenue, other income, and mortgage banking income, only partially offset by growth in trust income. Brokerage and insurance revenue decreased \$0.8 million, or 9%. This reflected a \$0.7 million, or 20%, decline in insurance revenue mainly due to a decrease in title insurance revenue reflecting the slowdown in mortgage refinancing activity, as well as a \$0.2 million, or 3%, decline in annuity revenues. The annuity revenue decrease of 3% was less than the 6% decline in annuity sales, reflecting a shift from lower-margin fixed annuity sales in the year-ago quarter to higher-margin variable and indexed annuity sales in the current quarter. While mutual fund sales volume declined 24%, mutual fund revenue increased 23%, as sales in the year-ago quarter consisted of more low-margin, individual trades greater than \$1 million each. Trust income increased \$1.2 million, or 8%, mainly due to revenue growth in personal trust and investment management business. Total trust assets grew 15% to \$39.2 billion at June 30, 2003. For the same periods, managed assets grew 6% to \$9.3 billion from \$8.8 billion. Asset growth resulted from a combination of market value growth and strong new business development, which continued to reflect the success of the new business delivery model utilizing the brokerage sales force to sell asset management services.

Non-interest expense increased \$2.7 million, or 10%, from the 2003 second quarter primarily due to a 14% increase in personnel costs. This increase reflected growth in the brokerage sales force to support the new sales distribution model, an increase in private banking relationship managers to support the expansion of the private banking presence in selected markets, higher trust and investment management sales commissions from increased new business development, and higher benefit costs.

2004 Second Quarter versus 2004 First Quarter

PFG's \$6.3 million net contribution to the company's operating earnings in the second quarter of 2004 was down \$0.8 million, or 11%, from the 2004 first quarter. The decrease was primarily due to higher provision for credit losses, lower non-interest income driven by a decline in broker and insurance revenue, only partially offset by lower non-interest expense. The ROA in the second quarter was 1.67%, down slightly from 1.92% in the first quarter. The ROE was 21.3% in the current quarter, down from 24.2% in the previous quarter.

Net interest income was essentially unchanged from the previous quarter. Average loan balances increased by \$34 million, or 3%, while average deposit balances decreased by \$23 million, or 2%. Average loan growth in the current quarter was primarily in C&I and CRE loans. The decline in deposit balances partially reflected normal historical trends resulting from annual income tax payments, but also some balance run-off due to higher competitive rates, most notably in East Michigan. The decline in deposits, combined with the increase in lower-margin consumer loans, was reflected in a 10 basis point decline in the net interest margin to 3.14% from 3.24% in the first quarter.

Provision for credit losses increased \$1.2 million from the prior quarter. The increase was largely due to a provision credit in the first quarter 2004 resulting from a large, fully-reserved loan being brought current. Although the annualized total net charge-off ratio increased to 0.27% in the current quarter from 0.06% in the first quarter, this had little impact on provision expense in the current period as several of the larger second quarter charge-offs had been fully reserved.

Non-interest income, net of fees shared with other business units, decreased \$0.9 million, or 3%, from the first quarter, primarily reflecting a \$2.1 million, or 20%, decline in brokerage and insurance income. Annuity revenue decreased \$1.5 million as sales volume decreased 17%. The decline in annuity sales was attributable to several factors, including a slowdown in customer demand given future interest rate movement uncertainty, increased focus on growing deposits including increased rates offered on CDs, and normal post income-tax season sales decline. Mutual fund revenue declined 16% as sales in the current quarter consisted of more low-margin, individual trades greater than \$1 million each. Insurance income decreased \$0.2 million from the prior quarter primarily due to reduced sales in the life insurance agency business.

Trust income increased \$0.5 million, or 3%, from the previous quarter. While total trust and managed assets both grew only slightly, the increase in related income was driven by growth in total personal trust assets and managed personal trust assets, both of which grew by more than 4% during the quarter. Personal trust asset growth continued despite flat market trends, which illustrated strong new business development due in part to the success of the new business delivery model, which utilizes the brokerage sales force to sell asset management services. Other income increased \$0.4 million from the first quarter due to increased income from annual BOLI insurance premiums and securities gains.

Non-interest expense decreased \$0.9 million, or 3%, from the prior quarter, primarily reflecting lower personnel costs due to a decline in commission expense as a result of the reduction in brokerage and insurance revenue.

2004 First Six Months versus 2003 First Six Months

PFG's \$13.3 million net contribution to the company's operating earnings for the first six months of 2004 was essentially unchanged from the first six months of 2003 as the benefits of the growth in net interest income, lower provision for credit losses, as well as higher non-interest income, was offset by increased non-interest expense. The ROA for 2004 year-to-date was 1.79%, down from 2.10% for the same period in 2003. The ROE was 22.6% and 26.7% for the first six months of 2004 and 2003, respectively.

Net interest income for the first six months of 2004 increased \$3.0 million, or 16%, from the comparable 2003 period as a result of strong growth in average loans and deposits. Average loans increased \$212 million, or 19%, while average deposit balances increased \$125 million, or 13%. Much of the loan growth was in residential real estate loans and home equity loans and lines, largely due to historically low mortgage rates. Loan growth also reflected an initiative to add relationship managers in 2004 in several markets targeted for growth opportunities, most notably East and West Michigan and Central and Southern Ohio. The deposit growth occurred mainly in consumer and commercial interest bearing accounts, which increased \$88 million, or 14%, aided by promotional rate offerings resulting in both new accounts and a redirection of sweep account balances from money market mutual fund accounts. Non-interest bearing deposits also contributed to the growth in deposits, increasing \$30 million, or 21%, from the comparable year-ago period. The growth in lower-margin consumer loans contributed to the 9 basis point decline in the net interest margin to 3.18% for the first half 2004 from 3.27% in the year-ago period.

Provision for credit losses decreased \$1.3 million in the current six-month period from the prior-year period. This was due to a methodology change in 2003, which increased provision expense for unfunded commitments and unsecured credits. The annualized total net charge-off ratio for the first six months of 2004 was 0.17%, unchanged from the first half of 2003.

Non-interest income, net of fees shared with other business units, increased \$1.2 million, or 2%, from the first six months of 2003 mainly as a result of increased trust income. Trust income increased \$2.6 million, or 9%, primarily due to revenue growth in the personal trust and investment management business. Total trust assets grew 15% to \$39.2 billion at June 30, 2004, from \$34.2 billion at June 30, 2003. For the same period, managed assets grew 6% from \$8.8 billion to \$9.3 billion. Asset growth resulted from a combination of market value growth and strong new business development, which continued to reflect the success of the new business delivery model utilizing the brokerage sales force to sell asset management services.

Brokerage and insurance revenue was essentially unchanged from the prior-year six-month period as increased mutual fund revenue was offset by a decrease in insurance fees and annuity income. Mutual fund revenue increased \$1.0 million, or 52%, despite a 21% decline in mutual fund sales volumes as sales in the first half of 2003 consisted of more low-margin, individual trades greater than \$1 million each. Annuity revenue decreased \$0.3 million due to a decline in sales volume. Annuity sales were extremely strong in early 2003 as promotional rates helped capture a significant amount of maturing CD business. Insurance income decreased \$1.7 million, or 23%, primarily due to a change in accounting methodology in 2004 related to insurance coverage provided on automobile loans and leases, whereby agency commissions and excess premium refunds are no longer recognized as insurance revenue.

Mortgage banking income decreased by \$0.6 million due to increased mortgage portfolio servicing costs resulting from the growth in residential real estate mortgage loans and due to a change in fee sharing methodology that resulted in reduced income shared by Huntington Mortgage Group. Other income decreased by \$0.8 million as a result of a decline in fee sharing revenue received from Huntington Capital Corporation based on fewer interest rate swap referrals, and investment portfolio gains realized in the year-ago period.

Non-interest expense increased \$5.5 million, or 10%, in the first half 2004 from the first half 2003, primarily due to 13% increase in personnel costs. This increase reflected the same factors which impacted 2004 second quarter versus 2003 second quarter performance. In addition, other expense increased \$1.4 million, or 6%, reflecting a \$1.1 million increase in allocated expense for corporate overhead and other indirect and product-related expenses.

| Table 2 | 0 - Priv | ate Fina | ncial G | roup ⁽¹⁾ |
|---------|----------|----------|---------|---------------------|

| Table 20 - Private Financial Group ⁽¹⁾ | 2004 | 2004 | 2003 Second | 2Q04 vs. 2Q03 | | 2004 | 2003 | 1H04 vs. 1H03 | |
|---|------------|-----------|----------------|---------------|--------|-----------|-----------|------------------|--------|
| | Second | First | | Amount | % | 6 Months | 6 Months | Amount | % |
| INCOME STATEMENT (in thousands) | . <u> </u> | · | | | | | | | |
| Net Interest Income | \$11,166 | \$11,129 | \$ 9,785 | \$ 1,381 | 14.1% | \$ 22,295 | \$ 19,280 | \$ 3,015 | 15.6% |
| Provision for credit losses | 654 | (557) | (457) | 1,111 | NM | 97 | 1,443 | (1,346) | -93.3% |
| Net Interest Income After Provision for Credit | | | | | | | | | |
| Losses | 10,512 | 11,686 | 10,242 | 270 | 2.6% | 22,198 | 17,837 | 4,361 | 24.4% |
| Service charges on deposit accounts | 1,002 | 925 | 966 | 36 | 3.7% | 1,927 | 1,952 | (25) | -1.3% |
| Brokerage and insurance income | 8,488 | 10,616 | 9,287 | (799) | -8.6% | 19,104 | 19,218 | (114) | -0.6% |
| Trust services | 16,483 | 16,031 | 15,294 | 1,189 | 7.8% | 32,514 | 29,937 | 2,577 | 8.6% |
| Mortgage banking | (104) | (129) | 188 | (292) | NM | (233) | 367 | (600) | NM |
| Other service charges and fees | 116 | 100 | 113 | 3 | 2.7% | 216 | 211 | 5 | 2.4% |
| Other | 1,445 | 1,084 | 2,036 | (591) | -29.0% | 2,529 | 3,369 | (840) | -24.9% |
| Total Non-Interest Income Before Securities | | | | | | | | | |
| Gains | 27,430 | 28,627 | 27,884 | (454) | -1.6% | 56,057 | 55,054 | 1,003 | 1.8% |
| Securities gains | 250 | | (34) | 284 | NM | 250 | 9 | 241 | NM |
| Total Non-Interest Income | 27,680 | 28,627 | 27,850 | (170) | -0.6% | 56,307 | 55,063 | 1,244 | 2.3% |
| Personnel costs | 17,047 | 17,800 | 14,948 | 2,099 | 14.0% | 34,847 | 30,718 | 4,129 | 13.4% |
| Other | 11,512 | 11,661 | 10,938 | 574 | 5.2% | 23,173 | 21,803 | 1,370 | 6.3% |
| Total Non-Interest Expense | 28,559 | 29,461 | 25,886 | 2,673 | 10.3% | 58,020 | 52,521 | 5,499 | 10.5% |
| | 20,557 | 29,401 | 25,000 | 2,075 | 10.570 | 50,020 | 52,521 | 5,499 | 10.570 |
| Income Before Provision for Income Taxes | 9,633 | 10,852 | 12,206 | (2,573) | -21.1% | 20,485 | 20,379 | 106 | 0.5% |
| Provision for income taxes (2) | 3,372 | 3,798 | 4,272 | (900) | -21.1% | 7,170 | 7,133 | 37 | 0.5% |
| Net Income - Operating (1) | \$ 6,261 | \$ 7,054 | \$ 7,934 | \$(1,673) | -21.1% | \$ 13,315 | \$ 13,246 | \$ 69 | 0.5% |
| Revenue - Fully Taxable Equivalent (FTE) | | | | | | | | | |
| Net Interest Income | \$11,166 | \$11,129 | \$ 9,785 | \$ 1,381 | 14.1% | \$ 22,295 | \$ 19,280 | \$ 3,015 | 15.6% |
| Tax Equivalent Adjustment (2) | 9 | 9 | 10 | (1) | -10.0% | 18 | 25 | (7) | -28.0% |
| Net Interest Income (FTE) | 11,175 | 11,138 | 9,795 | 1,380 | 14.1% | 22,313 | 19,305 | 3,008 | 15.6% |
| Non-Interest Income | 27,680 | 28,627 | 27,850 | (170) | -0.6% | 56,307 | 55,063 | 1,244 | 2.3% |
| Total Revenue (FTE) | \$ 38,855 | \$ 39,765 | \$ 37,645 | \$ 1,210 | 3.2% | \$ 78,620 | \$ 74,368 | \$ 4,252 | 5.7% |
| Total Revenue Excluding Securities Gains (FTE) | \$ 38,605 | \$ 39,765 | \$ 37,679 | \$ 926 | 2.5% | \$ 78,370 | \$ 74,359 | \$ 4,011 | 5.4% |
| SELECTED AVERAGE BALANCES (in | | | | | | | | | _ |
| millions) | | | | | | | | | |
| Loans: C&I | \$ 339 | \$ 320 | \$ 311 | \$ 28 | 9.0% | \$ 330 | \$ 316 | \$ 14 | 4.4% |
| CRE | \$ 333 | \$ 520 | \$ 511 | \$ 20 | 9.070 | \$ 550 | \$ 510 | φ 1 4 | 4.470 |
| Construction | 21 | 24 | 23 | (2) | -8.7% | 23 | 20 | 3 | 15.0% |
| Commercial | 176 | 166 | 161 | 15 | 9.3% | 171 | 156 | 15 | 9.6% |
| Consumer | | | | | | | | | |
| Home equity loans & lines of credit | 294 | 287 | 250 | 44 | 17.6% | 291 | 245 | 46 | 18.8% |
| Residential mortgage Other loans | 512 | 511 | 398 | 114 | 28.6% | 512 | 379 | 133 | 35.1% |
| Other loans | 8 | 8 | 7 | 1 | 14.3% | 8 | 7 | 1 | 14.3% |
| Total Consumer | 814 | 806 | 655 | 159 | 24.3% | 811 | 631 | 180 | 28.5% |
| Total Loans | \$ 1,350 | \$ 1,316 | \$ 1,150 | \$ 200 | 17.4% | \$ 1,335 | \$ 1,123 | \$ 212 | 18.9% |
| Denesita | | | | | | | | | |
| Deposits: Non-interest bearing deposits | \$ 174 | \$ 169 | \$ 141 | \$ 33 | 23.4% | \$ 172 | \$ 142 | \$ 30 | 21.1% |
| Interest bearing demand deposits | 712 | 753 | 667 | 45 | 6.7% | 733 | 645 | 88 | 13.6% |
| Savings deposits | 49 | 46 | 55 | (6) | -10.9% | 48 | 51 | (3) | -5.9% |
| Domestic time deposits | 106 | 96 | 95 | 11 | 11.6% | 101 | 98 | 3 | 3.1% |
| Foreign time deposits | 21 | 21 | 17 | 4 | 23.5% | 21 | 14 | 7 | 50.0% |
| Total Deposits | \$ 1,062 | \$ 1,085 | \$ 975 | \$ 87 | 8.9% | \$ 1,075 | \$ 950 | \$ 125 | 13.2% |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 Calculated assuming a 35% tax rate.
 NM, not a meaningful value.

Table 20 - Private Financial Group⁽¹⁾

| | 2004 | 2004 | 2003 Second | 2Q04 vs. 2Q03 | | 2004 | 2003 | 1H04 vs. 1H03 | |
|--|--------|----------|----------------|---------------|---------------|----------|----------|---------------|---------------|
| | Second | First | | Amount | % | 6 Months | 6 Months | Amount | % |
| PERFORMANCE METRICS | | | | | | | | | |
| Return on average assets | 1.67% | 1.92% | 2.45% | -0.78% | | 1.79% | 2.10% | -0.32% | |
| Return on average equity | 21.3% | 24.2% | 30.3% | -9.0% | | 22.6% | 26.7% | -4.1% | |
| Net interest margin | 3.14% | 3.24% | 3.23% | -0.09% | | 3.18% | 3.27% | -0.09% | |
| Efficiency ratio | 74.0% | 74.1% | 68.7% | 5.3% | | 74.0% | 70.6% | 3.4% | |
| CREDIT QUALITY | | | | | | | | | |
| Net Charge-offs by Loan Type (in thousands) | | | | | | | | | |
| C&I | \$ 840 | \$ 16 | \$ 333 | \$ 507 | NM | \$ 856 | \$ 635 | \$ 221 | 34.8% |
| CRE | _ | _ | 2 | (2) | -100.0% | _ | 2 | (2) | -100.0% |
| | | | | | | | | | |
| Total commercial | 840 | 16 | 335 | 505 | NM | 856 | 637 | 219 | 34.4% |
| Consumer | 840 | 10 | 333 | 505 | INIM | 850 | 03/ | 219 | 54.4% |
| Home equity loans & lines of credit | _ | 160 | 62 | (62) | -100.0% | 160 | 261 | (101) | -38.7% |
| Residential mortgage | _ | 100 | 02 | (02) | -100.0% NM | 100 | 201 | (101) | -100.0% |
| Other loans | 60 | 32 | 23 | 37 | NM | 92 | 34 | 58 | -100.0% NM |
| Other loans | 00 | 32 | 23 | 57 | INIVI | 92 | 34 | 38 | INIVI |
| Total consumer | 60 | 192 | 85 | (25) | -29.4% | 252 | 316 | (64) | -20.3% |
| | | | | (20) | | | | (01) | 20.570 |
| Total Net Charge-offs | \$ 900 | \$ 208 | \$ 420 | \$ 480 | NM | \$ 1,108 | \$ 953 | \$ 155 | 16.3% |
| Net Charge-offs - annualized percentages | 4.000/ | 0.000/ | 0.400/ | 0.550/ | | | 0.4407 | 0.444/ | |
| C&I | 1.00% | 0.02% | 0.43% | 0.57% | | 0.52% | 0.41% | 0.11% | |
| CRE | 0.00% | 0.00% | 0.00% | 0.00% | | 0.00% | 0.00% | 0.00% | |
| | | | | | | | | | |
| Total commercial and commercial real estate | 0.63% | 0.01% | 0.27% | 0.36% | | 0.33% | 0.26% | 0.07% | |
| Consumer | | | | 0.00% | | | | 0.00% | |
| Home equity loans & lines of credit | 0.00% | 0.22% | 0.10% | -0.10% | | 0.11% | 0.21% | -0.10% | |
| Residential mortgage | 0.00% | 0.00% | 0.00% | 0.00% | | 0.00% | 0.01% | -0.01% | |
| Other loans | 3.02% | 1.61% | 1.32% | 1.70% | | 2.31% | 0.98% | 1.33% | |
| | | <u> </u> | | | | | | | |
| Total consumer | 0.03% | 0.10% | 0.05% | -0.02% | | 0.06% | 0.10% | -0.04% | |
| Tetel Net Change affe | 0.27% | 0.06% | 0.15% | 0.12% | | 0.17% | 0.17% | 0.00% | |
| Total Net Charge-offs | 0.27% | 0.06% | 0.15% | 0.12% | | 0.17% | 0.17% | 0.00% | |
| Non-performing Assets (NPA) (in millions) | | | | | | | | | |
| C&I | \$2 | \$ 2 | \$ 1 | \$ 1 | 100.0% | \$ 2 | \$ 1 | \$ 1 | 100.0% |
| CRE | 1 | _ | _ | 1 | NM | 1 | _ | 1 | NM |
| Residential mortgage | 3 | 1 | 1 | 2 | NM | 3 | 1 | 2 | NM |
| Total Non-accrual Loans | 6 | 3 | 2 | 4 | NM | 6 | 2 | 4 | NM |
| Renegotiated loans | 0 | 5 | 2 | - | NM | _ | 2 | - | NM |
| Relegonated ioans | | | | | | | | | INIVI |
| Total Non-performing Loans (NPL) | 6 | 3 | 2 | 4 | NM | 6 | 2 | 4 | NM |
| Other real estate, net (OREO) | _ | — | _ | _ | NM | _ | _ | _ | NM |
| Total Non-performing Assets | \$ 6 | \$ 3 | \$ 2 | \$ 4 | NM | \$ 6 | \$ 2 | \$ 4 | NM |
| | | | | | | | | | |
| Accruing loans past due 90 days or more | \$ 2 | \$ 4 | \$ 4 | \$ (2) | -50.0% | \$ 2 | \$ 4 | \$ (2) | -50.0% |
| Allowance for Loan and Lease Losses (ALLL) (eop) | \$ 9 | \$ 9 | \$ 7 | \$ 2 | 28.6% | \$ 9 | \$ 7 | \$ 2 | 28.6% |
| ALLL as a % of total loans and leases | 0.65% | 0.68% | 0.59% | 0.06% | | 0.65% | 0.59% | 0.06% | |
| ALLL as a % of NPLs | 150.0% | 300.0% | 350.0% | -200.0% | | 150.0% | 350.0% | -200.0% | |
| ALLL + OREO as a % of NPAs | 150.0% | 300.0% | 350.0% | -200.0% | | 150.0% | 350.0% | -200.0% | |
| NPLs as a % of total loans and leases | 0.43% | 0.23% | 0.17% | 0.26% | | 0.43% | 0.17% | 0.26% | |
| NPAs as a % of total loans and leases + OREO | 0.43% | 0.23% | 0.17% | 0.26% | | 0.43% | 0.17% | 0.26% | |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 NM, not a meaningful value.
 eop, end of period.

| | 2004 Second | 2004 | 2003 Second | 2Q04 vs. 2Q03 | | 2004 | 2003 | 1H04 vs. 1H03 | |
|--|----------------|-----------------|-----------------|---------------|-----------------|-----------------|-----------------|----------------|---------------|
| | | First | | Amount | % | 6 Months | is 6 Months | Amount | % |
| SUPPLEMENTAL DATA | | | | | | | | | |
| # employees - full-time equivalent (eop) | 708 | 703 | 695 | 13 | 1.9% | 708 | 695 | 13 | 1.9% |
| # licensed bankers (eop) | 698 | 701 | 685 | 13 | 1.9% | 698 | 685 | 13 | 1.9% |
| Brokerage and Insurance Income (in thousands) | | | | | | | | | |
| Mutual fund revenue | \$ 1,355 | \$ 1,610 | \$ 1,104 | \$ 251 | 22.7% | \$ 2,965 | \$ 1,954 | \$ 1,011 | 51.7% |
| Annuities revenue | 6,776 | 8,229 | 6,982 | (206) | -3.0% | 15,005 | 15,307 | (302) | -2.0% |
| 12b-1 fees | 600 1,280 | 535 1,290 | 525 1,187 | 75 93 | 14.3% 7.8% | 1,135 | 1,080 2,052 | 55 518 | 5.1% 25.2% |
| Discount brokerage commissions and other | 1,280 | 1,290 | 1,187 | 93 | 7.870 | 2,570 | 2,032 | 518 | 23.270 |
| Total retail investment sales | 10,011 | 11,664 | 9,798 | 213 | 2.2% | 21,675 | 20,393 | 1,282 | 6.3% |
| Investment banking fees | — | _ | — | — | NM | — | — | | NM |
| Insurance fees and revenue | 2,782 | 2,936 | 3,459 | (677) | -19.6% | 5,718 | 7,426 | (1,708) | -23.0% |
| Total Brokerage and Insurance Income | \$ 12,793 | \$ 14,600 | \$ 13,257 | \$ (464) | -3.5% | \$ 27,393 | \$ 27,819 | \$ (426) | -1.5% |
| | | <u> </u> | ······ | | | | | | |
| Fee sharing | 4,305 | 3,984 | 3,970 | 335 | 8.4% | 8,289 | 8,601 | (312) | -3.6% |
| Total Brokerage and Insurance Income (net of fee sharing) | \$ 8,488 | \$ 10,616 | \$ 9,287 | \$ (799) | -8.6% | \$ 19,104 | \$ 19,218 | \$ (114) | -0.6% |
| | | | | | | | | | |
| Mutual fund sales volume, excluding direct trades (in thousands) | \$ 58,002 | \$ 42,965 | \$ 75,995 | (17,993) | -23.7% | \$ 100,967 | \$ 128,263 | (27,296) | -21.3% |
| Annuities sales volume (in thousands) | 133,408 | 161,332 | 142,344 | (8,936) | -6.3% | 294,740 | 316,218 | (21,478) | -6.8% |
| Trust Services Income (in thousands) | | | | | | | | | |
| Personal trust revenue | \$ 8,423 | \$ 8,191 | \$ 7,502 | \$ 921 | 12.3% | \$ 16,614 | \$ 14,931 | \$ 1,683 | 11.3% |
| Huntington funds revenue | 5,195 | 5,230 | 4,805 | 390 | 8.1% | 10,425 | 9,480 | 945 403 | 10.0% |
| Institutional trust revenue Corporate trust revenue | 2,176 900 | 2,170 708 | 2,218 1,031 | (42) (131) | -1.9% -12.7% | 4,346 1,608 | 3,943 2,077 | (469) | 10.2% |
| Other trust revenue | | | | (151) | NM | | 2,077 | (40) | -22.070 NM |
| | | | | | | | | | |
| Total Trust Services Income | \$ 16,694 | \$ 16,299 | \$ 15,556 | \$ 1,138 | 7.3% | \$ 32,993 | \$ 30,431 | \$ 2,562 | 8.4% |
| Fee sharing | 211 | 268 | 262 | (51) | -19.5% | 479 | 494 | (15) | -3.0% |
| Total Trust Services Income (net of fee sharing) | \$ 16,483 | \$ 16,031 | \$ 15,294 | \$ 1,189 | 7.8% | \$ 32,514 | \$ 29,937 | \$ 2,577 | 8.6% |
| Assets Under Management (eop) (in billions) | | | | | _ | | | | _ |
| Personal trust | \$ 5.2 | \$ 5.0 | \$ 4.7 | \$ 0.4 | 9.4% | \$ 5.2 | \$ 4.7 | \$ 0.4 | 9.4% |
| Huntington funds | 2.9 | 3.0 | 3.0 | (0.1) | -1.8% | 2.9 | 3.0 | (0.1) | -1.8% |
| Institutional trust | 0.5 | 0.6 | 0.5 | 0.0 | 9.8% | 0.5 | 0.5 | 0.0 | 9.8% |
| Corporate trust | 0.0 | 0.0 | 0.1 | (0.1) | -62.0% | 0.0 | 0.1 | (0.1) | -62.0% |
| Haberer | 0.6 | 0.5 | 0.5 | 0.1 | 27.7% | 0.6 | 0.5 | 0.1 | 27.7% |
| Other | — | — | — | _ | NM | — | — | — | NM |
| Total Assets Under Management | \$ 9.3 | \$ 9.2 | \$ 8.8 | \$ 0.5 | 5.8% | \$ 9.3 | \$ 8.8 | \$ 0.5 | 5.8% |
| Total Assets Under Management | 3 3.3 | 3 9.2 | \$ 8.8 | 3 0.5 | 5.870 | \$ 9.3 | \$ 8.8 | \$ 0.5 | 5.870 |
| Total Trust Assets (eop) (in billions) | | | | | | | | | |
| Personal trust | \$ 8.9 | \$ 8.5 | \$ 7.7 | \$ 1.2 | 15.4% | \$ 8.9 | \$ 7.7 | \$ 1.2 | 15.4% |
| Huntington funds | 2.9 | 3.0 | 3.0 | (0.1) | -3.0% | 2.9 | 3.0 | (0.1) | -3.0% |
| Institutional trust | 23.9 3.5 | 24.6 3.1 | 20.3 3.2 | 3.6 0.3 | 17.7% 8.4% | 23.9 3.5 | 20.3 3.2 | 3.6 0.3 | 17.7% 8.4% |
| Corporate trust | 3.5 | 3.1 | 3.2 | 0.3 | 8.4% | 3.5 | 3.2 | 0.3 | 8.4% |
| Total Trust Assets | \$ 39.2 | \$ 39.1 | \$ 34.2 | \$ 5.0 | 14.5% | \$ 39.2 | \$ 34.2 | \$ 5.0 | 14.5% |
| | | | | | — | | | | |
| Mutual Fund Data # Huntington mutual funds (eop) | 29 | 24 | 24 | 5 | | 29 | 24 | 5 | |
| Sales penetration ⁽³⁾ | 5.9% | 6.5% | 6.8% | -0.9% | | 6.2% | 7.0% | -0.8% | |
| 1 | | | | | 0.80/ | | | | 7 70/ |
| Revenue penetration <i>(whole dollars)</i> ⁽⁴⁾ Profit penetration <i>(whole dollars)</i> ⁽⁵⁾ | \$ 3,270 | \$ 4,039 | \$ 3,298 | \$ (28) | -0.8% | \$ 3,700 | \$ 3,436 | \$ 264 | 7.7% |
| Profit penetration (whole dollars) ⁽³⁾ Average sales per licensed banker (whole dollars) annualized | 987 70,030 | 1,366 | 1,244 | (257) | -20.7% | 1,173 | 1,282 70,796 | (109) 4,978 | -8.5% 7.0% |
| Average sales per licensed banker (whole dollars) annualized Average revenue per licensed banker (whole dollars) annualized | 3,319 | 81,567 3,967 | 68,897 3,049 | 1,133 270 | 1.6% 8.9% | 75,774 3,639 | 3,203 | 4,978 | 13.6% |
| Average revenue per neenseu banker (whole abnurs) annualizea | 3,319 | 5,907 | 3,049 | 270 | 0.970 | 3,039 | 5,205 | 430 | 15.0% |

(1)

(3)

(4) (5)

Operating basis, see page 50 for definition. Sales (dollars invested) of mutual funds and annuities divided by bank's retail deposits. Investment program revenue per million of the bank's retail deposits. Contribution of investment program to pretax profit per million of the bank's retail deposits. Contribution is difference between program revenue and program expenses.

NM, not a meaningful value.

eop, end of period.

Treasury / Other

The Treasury/Other segment includes revenue and expense related to assets, liabilities, and equity that are not directly assigned or allocated to one of the other three business segments. Assets included in this segment include bank owned life insurance, investment securities, and mezzanine loans originated through Huntington Capital Markets.

Net interest income includes the net impact of administering Huntington's investment securities portfolios as part of overall liquidity management. A match-funded transfer pricing system is used to attribute appropriate funding interest income and interest expense to other business segments. As such, net interest income includes the net impact of any over or under allocations arising from centralized management of interest rate risk. Furthermore, net interest income includes the net impact of derivatives used to hedge interest rate sensitivity as well as net interest income related to Huntington Capital Markets loan activity.

Non-interest income includes fee income related to Huntington Capital Markets activities, and miscellaneous non-interest income not allocated to other business segments, including bank owned life insurance income. Fee income also includes MSR temporary impairment valuation recoveries or impairments, as well as any securities gains or losses, which are used to mitigate MSR valuation changes. Prior to 2004, changes in MSR temporary impairment valuations were reflected in the Regional Banking business segment, whereas securities gains or loss were reflected in the Treasury/Other segment. Since securities gains or losses are used to mitigate MSR valuation changes, and since this risk is managed centrally, for 2004 reporting both MSR valuation changes, as well as securities gains or losses, are reflected in Treasury/Other results.

Non-interest expense includes expenses associated with Huntington Capital Markets activities, as well as certain corporate administrative and other miscellaneous expenses not allocated to other business segments.

The provision for income taxes for each of the other business segments is calculated at a statutory 35% tax rate though the company's overall effective tax rate is lower. As a result, the provision for income taxes in Treasury/Other includes the difference between the actual effective tax rate and the statutory tax rate used to allocate income taxes to the other segments.

2004 Second Quarter versus 2003 Second Quarter

Treasury/Other net income declined \$15.2 million, or 45% from the same quarter last year. This reflected a combination of higher non-interest expense, lower non-interest income, and lower net interest income, partially offset by lower provision for income taxes.

Net interest income decreased \$2.0 million, or 10%, from the year-ago quarter. Driving this variance were a \$5.3 million increase in wholesale funding costs and \$7.5 million of lower revenue from derivatives activities, offset by \$1.1 million of higher interest income in the Capital Markets margin and \$10.0 million of higher interest income on securities.

The provision for credit losses was \$0.1 million higher than last year, attributable to increased Huntington Capital Markets lending activity.

Non-interest income was \$9.6 million, or 40%, lower than in the year-ago quarter, reflecting the impact of using securities losses to offset MSR impairment recoveries. In 2004, MSR impairment and recovery is reflected in the Treasury/Other segment, whereas in the year-ago quarter MSR impairment of \$6.4 million was reflected in Regional Banking. Other non-interest income was also down from the year-ago quarter by \$3.1 million, reflecting lower income from trading activities.

Non-interest expense for operational, administrative, and support groups, which was not specifically allocated to the other business segments was up \$12.3 million from last year. Driving this variance was \$5.5 million of higher incentive plan costs in the second quarter of 2004. Furthermore, \$5.8 million of other expense attributable to costs related to investments in partnerships generating tax benefits occurred in the second quarter of 2004.

The provision for income taxes declined \$8.8 million, reflecting lower pre-tax income, partially offset by a higher overall effective tax rate in the 2004 second quarter, compared with the year-ago quarter.
2004 Second Quarter versus 2004 First Quarter

Treasury/Other's net income was \$12.5 million, or 40%, lower than the first quarter. This reflected a combination of lower non-interest income, lower net interest income, and higher non-interest expense, partially offset by lower provision for credit losses.

The net interest margin for Treasury/Other was down \$7.0 million, or 27%. Driving this variance was lower net earnings from Huntington's transfer pricing system to centrally manage interest rate risk. These earnings are allocated to the other lines of business units'.

Non-interest income was \$7.1 million lower in the second quarter compared with the first quarter. Securities losses, offset by MSR impairment recovery, drive most of this reduction. Furthermore, investment banking activities from Huntington Capital Markets contributed \$2.4 million less revenue in the second quarter, than in the first quarter.

Non-interest expense for Treasury/Other increased \$2.2 million, or 13%, from the first quarter. Second quarter expenses include \$5.8 million of other expense attributable to costs related to investments in partnerships generating tax benefits. Offsetting this variance is a higher net amount of expenses allocated to the other business segments.

The provision for credit losses decline \$2.3 million, reflecting lower NPAs.

The provision for income taxes decreased \$1.5 million, reflecting the benefit of the company's lower overall effective tax rate versus the 35% effective tax rate used to allocate provision for income taxes to each line of business, partially offset by an increase in the company's higher overall effective tax rate in the current quarter compared with the first quarter.

2004 First Six Months versus 2003 First Six Months

Treasury/Other net income for the first six months of 2004 declined \$11.8 million, or 19%, from the comparable year-ago period. This reflected a combination of factors primarily consisting of higher non-interest expense, lower non-interest income, and an increase in provision for credit losses, partially offset by lower provision for income taxes.

Net interest income decreased \$0.3 million, or 1%, from the comparable year-ago period. Driving this variance were a \$10.1 million increase in wholesale funding costs and \$14.8 million of lower revenue from derivatives activities, offset by higher interest income in the Huntington Capital Markets margin and \$21 million of higher interest income on securities.

The provision for credit losses was \$2.5 million higher than the year-ago period, attributable to increased Huntington Capital Markets lending activity.

Non-interest income was \$3.6 million, or 9%, lower than in the year-ago period, reflecting the impact of using securities losses to offset MSR impairment recoveries.

Non-interest expense for operational, administrative, and support groups, which was not specifically allocated to other business segments increased \$14.7 million from last year, including \$5.5 million of higher incentive plan costs, \$5.8 million of other expense attributable to costs related to investments in partnerships generating tax benefits, and higher sales incentives of \$2.4 million in Huntington Capital Markets.

The provision for income taxes declined \$9.4 million, reflecting the benefit of the company's lower overall effective tax rate versus the 35% effective tax rate used to allocate provision for income taxes to each line of business, partially offset by the company's higher total effective tax rate in the first six months of 2004 versus the comparable year-ago period.



Table 21 - Treasury/Other⁽¹⁾

| | 2004 | 2004 | 2003 2Q04 vs. 2Q03 | | 2Q03 | 2004 2003 | | 1H04 vs. 1H03 | | |
|--|-----------|------------|--------------------|-------------|--------|---------------|-----------|---------------|--------|--|
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % | |
| INCOME STATEMENT (in thousands) | | | . <u> </u> | | | | | | | |
| Net Interest Income | \$ 18,557 | \$ 25,543 | \$20,511 | \$ (1,954) | -9.5% | \$ 44,100 | \$ 44,446 | \$ (346) | -0.8% | |
| Provision for credit losses | 61 | 2,393 | (66) | 127 | NM | 2,454 | (60) | 2,514 | NM | |
| Net Interest Income After Provision for | | | | | | | | | | |
| Credit Losses | 18,496 | 23,150 | 20,577 | (2,081) | -10.1% | 41,646 | 44,506 | (2,860) | -6.4% | |
| Service charges on deposit accounts | 7 | 10 | 10 | (3) | -30.0% | 17 | 22 | (5) | -22.7% | |
| Brokerage and insurance income | 185 | 215 | 216 | (31) | -14.4% | 400 | 439 | (39) | -8.9% | |
| Mortgage banking | 10,199 | (10,200) | _ | 10,199 | NM | (1) | | (1) | NM | |
| Bank Owned Life Insurance income | 11,309 | 10,485 | 11,043 | 266 | 2.4% | 21,794 | 22,180 | (386) | -1.7% | |
| Other | 2,701 | 5,904 | 5,818 | (3,117) | -53.6% | 8,605 | 8,887 | (282) | -3.2% | |
| | | - <u>,</u> | | (-) -) | | | | | | |
| Total Non-Interest Income Before Securities Gains | 24,401 | 6,414 | 17,087 | 7,314 | 42.8% | 30,815 | 31,528 | (713) | -2.3% | |
| Securities gains | (9,949) | 15,090 | 6,921 | (16,870) | NM | 5,141 | 8,076 | (2,935) | -36.3% | |
| Total Non-Interest Income | 14,452 | 21,504 | 24,008 | (9,556) | -39.8% | 35,956 | 39,604 | (3,648) | -9.2% | |
| | | | | | | | | | | |
| Total Non-Interest Expense | 19,986 | 17,732 | 7,679 | 12,307 | NM | 37,718 | 23,050 | 14,668 | 63.6% | |
| Income Before Provision for Income Taxes | 12,962 | 26,922 | 36,906 | (23,944) | -64.9% | 39,884 | 61,060 | (21,176) | -34.7% | |
| Provision for income taxes ⁽²⁾ | (5,809) | (4,352) | 2,985 | (8,794) | NM | (10,161) | (758) | (9,403) | NM | |
| Net Income - Operating ⁽¹⁾ | \$ 18,771 | \$ 31,274 | \$ 33,921 | \$ (15,150) | -44.7% | \$ 50,045 | \$61,818 | \$(11,773) | -19.0% | |
| Revenue - Fully Taxable Equivalent (FTE) | | | | | | | | | | |
| Net Interest Income | \$18,557 | \$ 25,543 | \$20,511 | \$ (1,954) | -9.5% | \$ 44,100 | \$ 44,446 | \$ (346) | -0.8% | |
| Tax Equivalent Adjustment (2) | 2,660 | 2,765 | 1,755 | 905 | 51.6% | 5,425 | 3,505 | 1,920 | 54.8% | |
| Net Interest Income (FTE) | 21,217 | 28,308 | 22,266 | (1,049) | -4.7% | 49,525 | 47,951 | 1,574 | 3.3% | |
| Non-Interest Income | 14,451 | 21,504 | 24,008 | (9,557) | -39.8% | 35,955 | 39,604 | (3,649) | -9.2% | |
| Total Revenue (FTE) | \$ 35,668 | \$ 49,812 | \$46,274 | \$ (10,606) | -22.9% | \$ 85,480 | \$ 87,555 | \$ (2,075) | -2.4% | |
| Total Revenue Excluding Securities Gains | | | | | _ | | | | _ | |
| (FTE) | \$45,617 | \$ 34,722 | \$ 39,353 | \$ 6,264 | 15.9% | \$ 80,339 | \$ 79,479 | \$ 860 | 1.1% | |
| SELECTED AVERAGE BALANCES (in | | | | | | | | | | |
| millions) | | | · · · · | | 10 | | | | | |
| Securities | \$ 5,233 | \$ 5,045 | \$ 3,495 | \$ 1,737 | 49.7% | \$ 5,139 | \$ 3,310 | \$ 1,829 | 55.3% | |
| Loans: | | | • | | | · · · · | | . | | |
| C&I | \$ 102 | \$ 70 | \$ 52 | \$ 50 | 96.2% | \$ 86 | \$ 17 | \$ 69 | NM | |
| CRE | - | | | - | 201 | - | 10 | (10) | | |
| Construction | 1 | 0 | (1) | 2 | NM | 1 | 19 | (18) | -94.7% | |
| Commercial | 59 | 62 | 61 | (2) | -3.3% | 61 | 75 | (14) | -18.7% | |
| Total Loans | \$ 162 | \$ 132 | \$ 112 | \$ 50 | 44.6% | \$ 148 | \$ 111 | \$ 37 | 33.3% | |
| | | | | | _ | | | | | |
| Deposits: | | | | | | | | | | |
| Brokered time deposits and negotiable | 1 525 | 1 007 | 1 0 4 1 | 107 | 40.00/ | 1 000 | 1 100 | (0) | 50 10/ | |
| CDs | 1,737 | 1,907 | 1,241 | 496 | 40.0% | 1,822 | 1,198 | 624 | 52.1% | |
| Foreign time deposits | 97 | 101 | 80 | 17 | 21.3% | 99 | 138 | (39) | -28.3% | |
| Total Deposits | \$ 1,834 | \$ 2,008 | \$ 1,321 | \$ 513 | 38.8% | \$ 1,921 | \$ 1,336 | \$ 585 | 43.8% | |
| | | | | | | | | | | |

Operating basis, see page 50 for definition.
 Reconciling difference between company's actual effective tax rate and 35% tax rate allocated to each business segment.
 NM, not a meaningful value.

Table 21 - Treasury/Other⁽¹⁾

| Table 21 - Treasury/Other ⁽¹⁾ | | | | • • • • • | | | | | |
|---|----------|--------|--------|-----------|---------|----------|----------|----------|---------|
| | 2004 | 2004 | 2003 | 2Q04 vs. | 2Q03 | 2004 | 2003 | 1H04 vs. | 1H03 |
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| PERFORMANCE METRICS | | | | | | | | | |
| Return on average assets | 1.07% | 1.87% | 2.62% | -1.55% | | 1.45% | 2.49% | -1.04% | |
| Return on average equity | 9.8% | 17.8% | 22.9% | -13.1% | | 13.6% | 20.3% | -6.7% | |
| Net interest margin | 1.53% | 2.13% | 2.37% | -0.84% | | 1.82% | 2.71% | -0.89% | |
| Efficiency ratio | 43.8% | 51.1% | 19.5% | 24.3% | | 46.9% | 29.0% | 17.9% | |
| CREDIT QUALITY | | | | | | | | | |
| Net Charge-offs by Loan Type (in thousands) | | | | | | | | | |
| C&I | \$ (23) | \$ — | \$ — | \$ (23) | NM | \$ (23) | \$ — | \$ (23) | NM |
| CRE | 1,999 | 1 | | 1,999 | NM | 2,000 | — | 2,000 | NM |
| | | | | | | | | | |
| Total commercial | 1,976 | 1 | | 1,976 | NM | 1,977 | | 1,977 | NM |
| Total Net Charge-offs | \$ 1,976 | \$ 1 | \$ — | \$ 1,976 | NM | \$ 1,977 | \$ — | \$ 1,977 | NM |
| Not Charge offs annualized nercontages | | | | | | | | | |
| Net Charge-offs - annualized percentages C&I | -0.09% | 0.00% | 0.00% | -0.09% | | -0.05% | 0.00% | -0.05% | |
| CRE | 13.40% | 0.00% | 0.00% | 13.40% | | 6.47% | 0.00% | 6.47% | |
| | 13.40 /0 | 0.0170 | 0.0070 | 13.4070 | | 0.4770 | 0.0070 | 0.4770 | |
| Total commercial | 4.91% | 0.00% | 0.00% | 4.91% | | 2.68% | 0.00% | 2.68% | |
| Total Net Charge-offs | 4.91% | 0.00% | 0.00% | 4.91% | | 2.68% | 0.00% | 2.68% | |
| | | | | | | | | | |
| Non-performing Assets (NPA) (in millions) | | | | | | | | | |
| C&I | \$ — | \$ 1 | \$ 1 | \$ (1) | -100.0% | \$ — | \$ 1 | \$ (1) | -100.0% |
| CRE | 8 | 11 | — | 8 | NM | 8 | — | 8 | NM |
| Total Non-accrual Loans | 8 | 12 | 1 | 7 | NM | 8 | 1 | 7 | NM |
| Renegotiated loans | — | — | — | — | NM | _ | — | — | NM |
| | | | | | | | | | |
| Total Non-performing Loans (NPL) | 8 | 12 | 1 | 7 | NM | 8 | 1 | 7 | NM |
| Other real estate, net (OREO) | (1) | (1) | | (1) | NM | (1) | | (1) | NM |
| Total Non-performing Assets | \$ 7 | \$ 11 | \$ 1 | \$6 | NM | \$ 7 | \$ 1 | \$6 | NM |
| | | | | | | | | | |
| Accruing loans past due 90 days or more | \$ — | \$ — | \$ — | \$ — | NM | \$ — | \$ — | \$ — | NM |
| Allowance for Loan and Lease Losses | | | | | | | | | |
| (ALLL)(eop) | \$ 42 | \$ 77 | \$ 39 | \$ 3 | 7.7% | \$ 42 | \$ 39 | \$ 3 | 7.7% |
| ALLL as a % of total loans and leases | 29.79% | 53.47% | 20.97% | 8.82% | | 29.79% | 20.97% | 8.82% | |
| ALLL as a % of NPLs | NM | NM | NM | NM | | NM | NM | NM | |
| ALLL + OREO as a % of NPAs | NM | NM | NM | NM | | NM | NM | NM | |
| NPLs as a % of total loans and leases | 5.67% | 8.33% | 0.54% | 5.13% | | 5.67% | 0.54% | 5.13% | |
| NPAs as a % of total loans and leases + OREO | 5.00% | 7.69% | 0.54% | 4.46% | | 5.00% | 0.54% | 4.46% | |
| SUPPLEMENTAL DATA | | | | | | | | | |
| # employees - full-time equivalent (eop) | 1,975 | 1,956 | 2,016 | (41) | -2.0% | 1,975 | 2,016 | (41) | -2.0% |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 NM, not a meaningful value.
 eop, end of period.

Table 22 - Total Company⁽¹⁾

| | 2004 | 2004 | | 2003 | 2Q04 vs. 2 | Q03 | | 2004 | | 2003 | 1H04 vs. 1 | H03 |
|--|--------------------|-------------------|----|------------------|----------------------|--------------|----|------------------|----|------------------|---------------------|----------|
| | Second | First | | Second | Amount | % | 6 | Months | _ | 6 Months | Amount | % |
| INCOME STATEMENT (in thousands) | | | | | | | | | | | | |
| Net Interest Income | \$ 222,563 | \$ 222,685 | \$ | 202,441 | \$ 20,122 | 9.9% | \$ | 445,248 | \$ | 404,200 | \$ 41,048 | 10.2% |
| Provision for credit losses | 5,027 | 25,596 | | 49,193 | (44,166) | -89.8% | | 30,623 | | 86,037 | (55,414) | -64.4% |
| Net Interest Income After Provision for Credit Losses | 217,536 | 197,089 | | 153,248 | 64,288 | 42.0% | | 414,625 | | 318,163 | 96,462 | 30.3% |
| Operating lease income Service charges on deposit | 78,706 | 88,867 | | 128,574 | (49,868) | -38.8% | | 167,573 | | 266,767 | (99,194) | -37.2% |
| accounts | 43,596 | 41,837 | | 40,914 | 2,682 | 6.6% | | 85,433 | | 80,783 | 4,650 | 5.8% |
| Brokerage and insurance income | 13,523 | 15,197 | | 14,196 | (673) | -4.7% | | 28,720 | | 29,693 | (973) | -3.3% |
| Frust services | 16,708 | 16,323 | | 15,580 | 1,128 | 7.2% | | 33,031 | | 30,491 | 2,540 | 8.3% |
| Mortgage banking Bank Owned Life Insurance | 23,322 | (4,296) | | 7,185 | 16,137 | NM | | 19,026 | | 18,310 | 716 | 3.9% |
| income | 11,309 | 10,485 | | 11,043 | 266 | 2.4% | | 21,794 | | 22,180 | (386) | -1.7% |
| Other service charges and fees | 10,645 | 9,513 | | 11,372 | (727) | -6.4% | | 20,158 | | 21,710 | (1,552) | -7.1% |
| Other | 24,659 | 25,619 | | 27,704 | (3,045) | -11.0% | | 50,278 | | 48,105 | 2,173 | 4.5% |
| | | | _ | ., | (-)) | | | | - | | , | |
| Total Non-Interest Income Before | 222 4/8 | 202 545 | | 256 569 | (24,100) | 12 20/ | | 426 012 | | 518.020 | (02.02() | 17.00/ |
| Securities Gains | 222,468 (9,230) | 203,545 15,090 | | 256,568 6,887 | (34,100) (16,117) | -13.3% NM | | 426,013 5,860 | | 518,039 8,085 | (92,026) (2,225) | -17.8% |
| securities gains | (9,230) | 15,090 | | 0,887 | (10,117) | | | 3,000 | _ | 8,085 | (2,223) | -27.370 |
| Fotal Non-Interest Income | 213,238 | 218,635 | | 263,455 | (50,217) | -19.1% | | 431,873 | _ | 526,124 | (94,251) | -17.9% |
| Operating lease expense | 62,563 | 70,710 | | 102,939 | (40,376) | -39.2% | | 133,273 | | 214,527 | (81,254) | -37.9% |
| Personnel costs | 119,715 | 121,624 | | 105,242 | 14,473 | 13.8% | | 241,339 | | 218,331 | 23,008 | 10.5% |
| Other | 99,875 | 93,320 | | 94,167 | 5,708 | 6.1% | | 193,195 | | 185,969 | 7,226 | 3.9% |
| Total Non-Interest Expense | 282,153 | 285,654 | | 302,348 | (20,195) | -6.7% | | 567,807 | | 618,827 | (51,020) | -8.2% |
| Income Before Provision for | | | | | | | | | | | | |
| Income Taxes | 148,621 | 130,070 | | 114,355 | 34,266 | 30.0% | | 278,691 | | 225,460 | 53,231 | 23.6% |
| Provision for income taxes | 41,672 | 31,750 | | 30,092 | 11,580 | 38.5% | | 73,422 | | 56,783 | 16,639 | 29.3% |
| Net Income - Operating ⁽¹⁾ | \$ 106,949 | \$ 98,320 | \$ | 84,263 | \$ 22,686 | 26.9% | \$ | 205,269 | \$ | 168,677 | \$ 36,592 | 21.7% |
| Revenue - Fully Taxable Equivalent (FTE) | | | | | | | - | | _ | | | |
| Net Interest Income | \$ 222,563 | \$ 222,685 | \$ | 202,441 | \$ 20,122 | 9.9% | \$ | 445,248 | \$ | 404,200 | \$ 41,048 | 10.2% |
| Tax Equivalent Adjustment (2) | 2,919 | 3,023 | | 2,076 | 843 | 40.6% | | 5,942 | | 4,172 | 1,770 | 42.4% |
| Net Interest Income (FTE) | 225,482 | 225,708 | | 204,517 | 20,965 | 10.3% | | 451,190 | | 408,372 | 42,818 | 10.5% |
| Non-Interest Income | 213,238 | 218,635 | | 263,455 | (50,217) | -19.1% | | 431,873 | | 526,124 | (94,251) | -17.9% |
| Fotal Revenue (FTE) | \$ 438,720 | \$ 444,343 | \$ | 467,972 | \$ (29,252) | -6.3% | \$ | 883,063 | \$ | 934,496 | \$ (51,433) | -5.5% |
| Fotal Revenue Excluding | | | - | | | | | | _ | | | |
| Securities Gains (FTE) | \$ 447,950 | \$ 429,253 | \$ | 461,085 | \$ (13,135) | -2.8% | \$ | 877,203 | \$ | 926,411 | \$ (49,208) | -5.3% |
| SELECTED AVERAGE BALANCES (in millions) | | | | | | | | | | | | |
| Loans: | 0 = = = = = = = | 0 5065 | ¢ | 5 (0) | ¢ (00) | 1 (0/ | đ | E 451 | ሱ | E. (05 | ¢ (174) | 2.10/ |
| C&I CRE | \$ 5,536 | \$ 5,365 | \$ | 5,626 | \$ (90) | -1.6% | \$ | 5,451 | \$ | 5,625 | \$ (174) | -3.1% |
| CRE Construction | 1,322 | 1,322 | | 1,239 | 83 | 6.7% | | 1,322 | | 1,214 | 108 | 8.9% |
| Commercial | 2,906 | 2,876 | | 2,621 | 285 | 10.9% | | 2,891 | | 2,593 | 298 | 11.5% |
| Consumer | _, | _,,,,, | | _, | 200 | | | ., | | -,- > 0 | 270 | / 0 |
| Auto leases - indirect | 2,139 | 1,988 | | 1,306 | 833 | 63.8% | | 2,064 | | 1,153 | 911 | 79.0% |
| Auto loans - indirect | 2,337 | 3,041 | | 2,830 | (493) | -17.4% | | 2,689 | | 2,954 | (265) | -9.0% |
| Home equity loans & lines | | • | | | | aa a= : | | | | | | . |
| of credit | 4,142 | 3,880 | | 3,359 | 783 | 23.3% | | 4,011 | | 3,298 | 713 | 21.6% |
| Residential mortgage Other loans | 2,986 399 | 2,674 | | 1,887 | 1,099 | 58.2% | | 2,830 | | 1,859 | 971 | 52.2% |
| Other ioans | | 356 | | 379 | 20 | 5.3% | | 377 | | 383 | (6) | -1.6% |
| Total Consumer | 12,003 | 11,939 | | 9,761 | 2,242 | 23.0% | | 11,971 | | 9,647 | 2,324 | 24.1% |
| Total Loans | \$ 21,767 | \$ 21,502 | \$ | 19,247 | \$ 2,520 | 13.1% | \$ | 21,635 | \$ | 19,079 | \$ 2,556 | 13.4% |
| Operating lease assets | \$ 977 | \$ 1,166 | \$ | 1,802 | \$ (825) | -45.8% | \$ | 1,070 | \$ | 1,937 | \$ (867) | -44.8% |
| Deposits: | | | | | | | | | | | | |

| Non-interest bearing deposits | \$ 3,223 | \$ 3,017 | \$ 3,0 | 46 \$ 1 | 77 5.8% | \$ | 3,120 | \$ 2 | .,984 | \$ 136 | 4.6% |
|-------------------------------|-----------|-----------|---------|-----------|------------|----|--------|-------|-------|----------|--------|
| Interest bearing demand | | | | | | | | | | | |
| deposits | 7,168 | 6,609 | 6,1 | 00 1,0 | 68 17.5% | | 6,889 | 5 | ,868 | 1,021 | 17.4% |
| Savings deposits | 2,839 | 2,819 | 2,8 | 04 | 35 1.2% | | 2,829 | 2 | ,788 | 41 | 1.5% |
| Domestic time deposits | 3,795 | 3,824 | 4,2 | 79 (4 | 84) -11.3% | | 3,810 | 4 | ,347 | (537) | -12.4% |
| Brokered time deposits and | | | | | | | | | | | |
| negotiable CDs | 1,737 | 1,907 | 1,2 | 41 4 | 96 40.0% | | 1,822 | 1 | ,198 | 624 | 52.1% |
| Foreign time deposits | 542 | 549 | 4 | 26 1 | 16 27.2% | | 545 | | 470 | 75 | 16.0% |
| | | | | | | | | | | | |
| Total Deposits | \$ 19,304 | \$ 18,725 | \$ 17,8 | 96 \$ 1,4 | 08 7.9% | \$ | 19,015 | \$ 17 | ,655 | \$ 1,360 | 7.7% |
| _ | | | | | | _ | | - | | | |

Operating basis, see page 50 for definition.
 Calculated assuming a 35% tax rate.
 NM, not a meaningful value.

Table 22 - Total Company⁽¹⁾

| Table 22 - Total Company ⁽¹⁾ | 2004 | 2004 | 2003 | 2Q04 vs. 2 | Q03 | 2004 | 2003 | 1H04 vs. 1 | H03 |
|---|----------------|-------------|----------------|------------------|--------|----------------|----------------|------------------|--------|
| | Second | First | Second | Amount | % | 6 Months | 6 Months | Amount | % |
| PERFORMANCE METRICS | | | | | | | | | _ |
| Return on average assets | 1.37% | 1.28% | 1.20% | 0.17% | | 1.33% | 1.23% | 0.10% | |
| Return on average equity | 18.5% | 17.4% | 15.7% | 2.8% | | 17.9% | 15.8% | 2.1% | |
| Net interest margin | 3.29% | 3.36% | 3.47% | -0.18% | | 3.32% | 3.52% | -0.20% | |
| Efficiency ratio | 63.0% | 66.5% | 65.6% | -2.6% | | 64.7% | 66.8% | -2.1% | |
| CREDIT QUALITY (in thousands) | | | | | | | | | |
| Net Charge-offs by Loan Type | | | | | | | | | |
| C&I | \$ (2,803) | \$ 5,956 | \$ 26,546 | \$ (29,349) | NM | \$ 3,153 | \$ 41,450 | \$ (38,297) | -92.4% |
| CRE | 2,940 | 1,637 | 607 | 2,333 | NM | 4,577 | 1,153 | 3,424 | NM |
| Total commercial | 137 | 7,593 | 27,153 | (27,016) | -99.5% | 7,730 | 42,603 | (34,873) | -81.9% |
| Consumer | | | | | | | | | |
| Auto leases | 2,159 | 3,159 | 1,422 | 737 | 51.8% | 5,318 | 2,342 | 2,976 | NM |
| Auto loans | 5,604 | 13,422 | 7,524 | (1,920) | -25.5% | 19,026 | 18,147 | 879 | 4.8% |
| Home equity loans & lines of credit | 3,019 | 3,116 | 3,671 | (652) | -17.8% | 6,135 | 7,724 | (1,589) | -20.6% |
| Residential mortgage | 302 | 316 | 267 | 35 275 | 13.1% | 618 | 412 | 206 | 50.0% |
| Other loans | 1,294 | 1,021 | 1,019 | 275 | 27.0% | 2,315 | 2,664 | (349) | -13.1% |
| Total consumer | 12,378 | 21,034 | 13,903 | (1,525) | -11.0% | 33,412 | 31,289 | 2,123 | 6.8% |
| Total Net Charge-offs | \$ 12,515 | \$ 28,627 | \$ 41,056 | \$ (28,541) | -69.5% | \$ 41,142 | \$ 73,892 | \$ (32,750) | -44.3% |
| Total Net Charge ons | \$ 12,515 | \$ 20,027 | \$ 41,050 | \$ (20,541) | 09.570 | \$ 41,142 | \$ 75,672 | \$ (52,750) | 44.570 |
| Net Charge-offs - annualized percentages | | | | | | | | | |
| C&I | -0.20% | 0.44% | 1.89% | -2.09% | | 0.12% | 1.47% | -1.35% | |
| CRE | 0.28% | 0.16% | 0.06% | 0.22% | | 0.22% | 0.06% | 0.16% | |
| Total commercial | 0.01% | 0.32% | 1.14% | -1.13% | | 0.16% | 0.90% | -0.74% | |
| Consumer | | | | | | | | | |
| Auto leases | 0.40% | 0.64% | 0.44% | -0.04% | | 0.52% | 0.40% | 0.12% | |
| Auto loans | 0.96% | 1.77% | 1.06% | -0.10% | | 1.42% | 1.23% | 0.19% | |
| Home equity loans & lines of credit | 0.29% 0.04% | 0.32% | 0.44% 0.06% | -0.15% -0.02% | | 0.31% 0.04% | 0.47% 0.04% | -0.17% 0.00% | |
| Residential mortgage Other loans | 1.30% | 0.05% 1.15% | 1.08% | 0.22% | | 1.23% | 1.39% | -0.16% | |
| Otici Ioalis | 1.50 /0 | 1.1570 | 1.0070 | 0.2270 | | 1.23 /0 | 1.5770 | -0.1070 | |
| Total consumer | 0.41% | 0.70% | 0.57% | -0.16% | | 0.56% | 0.65% | -0.09% | |
| Total Net Charge-offs | 0.23% | 0.53% | 0.85% | -0.62% | | 0.38% | 0.77% | -0.39% | |
| Total Net Charge-ons | 0.23 /0 | 0.5570 | 0.8570 | -0.0270 | | 0.58 /0 | 0.7776 | -0.5976 | |
| Non-performing Assets (NPA) (in millions) | | | | | | | | | |
| C&I | \$ 32 | \$ 45 | \$ 86 | \$ (54) | -62.8% | \$ 32 | \$ 86 | \$ (54) | -62.8% |
| CRE | 16 | 20 | 22 | (6) | -27.3% | 16 | 22 | (6) | -27.3% |
| Residential mortgage | 14 | 12 | 12 | 2 | 16.7% | 14 | 12 | 2 | 16.7% |
| Total Non-accrual Loans | 62 | 77 | 120 | (58) | -48.3% | 62 | 120 | (58) | -48.3% |
| Renegotiated loans | _ | _ | | (56) | NM | | | (50) | NM |
| | | | | | | | | | |
| Total Non-performing Loans (NPL) | 62 | 77 | 120 | (58) | -48.3% | 62 | 120 | (58) | -48.3% |
| Other real estate, net (OREO) | 13 | 15 | 14 | (1) | -7.1% | 13 | 14 | (1) | -7.1% |
| Total Non-performing Assets | \$ 75 | \$ 92 | \$ 134 | \$ (59) | -44.0% | \$ 75 | \$ 134 | \$ (59) | -44.0% |
| | | | | | | | | . () | |
| Accruing loans past due 90 days or more | \$ 51 | \$ 60 | \$ 55 | \$ (4) | -7.3% | \$ 51 | \$ 55 | \$ (4) | -7.3% |
| Allowance for Loan and Lease Losses (ALLL) | | | | | | | | | |
| (eop) | \$ 287 | \$ 295 | \$ 308 | \$ (21) | -6.8% | \$ 287 | \$ 308 | \$ (21) | -6.8% |
| ALLL as a % of total loans and leases | 1.32% | 1.39% | 1.61% | -0.29% | | 1.32% | 1.61% | -0.29% | |
| ALLL as a % of NPLs | 464.5% | 383.0% | 256.1% | 208.4% | | 464.5% | 256.1% | 208.4% | |
| ALLL + OREO as a % of NPAs | 400.0% | 337.0% | 240.3% | 159.7% | | 400.0% | 240.3% | 159.7% | |
| NPLs as a % of total loans and leases NPAs as a % of total loans and leases + OREO | 0.28% 0.34% | 0.36% | 0.63% | -0.35% | | 0.28% 0.34% | 0.63% | -0.35% -0.36% | |
| | 0.34 70 | 0.43% | 0.70% | -0.30% | | 0.34 70 | 0.70% | -0.3070 | |
| SUPPLEMENTAL DATA | 0.5.15 | | | | 0 | 0.5.15 | | | |
| # employees - full-time equivalent | 8,045 | 7,915 | 7,986 | 59 | 0.7% | 8,045 | 7,986 | 59 | 0.7% |
| | | | | | | | | | |

Operating basis, see page 50 for definition.
 NM, not a meaningful value.
 eop, end of period.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Quantitative and qualitative disclosures for the current period are found beginning on page 45 of this report, which includes changes in market risk exposures from disclosures presented in Huntington's 2003 Form 10-K.

Item 4. Controls and Procedures

Huntington's management, with the participation of its Chief Executive Officer and the Chief Financial Officer, evaluated the effectiveness of Huntington's disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act) as of the end of the period covered by this report. Based upon such evaluation, Huntington's Chief Executive Officer and Chief Financial Officer have concluded that, as of the end of such period, Huntington's disclosure controls and procedures are effective.

There have not been any changes in Huntington's internal control over financial reporting (as such term is defined in Rules 13a-15(f) ad 15d-15(f) under the Exchange Act) during the fiscal quarter to which this report relates that have materially affected, or are reasonably likely to materially affect, Huntington's internal control over financial reporting.

PART II. OTHER INFORMATION

In accordance with the instructions to Part II, the other specified items in this part have been omitted because they are not applicable or the information has been previously reported.

Item 2. Changes in Securities and Use of Proceeds

(e)

| Period | Total Number of Shares Purchased | Average Price Paid Per Share | Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs | Maximum Number of Shares that May Yet Be Purchased Under the Plans or Programs ⁽¹⁾ |
|---------------------------------|-------------------------------------|---------------------------------|--|---|
| April 1, 2004 to April 30, 2004 | _ | _ | | 7,500,000 |
| May 1, 2004 to May 31, 2004 | — | | | 7,500,000 |
| June 1, 2004 to June 30, 2004 | — | — | | 7,500,000 |
| Total | — | | — | 7,500,000 |

(1) Information is as of the end of the Period. On January 16, 2003, the Registrant announced that its board of directors had authorized the repurchase from time to time of 8,000,000 shares of the Registrant's common stock (the "2003 Repurchase Program"). The 2003 Repurchase Program did not have an expiration date. A total of 4,100,000 shares had been repurchased under the 2003 Repurchase Program in 2003. No shares were repurchased in the 2004 first quarter under the 2003 Repurchase Program or otherwise. On April 27, 2004, the Registrant announced that its board of directors had terminated the 2003 Repurchase Program and had authorized a new program for the repurchase in the open market or through privately negotiated transactions of 7,500,000 share of the Registrant's common stock (the "2004 Repurchase Program"). The 2004 Repurchase Program does not have an expiration date. The Registrant also intends to repurchase approximately 2.5 million shares associated with the Unizan Financial Corp. merger, and the remaining 5,000,000 shares would be repurchased from time to time whenever market conditions or other factors deem that to be appropriate.

Item 4. Submission of Matters to a Vote of Security Holders

Huntington held its annual meeting of shareholders on April 27, 2004. At this meeting, the shareholders approved the following management proposals:

| | | For | Against | Abstain/ Withheld |
|----|---|-------------|------------|----------------------|
| 1. | Election of directors to serve as Class II Directors until the 2007 Annual Meeting of Shareholders as follows: | | | |
| | Karen A. Holbrook | 194,112,217 | | 4,488,001 |
| | David P. Lauer | 193,206,154 | | 5,394,064 |
| | Kathleen H. Ransier | 191,772,564 | | 6,827,653 |
| 2. | Election of directors to serve as Class III Directors until the 2005 Annual Meeting of Shareholders as follows: | | | |
| | David L. Porteous | 193,539,668 | | 5,060,550 |
| 3. | Proposal to approve the Management Incentive Plan, as amended and restated. | 138,274,325 | 17,081,492 | 3,785,422 |
| 4. | Proposal to approve the 2004 Stock and Long-Term Incentive Plan. | 130,994,537 | 24,550,481 | 3,596,216 |
| 5. | Ratification of Deloitte & Touche LLP as independent auditors Huntington for the year 2004. | 191,730,116 | 4,663,300 | 2,206,802 |

Item 6. Exhibits and Reports on Form 8-K

(a) Exhibits

- 3(i)(a). Articles of Restatement of Charter, Articles of Amendment to Articles of Restatement of Charter, and Articles Supplementary previously filed as Exhibit 3(i) to Annual Report on Form 10-K for the year ended December 31, 1993, and incorporated herein by reference.
- (i)(b). Articles of Amendment to Articles of Restatement of Charter previously filed as Exhibit 3(i)(c) to Quarterly Report on Form 10-Q for the quarter ended March 31, 1998, and incorporated herein by reference.
- (ii)(a). Amended and Restated Bylaws as of July 16, 2002 previously filed as Exhibit 3(ii) to Quarterly Report on form 10-Q for the quarter ended June 30, 2003, and incorporated herein by reference.
- 4.(a). Instruments defining the Rights of Security Holders reference is made to Articles Fifth, Eighth, and Tenth of Articles of Restatement of Charter, as amended and supplemented. Instruments defining the rights of holders of long-term debt will be furnished to the Securities and Exchange Commission upon request.
- (b). Rights Plan, dated February 22, 1990, between Huntington Bancshares Incorporated and The Huntington National Bank (as successor to The Huntington Trust Company, National Association) previously filed as Exhibit 1 to Registration Statement on Form 8-A, filed with the Securities and Exchange Commission on February 22, 1990, and incorporated herein by reference.
- (c). Amendment No. 1 to the Rights Agreement, dated August 16, 1995— previously filed as Exhibit 4(b) to Form 8-K, dated August 16, 1995, and incorporated herein by reference.
- 10(a). Huntington Bancshares Incorporated Management Incentive Plan as amended and restated effective for plan years beginning on or after January 1, 2004.
- (b) Huntington Bancshares Incorporated 2004 Stock and Long-Term Incentive Plan.
- (c) Executive Deferred Compensation Plan of Huntington Bancshares Incorporated as amended and restated on February 18, 2004.
- 31.1 Sarbanes-Oxley Act 302 Certification Chief Executive Officer
- 31.2 Sarbanes-Oxley Act 302 Certification Chief Financial Officer
- 32.1 Sarbanes-Oxley Act 906 Certification Chief Executive Officer.
- 32.2 Sarbanes-Oxley Act 906 Certification Chief Financial Officer.
- (b) Reports on Form 8-K
 - 1. A report on Form 8-K, dated April 15 2004, was filed under report item numbers 7 and 12, concerning Huntington's results of operations for the first quarter ended March 31, 2004.
 - A report on Form 8-K, dated April 20, 2004, was filed under report item number 5, regarding Huntington's decision to terminate its Employee Stock Incentive Plan after it determined that the shares reserved for issuance of future option grants under the Plan would not be necessary following the approval of its 2004 Stock and Long-Term Incentive Plan.
 - 3. A report on Form 8-K, dated June 16, 2004, was filed under report items 5 and 7, announcing the delay in closing of the merger agreement between Huntington Bancshares Incorporated and Unizan Financial Corp.



SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Huntington Bancshares Incorporated (Registrant)

Date: August 9, 2004

Date: August 9, 2004

/s/ Thomas E. Hoaglin

Thomas E. Hoaglin Chairman, Chief Executive Officer and President

/s/ Donald R. Kimble

Donald R. Kimble Chief Financial Officer and Controller

HUNTINGTON BANCSHARES INCORPORATED

MANAGEMENT INCENTIVE PLAN As Amended and Restated Effective for Plan Years Beginning On or After January 1, 2004

ARTICLE I

PURPOSE; EFFECTIVE DATE

1.1 Purpose. The purpose of this Management Incentive Plan ("Plan") is to encourage, recognize, and reward exceptional levels of corporate, business unit, and individual performance. The Plan's intent is to use award dollars as a clear communication vehicle linking the interests of eligible officers with the interests of Huntington Bancshares Incorporated ("Corporation") by establishing a direct link between performance and incentive payments. The Plan serves to reinforce a management style which closely ties officer rewards to performance directly under his or her control and establishes the Corporation's willingness to reward individual performance that has a direct impact on incremental earnings. The purpose of this Plan is to provide incentive for key employees whose sustained performance directly influences the creation of shareholder value.

1.2 Effective Date. The Plan, as amended, will become effective as of January 1, 2004, if approved by a majority of the votes cast by the Corporation's shareholders at the annual meeting on April 27, 2004. No Awards will be paid under the Plan unless shareholder approval is obtained.

ARTICLE II DEFINITION OF TERMS

As used in the Plan, the following words shall have the meanings stated after them, unless otherwise specifically provided. In the Plan, words used in the singular shall include the plural, and words used in the plural shall include the singular. The gender of words used in this Plan shall include whatever may be appropriate under any particular circumstances.

(a) "Award" shall mean a cash incentive payment that may be due to a Participant under the Plan.

(b) "Base Salary" means the annual cash salary actually paid to a Participant for a particular Plan Year; provided however that Base Salary (a) excludes bonuses, incentive compensation, stock options, employer contributions to pension and benefit plans, and other forms of irregular payments, reimbursements and fringe benefits, and (b) includes any compensation that is deferred by the Participant pursuant to a nonqualified deferred compensation arrangement.

(c) "Board" or "Board of Directors" means the Board of Directors of the Corporation.

(d) "Change in Control" means, with respect to the Corporation, the occurrence of any of the following:

(1) Any "person" (as such term is used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), other than the Corporation or any person who as of the effective date is a director or officer of the Corporation or whose shares of common stock of the Corporation are treated as "beneficially owned" (as such term is used in Rule 13d-3 of the Exchange Act) by any such director or officer, becomes the beneficial owner,

directly or indirectly, of securities of the Corporation representing 25% or more of the combined voting power of the Corporation's then outstanding securities;

(2) Individuals who, as of the effective date any Award is granted hereunder, constitute the Board of Directors of the Corporation (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided, however, that any individual becoming a director subsequent to the date hereof whose election, or nomination for election, was approved by a vote of at least a majority of the directors comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding for this purpose any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board;

(3) A merger or consolidation of the Corporation, other than a merger or consolidation in which the voting securities of the Corporation immediately prior to the merger or consolidation continue to represent (either by remaining outstanding or being converted into securities of the surviving entity) 51% or more of the combined voting power of the Corporation or surviving entity immediately after the merger or consolidation with another entity;

(4) A sale, exchange, lease, mortgage, pledge, transfer, or other disposition (in a single transaction or a series of related transactions) of all or substantially all of

the assets of the Corporation which shall include, without limitation, the sale of assets or earning power aggregating more than 50% of the assets or earning power of the Corporation on a consolidated basis;

(5) a liquidation or dissolution of the Corporation;

(6) a reorganization, reverse stock split, or recapitalization of the Corporation which would result in any of the foregoing; or

(7) a transaction or series of related transactions having, directly or indirectly, the same effect as any of the foregoing.

(e) "Committee" means the Compensation Committee of the Board or such other committee appointed by the Board to administer the Plan. For purposes of granting, administering and certifying Awards to Covered Officers, the Committee or any sub-committee acting on behalf of the Committee shall be composed of two or more members of the Board each of whom is an "outside director" within the meaning of Section 162(m). Any Committee member who is not an "outside director" within the meaning of Section 162(m) shall abstain from participating in any decision to grant, administer or certify Awards to Covered Officers.

(f) "Corporation" means Huntington Bancshares Incorporated, a Maryland corporation, together with any and all Subsidiaries, and any successor thereto as provided in Section 8.2 herein.

(g) "Covered Officer" means any Participant who is designated as a Covered Officer by the Committee because it is anticipated that such Participant's compensation may exceed the limit under Section 162(m) and for whom any Award is intended to satisfy the Performance-Based Exception.



(h) "Extraordinary Events" means, with respect to the Corporation, any of the following (i) changes in tax law, generally accepted accounting principles or other such laws or provisions affecting reported financial results, (ii) accruals for reorganization and restructuring programs, (iii) special gains or losses in connection with the mergers and acquisitions or on the sales of branches or significant portions of the Corporation, (iv) any extraordinary non-recurring items as described in Accounting Principles Board Opinion No. 30 and/or in management's discussion and analysis of financial condition and results of operation appearing or incorporated by reference in the annual report on Form 10-K filed with the Securities and Exchange Commission, (v) gains on sales of auto loans, (vi) losses on the early repayment of debt, or (vii) any other events or occurrences of a similar nature as determined by the Committee.

(i) "Officer" means an officer of the Corporation or of a Subsidiary.

(j) "Participant" means an Officer selected by the Committee to participate in the Plan for a particular Plan Year.

(k) "Performance-Based Exception" means the performance-based exception from the tax deductibility limitations of Code Section 162(m).

(1) "Plan" means this Huntington Bancshares Incorporated Management Incentive Plan, as amended by the Corporation from time to time (formerly known as the Huntington Bancshares Incorporated Incentive Compensation Plan).

(m) "Plan Year" means the calendar year.

(n) "Qualifying Performance Criteria" means one or more of the following criteria upon which the achievement of specific, preestablished, objective performance goals for each Participant are based as determined by the Committee in connection with the grant and certification of Awards: (a) net income, (b) earnings per share, (c) return on equity, (d) return on

average equity, (e) return on assets, (f) return on average assets, (g) efficiency ratio determined as the ratio of total non-interest operating expenses (less amortization of intangibles) divided by total revenues (less net security gains), (h) non-interest income to total revenue ratio, (i) net interest margin, or (j) or other strategic milestones based on objective criteria established by the Committee, provided that, with respect to Covered Officers, such strategic milestones must be approved by the shareholders of the Corporation prior to the payment of any Award. The Qualifying Performance Criteria may be expressed by the Committee on the basis of performance criteria for the individual Participant, a specific business unit, Subsidiary or other affiliate or the Corporation as a whole or in any combination thereof. Qualifying Performance Criteria may be different for different Participants, as determined in the discretion of the Committee. Qualifying Performance Criteria with respect to Awards to Participants who are Covered Officers are limited to preestablished, objective performance goals as defined under Section 162(m). The Committee may include or exclude Extraordinary Events or any other objective events or occurrences in determining whether a performance goal based on the Qualifying Performance Criteria has been achieved; provided, however, that the Committee shall not have the discretion to increase the amount of an Award that would otherwise be due to a Participant who is a Covered Officer based on the preestablished performance goals for such Covered Officer with respect to the applicable Plan Year.

(o) "Section 162(m)" means Section 162(m) of the Internal Revenue Code of 1986, as amended, or any successor statute of similar import (the "Code") as interpreted by any regulations, governmental rulings or other official pronouncements promulgated under such statute.

(p) "Subsidiary" means a subsidiary of the Corporation of which at least 50% of the voting power is directly or indirectly owned or controlled by the Corporation.

ARTICLE III

ADMINISTRATION

3.1 Authority of the Committee. The Committee shall have full discretion to administer the Plan. The Committee is authorized to interpret and construe the Plan and to adopt such rules, regulations, and procedures for the administration of the Plan as the Committee deems necessary or advisable. The Committee may correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award in the manner and to the extent it shall deem desirable to carry the Plan into effect. Further, the Committee shall make all other determinations which may be necessary or advisable for the administration of the Plan. As permitted by law, the Committee may delegate its authority as identified herein, except that to the extent such delegation is not permitted under Code Section 162(m).

3.2 Decisions Binding. All determinations and decisions made by the Committee pursuant to the provisions of the Plan and all related orders and resolutions of the Board (provided, however, that only the Committee, or any subcommittee thereof, made up solely of 2 or more "outside directors" within the meaning of Code Section 162(m) shall participate in any decision, order or resolution to grant, administer, or certify Awards to Covered Officers) shall be final, conclusive, and binding on all persons, including the Corporation, its stockholders, Employees, Participants, and their estates and beneficiaries.

ARTICLE IV PLAN PARTICIPANTS

Participation in the Plan shall be limited to Officers who are specified by the Committee to be key employees whose performance may, in the opinion of the Committee, significantly contribute to the strategic performance and growth of the Corporation. The Committee shall select the Officers who will participate in the Plan for each Plan Year during the first ninety (90) days of the Plan Year (or such other date as may be permitted or required pursuant to Section 162(m)) and may select Officers who are hired or promoted during a Plan Year to participate for the remainder of the Plan Year. Selection to participate in this Plan in any Plan Year does not require the Committee to, or imply that the Committee will, select the same person to participate in the Plan in any subsequent Plan Year.

ARTICLE V

PERFORMANCE CRITERIA AND GOALS, MAXIMUM AWARD

5.1 Performance Criteria. Awards under the Plan may be based upon corporate, business unit, and individual performance; however, Awards under the Plan to Participants who are Covered Officers will be based solely upon the achievement of one or more performance goals based on one or more Qualifying Performance Criteria as selected by the Committee for a Plan Year. For all other Participants, Awards will be determined based upon performance goals established pursuant to the Qualifying Performance Criteria selected by the Committee and evaluations of the Participant's business unit and individual performance. Such evaluations will be made by the Participant's appropriate manager or senior officer. The Committee may select different Qualifying Performance Criteria for different incentive groups. 5.2 Performance Goals. The Committee shall establish annual written objective performance goals reflecting corporate performance. Performance goals based on the Qualifying Performance Criteria and the potential Award, expressed as a percentage of Base Salary, that will be payable upon attainment of such performance goals, will be established in writing not later than ninety (90) days after the commencement of the Plan Year to which the goals relate (or such earlier or later date as is permitted or required by Section 162(m)). Performance Goals for Participants who are not Covered Officers may be revised during the Plan Year based on Extraordinary Events or other factors. Potential Awards may vary among Participants in different incentive groups as determined by the Committee.

ARTICLE VI

PAYMENT OF AWARDS; MAXIMUM AWARD

6.1 Payment of Awards. Unless payment is deferred, Awards will be payable in cash as soon as practicable following the close of the Plan Year and calculation of the amount of the Awards; provided that Awards will be paid to Covered Officers only after the Committee has certified in writing in the minutes of a Committee meeting or otherwise that performance goals based on the Qualifying Performance Criteria applicable to Covered Officers and other material terms of the Plan have been satisfied. No Award will be paid to an Officer who is not employed by the Corporation or a Subsidiary on the day the Award is paid except in the case of death, disability, or retirement of the Officer or in the event that payment of the Award is deferred by the Committee or that a Change in Control of the Corporation has occurred. Awards are subject to federal, state and local income and other payroll tax withholding.

Notwithstanding the above, in the event a Participant dies, becomes disabled, or retires before payment of an Award, the Committee may, in its discretion, authorize payment to the Participant (or the Participant's estate or designated beneficiary) in such amount as the Committee deems appropriate.

6.2 Adjustments. The Committee may increase individual Awards based upon extraordinary circumstances; however, under no circumstance may the Committee increase a Covered Officer's Award above the amount determined based on the attainment of the specified performance goals identified in accordance with Section 5.2. In addition, notwithstanding the attainment of specified performance goals, the Committee has the discretion to reduce or eliminate an Award that would otherwise be paid to any Participant, including any Covered Officer, based on its evaluation of Extraordinary Events or any other factors, events or occurrences as determined by the Committee. However, notwithstanding Article IX or any provision of the Plan, an Award which is payable may not be reduced or eliminated following a Change in Control.

6.3 Maximum Award. Notwithstanding any other provision in the Plan, the maximum Award payable to a Participant for any Plan Year shall not exceed \$2,500,000.

ARTICLE VII

INTERIM AWARDS; CHANGE IN CONTROL

In the event of a Change in Control, the following provisions shall apply:

(a) The Committee shall make interim Awards based upon the Corporation's quarterly financial statements for the quarter ending immediately prior to or coinciding with the Change in Control.

(b) In determining the amount of interim Awards, the Committee shall follow the procedures for calculating Awards, except that the Committee shall annualize the actual level of year-to-date performance achieved with respect to each performance goal and such other performance objectives/assessments as the Committee shall determine. The amount of the Awards so calculated shall be payable on a pro-rated basis based upon the quarter ending immediately prior to or coinciding with the Change in Control in accordance with the following percentages:

First Quarter - 25% of the Award otherwise payable Second Quarter - 50% of the Award otherwise payable Third Quarter - 75% of the Award otherwise payable Fourth Quarter - 100% of the Award otherwise payable

(c) Notwithstanding the foregoing, each interim Award to be made under this Article 7 to any Participant who received an Award under this Plan for the Plan Year immediately preceding the year in which the Change in Control occurs shall not be less than the target award opportunity expressed as a percentage of Base Salary for the preceding Plan Year paid on a pro-rated basis as provided in subparagraph (b) above.

(d) The Committee shall grant an interim Award in accordance with this Article 7 to all Participants whether or not the Participants are employed by the Corporation when the Change in Control becomes effective unless the employment of such Participant was terminated for cause, as determined by the Corporation in its sole discretion.

ARTICLE VIII MISCELLANEOUS PROVISIONS

8.1 Guidelines. From time to time the Committee may adopt written guidelines for implementation and administration of the Plan.

8.2 Successors. All obligations of the Corporation under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Corporation, whether the existence of such successor is the result of a direct or indirect purchase of all or substantially all of the business and/or assets of the Corporation, or a merger, consolidation, or otherwise. 8.3 Unfunded Plans and Restrictions on Transfer. It is intended that the Plan be an "unfunded" plan for incentive compensation. The Committee may authorize the use of trusts or other arrangements to meet the obligations hereunder, provided, however, that the existence of such trusts or arrangements is consistent with the "unfunded" status of the Plan. Any benefits to which a Participant or his or her beneficiary may become entitled under this Plan shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, or charge, and any attempt to so transfer or encumber such benefits shall be void. This Plan does not give a Participant any interest, lien, or claim against any specific asset of the Corporation. Participants and beneficiaries shall have only the rights of a general unsecured creditor of the Corporation.

8.4 Status of Awards under Section 162(m). If any provision of the Plan or any agreement relating to an Award to a Covered Officer does not comply or is inconsistent with the requirements of Section 162(m), such provision or agreement shall be construed or deemed amended to the extent necessary to conform to such requirements. Notwithstanding the above, the Committee in its sole discretion may, with respect to any Award under the Plan, determine



that compliance with Section 162(m) is not desired after consideration of the goals of the Corporation's executive compensation philosophy and whether it is in the best interests of the Corporation to have such Award so qualified.

8.5 Deferrals of Awards. Unless otherwise provided by the Committee, a Participant may elect to defer payment of the Participant's Award under the Plan if deferral of an Award under the Plan is permitted pursuant to the terms of a deferred compensation program existing at the time the election to defer is permitted to be made, and the Participant complies with the terms of such program. Except in the situation of a Change in Control, the Committee may defer payment of an Award for such period as the Committee may determine.

8.6. Other Plans. Nothing in this Plan shall be construed as limiting the authority of the Committee, the Board of Directors, the Corporation or any Subsidiary to establish any other compensation plan, or as in any way limiting its or their authority to pay bonuses or supplemental compensation to any persons employed by the Company or a Subsidiary, whether or not such person is a Participant in this Plan and regardless of how the amount of such compensation or bonus is determined. However, no such plan will be established or operated in a way that entitles or allows a Covered Officer to receive an award under such plan as a substitution or supplement for not achieving goals under this Plan.

8.7 Expenses of Plan. The costs and expenses of administering the Plan will be borne by the Corporation.

8.8 No Employment Rights. No Participant has any right to be retained in the employ of the Corporation or any Subsidiary by virtue of participation in the Plan.

8.9 Governing Law. The Plan shall be governed by and construed according to the laws of the State of Ohio, without reference to its choice of law provisions.

ARTICLE IX AMENDMENT AND TERMINATION

The Corporation may at any time terminate, or from time to time, amend the Plan by action of the Board of Directors or by action of the Committee without shareholder approval unless such approval is required to satisfy the applicable provisions of Section 162(m).

HUNTINGTON BANCSHARES INCORPORATED 2004 STOCK AND LONG-TERM INCENTIVE PLAN

ARTICLE 1. ESTABLISHMENT, EFFECTIVE DATE, AND TERM

1.1 ESTABLISHMENT OF THE PLAN. Huntington Bancshares Incorporated, a Maryland corporation (hereinafter referred to as the "Corporation"), has established a longterm incentive compensation plan to be known as the "Huntington Bancshares Incorporated 2004 Stock and Long-Term Incentive Plan" (hereinafter referred to as the "Plan"), as set forth in this document. The Plan permits the grant of Nonqualified Stock Options, Incentive Stock Options, Restricted Stock, Restricted Stock Units, Stock Appreciation Rights, Deferred Stock, and Long-Term Performance Awards.

1.2 EFFECTIVE DATE. The Plan, if approved by the majority of the votes cast by the Corporation's stockholders at the annual meeting on April 27, 2004, shall become effective January 1, 2004 (the "Effective Date"), for Long-Term Performance Awards with cycles beginning on or after January 1, 2004, and for grants of Nonqualified Stock Options, Incentive Stock Options, Restricted Stock, Restricted Stock Units, Stock Appreciation Rights, and Deferred Stock beginning on or after April 27, 2004. The Plan shall remain in effect as provided in Article 1.4 herein. No Awards will be made under the Plan unless shareholder approval is obtained.

1.3 OBJECTIVES OF THE PLAN The objectives of the Plan are to help optimize the profitability and growth of the Corporation through incentives which are consistent with the Corporation's objectives and which link the interests of Participants to those of the Corporation's stockholders; to induce Participants to strive for the highest level of performance; and to promote teamwork among Participants.

The Plan is further intended to provide flexibility to the Corporation in its ability to motivate, attract, and retain the services of Participants who make significant contributions to the Corporation's success and the creation of shareholder value and to allow Participants to share in the success of the Corporation.

1.4 DURATION OF THE PLAN. The Plan shall commence on the Effective Date, as described in Article 1.2 herein, and shall remain in effect, subject to the right of the Board of Directors ("Board"), or a Committee delegated by the Board, to amend or terminate the Plan at any time pursuant to Article 18 herein. However, in no event may an Award be granted under the Plan on or after December 31, 2013.

ARTICLE 2. DEFINITIONS OF TERMS

As used in the Plan, the following words shall have the meanings stated after them, unless otherwise specifically provided. In the Plan, words used in the singular shall include the plural, and words used in the plural shall include the singular. The gender of words used in this Plan shall include whatever may be appropriate under any particular circumstances.

2.1 "AWARD" means, individually or collectively, a grant under this Plan of Nonqualified Stock Options, Incentive Stock Options, Restricted Stock, Restricted Stock Units, Stock Appreciation Rights, Deferred Stock Awards, or a Long-Term Performance Award.

2.2 "AWARD AGREEMENT" means a written statement prepared by the Corporation setting forth the terms and provisions applicable to Awards granted under this Plan.

2.3 "BENEFICIAL OWNER" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.

2.4 "BOARD" OR "BOARD OF DIRECTORS" means the Board of Directors of the Corporation.

2.5 "CHANGE IN CONTROL" means, with respect to the Corporation, the occurrence of any of the following:

(a) Any "person" (as such term is used in Sections 13(d) and 14(d) of the Exchange Act as in effect as of the date of this Agreement), other than the Corporation or any "person" who as of the Effective Date is a director or officer of the Corporation or whose shares of Common Stock of the Corporation are treated as "beneficially owned" (as such term is used in Rule 13d-3 of the Exchange Act as in effect as of the Effective Date) by any such director or officer, becomes the beneficial owner, directly or indirectly, of securities of the Corporation representing twenty-five percent (25%) or more of the combined voting power of the Corporation's then outstanding securities;

(b) Individuals who, as of the Effective Date, constitute the Board of Directors of the Corporation (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board, provided, however, that any individual becoming a director subsequent to the date hereof whose election, or nomination for election, was approved by a vote of at least a majority of the directors comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding for this purpose any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board;

(c) A merger or consolidation of the Corporation, other than a merger or consolidation in which the voting securities of the Corporation immediately prior to the merger or consolidation continue to represent (either by remaining outstanding or being converted into securities of the surviving entity) fifty-one percent (51%) or more of the combined voting power of the



Corporation or surviving entity immediately after the merger or consolidation with another entity;

(d) A sale, exchange, lease, mortgage, pledge, transfer, or other disposition (in a single transaction or a series of related transactions) of all or substantially all of the assets of the Corporation which shall include, without limitation, the sale of assets or earning power aggregating more than fifty percent (50%) of the assets or earning power of the Corporation on a consolidated basis;

(e) A liquidation or dissolution of the Corporation;

(f) A reorganization, reverse stock split, or recapitalization of the Corporation which would result in any of the foregoing; or

(g) A transaction or series of related transactions having, directly or indirectly, the same effect as any of the foregoing.

2.6 "CAUSE" means any of the following:

(a) The Participant shall have committed a felony or an intentional act of gross misconduct, moral turpitude, fraud, embezzlement, or theft in connection with the Participant's duties or in the course of the Participant's employment with the Corporation or any Subsidiary, and the Corporation shall have determined that such act is materially harmful to the Corporation;

(b) The Corporation or any Subsidiary shall have been ordered or directed by any federal or state regulatory agency with jurisdiction to terminate or suspend the Participant's employment and such order or directive has not been vacated or reversed upon appeal; or

(c) After being notified in writing by the Corporation to cease any particular competitive activity, the Participant shall have continued such competitive activity and the Corporation shall have determined that such act is materially harmful to the Corporation.

2.7 "CODE" means the Internal Revenue Code of 1986, as amended from time to time.

2.8 "COMMITTEE" means the Compensation Committee of the Board, as specified in Article 3 herein, or such other committee appointed by the Board to administer the Plan. For purposes of granting, administering and certifying Awards to Covered Employees, the Committee or any sub-committee acting on behalf of the Committee shall be composed of two (2) or more directors each of whom is an "outside director" within the meaning of Code Section 162(m). Any Committee member who is not an "outside director" within the meaning of Code Section 162(m) shall abstain from participating in any decision to grant, administer or certify Awards to Covered Employees.

2.9 "CORPORATION" means Huntington Bancshares Incorporated, a Maryland corporation, together with any and all Subsidiaries, and any successor thereto as provided in Article 22 herein.

2.10 "COVERED EMPLOYEE" means any Participant who is designated as a Covered Employee by the Committee because it is anticipated that his or her compensation may exceed the limit under Code Section 162(m) and for whom any Award is intended to satisfy the Performance-Based Exception.

2.11 "DEFERRAL PERIOD" means the period of time during which Deferred Stock is subject to deferral limitations under Article 10 herein.

2.12 "DEFERRED STOCK" means an Award granted to a Participant pursuant to Article 10 herein of the right to receive Shares at the end of a specified Deferral Period.

2.13 "DIRECTOR" means any individual who is a member of the Board of Directors of the Corporation.

2.14 "EFFECTIVE DATE" shall have the meaning ascribed to such term in Article 1.1 herein.

2.15 "EMPLOYEE" means any employee of the Corporation. Directors who are not employed by the Corporation shall not be considered Employees under this Plan.

2.16 "EXCHANGE ACT" means the Securities Exchange Act of 1934, as amended from time to time, or any successor act thereto.

2.17 "EXTRAORDINARY EVENTS" means, with respect to the Corporation, any of the following (i) changes in tax law, generally accepted accounting principles or other such laws or provisions affecting reported financial results, (ii) accruals for reorganization and restructuring programs, (iii) special gains or losses in connection with the mergers and acquisitions or on the sales of branches or significant portions of the Corporation, (iv) any extraordinary non-recurring items as described in Accounting Principles Board Opinion No. 30 and/or in management's discussion and analysis of financial condition and results of operation appearing or incorporated by reference in the annual report on Form 10-K filed with the Securities and Exchange Commission, (v) gains on sales of auto loans, (vi) losses on the early repayment of debt, or (vii) any other events or occurrences of a similar nature as determined by the Committee.

2.18 "FAIR MARKET VALUE" shall be, on any given date, (1) the mean between the highest and lowest selling prices at which the Shares were sold on the NASDAQ National Market or such other established securities market on which the Shares are traded, or, if there were no reported sales of Shares on such date, then, unless otherwise required pursuant to Treasury regulations under Code Section 422, the business day immediately preceding such date; or (2) the price that the Committee in good faith determines through any reasonable valuation method that a Share might change hands between a willing buyer and a willing seller, neither being under compulsion to buy or to sell and both having reasonable knowledge of the relevant facts. Notwithstanding the above, for purposes of broker-facilitated cashless exercises of Awards involving Shares under the Plan, "Fair Market Value" shall mean the real-time selling price of such Shares as reported by the broker facilitating such exercises.

2.19 "INCENTIVE STOCK OPTION" OR "ISO" means an option to purchase Shares granted under Article 6 herein and which is designated as an Incentive Stock Option and which is intended to meet the requirements of Code Section 422.

2.20 "IMMEDIATE FAMILY" means, with respect to a particular Participant, such Participant's child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, and shall include adoptive relationships.

2.21 "INSIDER" shall mean any person subject to the reporting requirements of Section 16 of the Exchange Act.

2.22 "LONG-TERM PERFORMANCE AWARD" means an Award to a Participant pursuant to Article 11 herein.

2.23 "NONEMPLOYEE DIRECTOR" means an individual who is a member of the Board of Directors of the Corporation but who is not an Employee of the Corporation.

2.24 "NONQUALIFIED STOCK OPTION" OR "NQSO" means an option to purchase Shares granted under Article 6 herein and which is not intended to meet the requirements of Code Section 422.

2.25 "OPTION" means an Incentive Stock Option, or a Nonqualified Stock Option granted to a Participant pursuant to Article 6 herein.

2.26 "OPTION PRICE" means the price at which a Share may be purchased by a Participant pursuant to an Option.

2.27 "PARTICIPANT" means an Employee or, solely with respect to a Nonqualified Stock Option Award, Restricted Stock Award, Restricted Stock Unit Award, or a Deferred Stock Award, a Nonemployee Director who has an outstanding Award granted under the Plan.

2.28 "PERFORMANCE-BASED EXCEPTION" means the performance-based exception from the tax deductibility limitations of Code Section 162(m).

2.29 "PERFORMANCE CYCLE" shall mean the two, three, or four calendar year period designated by the Committee during which the performance objectives or goals must be met.

2.30 "PERMISSIBLE TRANSFEREE" means any member of the Immediate Family of the Participant, any trust solely for the benefit of the Participant or members of the Participant's Immediate Family, or any partnership whose only partners are members of the Participant's Immediate Family.

2.31 "PERIOD OF RESTRICTION" means the period during which the transfer of Shares of Restricted Stock or Restricted Stock Units is limited in some way (based on the passage of

time, which, unless waived by the Committee, shall not be less than six (6) months from the date of grant, the achievement of performance objectives, or upon the occurrence of other events as determined by the Committee, in its discretion), and the Shares or RSUs are subject to a substantial risk of forfeiture, as provided in Article 7 and Article 8 herein.

2.32 "PERSON" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" as described in Section 13(d) thereof.

2.33 "QUALIFYING PERFORMANCE CRITERIA" means any one or more of the following objective performance criteria (either individually, alternatively, or in any combination, applied to either the Corporation as a whole or to a business unit or subsidiary, individually, alternatively, or in any combination and measured over a period of years, on an absolute basis, or relative to a pre-established target, to previous years' results, or to a designated comparison group, in each case as specified by the Committee) upon which the achievement of specific, pre-established, objective performance goals for each Participant are based as determined by the Committee in connection with the grant and certification of Awards: (a) net income, (b) earnings per share, (c) return on equity, (d) return on average equity, (e) return on assets, (f) return on average assets, (g) "efficiency ratio" determined as the ratio of total non-interest operating expenses (less amortization of intangibles) divided by total revenues (less net security gains), (h) non-interest income to total revenue ratio, (i) net interest margin, or (j) other strategic milestones based on objective criteria established by the Committee, provided that, with respect to Covered Employees, such strategic milestones must be approved by the shareholders of the Corporation prior to the payment of any Award. Qualifying Performance Criteria may be different for different Participants, as determined in the discretion of the Qualifying Performance Criteria has been achieved; provided, however, that the Committee shall not have the discretion to increase the amount of an Award that would otherwise be due to a Participant who is a Covered Employee based on such Covered Employee's pre-established performance Goals for the applicable Performance Criteria has been achieved; provided, however, that the Committee shall not have the discretion to increase the amount of an Award that would otherwise be due to a Participant who is a Covered Employee based on such Covered Employee's pr

2.34 "RESTRICTED STOCK" means an Award granted to a Participant pursuant to Article 7 herein.

2.35 "RESTRICTED STOCK UNIT" OR "RSU" means an Award granted to a Participant pursuant to Article 8 herein and which is settled (i) by the delivery of one (1) Share for each RSU, (ii) in cash in an amount equal to the Fair Market Value of one (1) Share for each RSU, or (iii) in a combination of cash and Shares, as determined by the Committee. The Award of an RSU represents the promise of the Corporation to deliver Shares, cash, or a combination thereof, as applicable, at the end of the Period of Restriction (or such later date as determined by the Committee) in accordance with and subject to the terms and conditions of the applicable Award Agreement, and is not intended to constitute a transfer of property within the meaning of Code Section 83(b).

2.36 "RETIREMENT" shall mean, in the case of an Employee, the retirement from the employ of the Corporation under one or more of the retirement plans of the Corporation, or as otherwise specified by the Committee and, in the case of a Nonemployee Director, shall mean the retirement from the Board at any time after the Nonemployee Director attains age fifty-five (55) and has served at least five (5) years as a Director.

2.37 "SHARES" means the shares of common stock of the Corporation.

2.38 "STOCK APPRECIATION RIGHT" OR "SAR" means an Award, granted alone or in connection with a related Option, designated as a SAR, pursuant to Article 9 herein.

2.39 "SUBSIDIARY or "SUBSIDIARIES" means any corporation or other entity whose financial statements are consolidated with the Corporation.

ARTICLE 3. ADMINISTRATION

3.1 AUTHORITY OF THE COMMITTEE. The Plan shall be administered by the Committee, except as limited by law or by the Charter or Bylaws of the Corporation, and subject to the provisions herein, the Committee shall have full power to select the Participants who shall participate in the Plan; determine the sizes and types of Awards; determine the terms and conditions of Awards in a manner consistent with the Plan; construe and interpret the Plan and any agreement or instrument entered into under the Plan as they apply to Participants; establish, amend, or waive rules and regulations for the Plan's administration as they apply to Participants; and (subject to the provisions of Article 18 herein) amend the terms and conditions of any outstanding Award to the extent such terms and conditions are within the discretion of the Committee as provided in the Plan. The Committee may correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award in the manner and to the extent it shall deem desirable to carry the Plan into effect. Further, the Committee shall make all other determinations which may be necessary or advisable for the administration of the Plan. As permitted by law, the Committee may delegate its authority as identified herein, except that to the extent such delegation is not permitted under Code Section 162(m) and the regulations thereunder.

3.2 DECISIONS BINDING. All determinations and decisions made by the Committee pursuant to the provisions of the Plan and all related orders and resolutions of the Board (provided, however, that only the Committee, or any subcommittee thereof, made up solely of 2 or more "outside directors" within the meaning of Code Section 162(m) shall participate in any decision, order or resolution to grant, administer, or certify Awards to Covered Employees) shall be final, conclusive, and binding on all persons, including the Corporation, its stockholders, Employees, Participants, and their estates and beneficiaries.

ARTICLE 4. SHARES SUBJECT TO THE PLAN AND MAXIMUM AWARDS

4.1 NUMBER OF SHARES AVAILABLE FOR GRANTS AND MAXIMUM AWARDS. Subject to adjustment as provided in this Article 4 herein, the maximum aggregate number of Shares hereby reserved for issuance to Participants under the Plan shall be twelve million (12,000,000) Shares. The Shares issued under the Plan may be authorized and unissued Shares, Shares purchased on the open market, or Shares held as treasury stock.

The following rules shall apply to grants of Awards under the Plan:

(a) The maximum aggregate number of Shares which may be subject to (1) option by one or more Option Awards pursuant to Article 6, (2) one or more SAR Awards (whether settled in cash, Shares) pursuant to Article 9, or (3) any combination of Option Awards or SAR Awards to a Participant shall be four million (4,000,000) Shares over any five (5) year period.

(b) The maximum aggregate cash Award or cash equivalent value of an Award of Shares that may be paid with respect to any specified Performance Cycle to a Participant pursuant to any Long-Term Performance Award pursuant to Article 11 shall be four million dollars (\$4,000,000).

(c) The maximum aggregate cash equivalent value of (1) Awards of Restricted Stock pursuant to Article 7, (2) Awards of RSUs pursuant to Article 8 (whether settled in cash, Shares, or a combination thereof), (3) Awards of Deferred Stock pursuant Article 10, or (4) any combination thereof that may be awarded to a Participant for any calendar year shall be four million dollars (\$4,000,000).

(d) The maximum aggregate number of (1) Shares of Restricted Stock awarded pursuant to Article 7, (2) Shares subject to an Award of RSUs pursuant to Article 8, (3) Shares of Deferred Stock awarded pursuant to Article 10, and (4) Long-Term Performance Award Shares awarded pursuant to Article 11 shall not exceed two million four hundred thousand (2,400,000) Shares authorized for issuance pursuant to this Article 4.1, subject to adjustment under Article 4.3, over the term of the Plan; provided, however, that the limitation of this subparagraph (d) shall not apply to any Restricted Stock Awards, RSU Awards, Deferred Stock Awards, or Long-Term Performance Awards to the extent earned on the basis of specific performance goals established by the Committee.

4.2 LAPSED AWARDS. If any Award granted under this Plan terminates, expires, or lapses for any reason, any Shares subject to such Award shall again be available for a grant of an Award under the Plan.

4.3 ADJUSTMENTS IN AUTHORIZED SHARES. In the event of any change in the number of outstanding Shares through the declaration and payment of a stock dividend or stock split, spin off, merger, or other reorganization, or through any recapitalization resulting in the combination or exchange of Shares in which the Corporation does not receive any consideration, a corresponding adjustment shall be made in the number of Shares which may be delivered under Article 4.1 and in the number and/or price of Shares subject to outstanding Awards granted under the Plan; provided, however, that the number of Shares subject to any Award shall always be a



whole number (by rounding down); provided, further, that the Committee shall, in its sole discretion, make any further adjustments as are necessary to prevent dilution or enlargement of rights.

Further, unless otherwise required by applicable law or regulation, Shares granted through the assumption of or in substitution for outstanding awards granted by a company that is merged or consolidated with, or acquired by, the Company shall not be subject to the Share limitations of Article 4.1.

ARTICLE 5. ELIGIBILITY AND PARTICIPATION

5.1 ELIGIBILITY. Persons eligible to participate in this Plan include any Employee and Nonemployee Director of the Corporation, including any Employee who is a member of the Board.

5.2 ACTUAL PARTICIPATION. Subject to the provisions of the Plan, the Committee may, from time to time, select from all eligible Employees and Nonemployee Directors, those to whom Awards shall be granted and shall determine the nature and amount of each Award.

ARTICLE 6. STOCK OPTIONS

6.1 GRANT OF OPTIONS. Subject to the terms and provisions of the Plan, Options may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee.

No Option shall be granted to any Employee or Nonemployee Director if, upon the granting of such Option, the number of Shares then subject to all Options to purchase held by the Employee or Nonemployee Director, as the case may be, plus the Shares then owned by such Employee or Nonemployee Director, would constitute more than ten (10%) of the total combined voting power of all classes of stock of the Corporation. For the purpose of the preceding sentence, an Employee or a Nonemployee Director shall be deemed to own all Shares which are attributable to him or her under Section 424(d) of the Code, including, without limiting the generality of the foregoing, shares owned by his or her brothers, sisters, spouse, ancestors, and lineal descendants.

The Committee may not grant ISOs under the Plan to any Employee which would permit the aggregate Fair Market Value (determined on the date of grant) of Shares with respect to which ISOs (under this and any other Plan of the Corporation) are exercisable for the first time by such Employee during any calendar year to exceed one hundred thousand dollars (\$100,000). Any excess shall be deemed a NQSO. No ISO shall be granted to a Nonemployee Director.

If Shares acquired upon exercise of an Incentive Stock Option are disposed of by a Participant prior to the expiration of either two (2) years from the date of grant of such Incentive Stock Option or one year from the transfer of Shares to such Participant pursuant to the exercise of such Incentive Stock Option, or in any other disqualifying disposition within the meaning of

Section 422 of the Code, such Participant shall notify the Corporation in writing of the date and terms of such disposition and shall cooperate with the Corporation with respect to any tax withholding required or resulting from such disqualifying dispositions. A disqualifying disposition by a Participant shall not affect the status of any other Incentive Stock Option granted under the Plan as an Incentive Stock Option.

6.2 AWARD AGREEMENT. Each Option grant shall be evidenced by an Award Agreement that shall specify the Option Price, the duration of the Option, the number of Shares to which the Option pertains, the date of grant, vesting restrictions, if any, and such other provisions as the Committee shall determine. The Award Agreement also shall specify whether the Option is intended to be an ISO or an NQSO.

6.3 OPTION PRICE. The Option Price for each grant of an Option under this Plan shall be determined by the Committee but shall be at least equal to one hundred percent (100%) of the Fair Market Value of a Share on the date the Option is granted; provided, however, that for Options granted through the assumption of or in substitution for outstanding awards granted by a company that is merged or consolidated with, or acquired by, the Company, the Option Price shall be determined by the Committee in its sole discretion and, if applicable, consistent with Code Section 424(a).

6.4 DURATION OF OPTIONS. Each Option granted to an Employee or Nonemployee Director shall expire at such time as the Committee shall determine at the time of grant; provided, however, that no Option shall be exercisable on or later than the tenth (10th) anniversary date of its grant.

6.5 EXERCISE OF OPTIONS. Except as otherwise provided in this Plan, Options granted under this Article 6 shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance determine, which need not be the same for each grant or for each Participant. Options granted under this Article 6 shall be exercised by the delivery to the Corporation of written or other notice acceptable to the Corporation setting forth the number of Shares with respect to which the Option is to be exercised.

The Option Price upon exercise of any Option shall be payable to the Corporation in full either: (a) in cash or its equivalent; (b) by tendering previously acquired Shares, including by attestation, having an aggregate Fair Market Value equal to the total Option Price (provided that the Shares which are tendered must have been held by the Participant for at least six (6) months prior to their tender); (c) by a combination of (a) and (b); (d) subject to applicable securities laws and restrictions, through a broker-facilitated cashless exercise procedure acceptable to the Committee, or (e) by any other means which the Committee determines to be consistent with the Plan's purpose and applicable law.

6.6 RESTRICTIONS ON SHARE TRANSFERABILITY. In addition to the foregoing, the Committee may impose such restrictions on any Shares acquired pursuant to the exercise of an Option granted under this Article 6 as it may deem advisable, including, without limitation, restrictions under applicable Federal securities laws, under the requirements of any stock

exchange or market upon which such Shares are then listed and/or traded, and under any blue sky or state securities laws applicable to such Shares.

6.7 EXERCISE UPON TERMINATION OF EMPLOYMENT. Except as otherwise provided in this Plan or as otherwise provided in the Award Agreement or by the Committee, in the event that the employment of a Participant is terminated for any reason other than death or Retirement, the rights under each then outstanding Option granted to the Participant pursuant to the Plan shall terminate upon the earlier of (1) the expiration of such Option, or (2) sixty (60) days after the Participant's termination of employment, unless such termination of employment was for Cause. If a Participant's employment is terminated for Cause, the rights under each then outstanding Option granted to the Participant pursuant to the Plan shall immediately terminate.

In the event that the employment of a Participant is terminated by reason of Retirement, each then outstanding Option of such Participant shall continue to be exercisable at such times and be subject to such restrictions and conditions, including expiration, as set forth in the applicable Award Agreement. Notwithstanding any other provision in the Plan to the contrary, in the event of the Retirement of a Participant, each then outstanding ISO not exercised within three (3) months of termination of employment shall automatically convert to an NQSO.

In the event that the employment of a Participant is terminated by reason of death or a Participant dies after Retirement, all such Participant's then outstanding Options shall become exercisable in full, and the executor or administrator of such Participant's estate or a person or persons who have acquired the Options directly from such Participant by bequest, inheritance, or by reason of written designation as a beneficiary on a form proscribed by the Corporation, shall have until the expiration dates of such Options or thirteen (13) months after the Participant's date of death, whichever first occurs, to exercise such Options.

In addition to the foregoing, the Committee may include such provisions in the Award Agreement entered into with each Participant as it deems advisable (which may be more restrictive than described above), which provisions need not be uniform among all Options issued pursuant to this Article 6, and which may reflect distinctions based on the reasons for termination of employment.

6.8 EXERCISE UPON TERMINATION OF DIRECTORSHIP. Except as otherwise provided in this Plan, if a Participant's status as a Nonemployee Director ceases for any reason other than Retirement or death, any NQSO granted to such Participant under the Plan shall terminate thirteen (13) months after the termination of such Participant's status as a Nonemployee Director; provided, however, that no Option shall be exercisable after its expiration date.

If a Participant's status as a Nonemployee Director ceases by reason of Retirement, then all such Participant's Options shall become exercisable in full, and such Participant may exercise such Options until their expiration date.

If a Participant's status as a Nonemployee Director ceases by reason of death, or a Participant who was a Nonemployee Director dies after Retirement, all such Participant's then outstanding Options shall become exercisable in full, and the executor or administrator of such Participant's estate or a person or persons who have acquired the Options directly from such Participant by bequest, inheritance, or by reason of written designation as a beneficiary on a form proscribed by the Corporation, shall have until the expiration dates of such Options or thirteen (13) months after the Participant's date of death, whichever first occurs, to exercise such Options.

6.9 NONTRANSFERABILITY OF OPTIONS.

(a) INCENTIVE STOCK OPTIONS. No ISO granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than (1) by will or by the laws of descent and distribution, or (2) if permitted under Code Section 422, by transfer to a trust, the beneficial owner of which, pursuant to Code Section 671 and applicable state law, is the Participant to whom the ISO was granted ("Grantor Beneficial Trust"). Further, all ISOs granted to a Participant under the Plan shall be exercisable during his or her lifetime only by such Participant or the trustee of the Grantor Beneficial Trust if permitted under Code Section 422.

(b) NONQUALIFIED STOCK OPTIONS. No NQSO granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated by a Participant, other than by will or by the laws of descent and distribution, except that any NQSO (i) may be transferred by a Participant without consideration to Permissible Transferees, but such transferees may not transfer such NQSO's to third parties except by will or the laws of descent and distribution and then only to a Permissible Transferee, and (ii) shall be subject to all other conditions and restrictions applicable to Options granted under the Plan prior to such transfer. Any transfer to a Permissible Transferee shall consist of Options covering a minimum of five thousand (5,000) Option Shares.

ARTICLE 7. RESTRICTED STOCK

7.1 GRANT OF RESTRICTED STOCK. Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant Shares of Restricted Stock to Participants in such amounts as the Committee shall determine.

7.2 RESTRICTED STOCK AGREEMENT. Each Restricted Stock grant shall be evidenced by a Restricted Stock Award Agreement that shall specify the Period(s) of Restriction, the number of Shares of Restricted Stock granted, and such other provisions as the Committee shall determine.

7.3 OTHER RESTRICTIONS. The Committee shall impose such other conditions and/or restrictions on any Shares of Restricted Stock granted pursuant to the Plan as it may deem advisable including, without limitation, a requirement that Participants pay a stipulated purchase price for each Share of Restricted Stock, restrictions based upon the achievement of specific
performance objectives (Corporation-wide, business unit, and/or individual), Qualifying Performance Criteria, a Performance Cycle, time-based restrictions, and/or restrictions under applicable Federal or state securities laws.

The Corporation shall retain the certificates representing Shares of Restricted Stock in the Corporation's possession until such time as all conditions and/or restrictions applicable to such Shares have been satisfied.

Except as otherwise provided in this Article 7, Shares of Restricted Stock covered by each Restricted Stock grant made under the Plan shall become freely transferable by the Participant after the last day of the applicable Period of Restriction.

7.4 VOTING RIGHTS. During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may, at the discretion of the Committee, exercise full voting rights with respect to those Shares.

7.5 DIVIDENDS AND OTHER DISTRIBUTIONS. During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may, at the discretion of the Committee, be credited with regular cash dividends paid with respect to the underlying Shares while they are so held. Such dividends may be paid currently, accrued as contingent cash obligations, or converted into additional Shares of Restricted Stock, upon such terms as the Committee establishes.

The Committee may apply any restrictions to the dividends that the Committee deems appropriate. Without limiting the generality of the preceding sentence, if the grant or vesting of Shares of Restricted Stock granted to a Covered Employee is designed to comply with the requirements of the Performance-Based Exception, the Committee may apply any restrictions it deems appropriate to the payment of dividends declared with respect to such Shares of Restricted Stock, such that the dividends and/or the Shares of Restricted Stock maintain eligibility for the Performance-Based Exception. Shares of Restricted Stock shall be subject to adjustment as provided in Article 4.3.

7.6. NONTRANSFERABILITY. During any Period(s) of Restriction, the Participant shall have no right to transfer any rights with respect to its Award of Shares of Restricted Stock.

ARTICLE 8. RESTRICTED STOCK UNITS

8.1 GRANT OF RSUs. Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant RSUs to Participants in such amounts as the Committee shall determine.

8.2 AWARD AGREEMENT. Each RSU shall be evidenced by an Award Agreement that shall specify the Period(s) of Restriction, the number of RSUs granted, and such other provisions as the Committee shall determine.



8.3 OTHER RESTRICTIONS. The Committee shall impose such other conditions and/or restrictions on any RSUs granted pursuant to the Plan as it may deem advisable including, without limitation, a requirement that Participants pay a stipulated purchase price for each RSU, restrictions based upon the achievement of specific performance objectives (Corporation-wide, business unit, and/or individual), Qualifying Performance Criteria, a Performance Cycle, time-based restrictions, and/or restrictions under applicable Federal or state securities laws.

8.4 VOTING RIGHTS. During the Period of Restriction, unless otherwise determined by the Committee in its discretion, Participants holding RSUs may not exercise any voting rights with respect to such RSUs.

8.5 DIVIDENDS AND OTHER DISTRIBUTIONS. During the Period of Restriction, unless otherwise determined by the Committee in its discretion, Participants holding RSUs shall not be entitled to any dividends or dividend equivalents with respect to such RSUs.

8.6. NONTRANSFERABILITY. During any Period(s) of Restriction, the Participant shall have no right to transfer any rights with respect to its Award of RSUs.

ARTICLE 9. STOCK APPRECIATION RIGHTS

9.1 GRANT OF SARs. Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant SARs to Participants in such amounts as the Committee shall determine. A SAR shall represent a right to receive a payment in cash, Shares, or a combination thereof, equal to the excess of the Fair Market Value of a specified number of Shares on the date the SAR is exercised over an amount (the "SAR exercise price") which shall be no less than the Fair Market Value on the date the SAR was granted (or the Option Price for SARs granted in tandem with an Option), as set forth in the applicable Award Agreement.

9.2 AWARD AGREEMENT. Each SAR grant shall be evidenced by an Award Agreement that shall specify the SAR exercise price, the duration of the SAR, the number of Shares to which the SAR pertains, whether the SAR is granted in tandem with the grant of an Option or is freestanding, and such other provisions as the Committee shall determine. SARs granted under this Article 9 shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance approve and which shall be set forth in the applicable Award Agreement, which need not be the same for each grant or for each Participant.

9.3 EXERCISE. SARs shall be exercised by the delivery to the Corporation of written or other notice of exercise acceptable to the Corporation, setting forth the number of Shares with respect to which the SAR is to be exercised. The date of exercise of the SAR shall be the date on which the Corporation shall have received notice from the Participant of the exercise of such SAR. SARs granted in tandem with the grant of an Option may be exercised for all or part of the Shares subject to the related Option upon the surrender of the right to exercise the equivalent portion of the related Option. SARs granted in tandem with the grant of an Option may be exercised only with respect to the shares for which its related Option is then exercisable.

With respect to SARs granted in tandem with an ISO, (a) such SAR will expire no later than the expiration of the underlying ISO, (b) the value of the payout with respect to such SAR may be for no more than 100% of the difference between the Option Price of the underlying ISO and the Fair Market Value of the Shares subject to the underlying ISO at the time such SAR is exercised, and (c) such SAR may be exercised only when the Fair Market Value of the Shares subject to the underlying ISO exceeds the Option Price of the ISO.

SARs granted independently from the grant of an Option may be exercised upon the terms and conditions contained in the applicable Award Agreement. In the event the SAR shall be payable in Shares, a certificate for the Shares acquired upon exercise of an SAR shall be issued in the name of the Participant as soon as practicable following receipt of notice of exercise. No fractional Shares will be issuable upon exercise of the SAR and, unless provided in the applicable Award Agreement or otherwise determined by the Committee, the Participant will receive cash in lieu of fractional Shares.

9.4 EXERCISE UPON TERMINATION OF EMPLOYMENT. Each Participant's Award Agreement shall set forth the extent to which the Participant shall have the right to exercise a SAR following termination of the Participant's employment with the Corporation. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the Award Agreement entered into the Participants, need not be uniform among all SARs issued pursuant to this Article 9, and may reflect distinctions based on the reasons for termination of employment.

9.5 NONTRANSFERABILITY. Unless otherwise determined by the Committee in its discretion, no SAR granted under this Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, SARs granted in tandem with an ISO granted to a Participant under the Plan shall be exercisable during the Participant's lifetime only by such Participant.

ARTICLE 10. DEFERRED STOCK AWARDS

10.1 GRANT OF DEFERRED STOCK. Subject to the terms and provisions of the Plan, the Committee may authorize the grant or sale of Deferred Stock to Participants in such amounts the Committee shall determine. Each such grant or sale shall constitute the agreement by the Corporation to deliver Shares to the Participant in the future in consideration of the performance of services, but subject to the fulfillment of such conditions during the Deferral Period as the Committee may specify. Each such grant or sale may be made without additional consideration of in consideration of a payment by such Participant that is less than the Fair Market Value of the Shares at the date of grant.

10.2 AWARD AGREEMENT. Each grant or sale of Deferred Stock shall be evidenced by an Award Agreement, which shall contain such terms and provisions, consistent with this Plan, as the Committee may approve.

10.3 DEFERRAL PERIOD. Each such grant or sale shall be subject, except (if the Committee shall so determine) in the event of a Change in Control or other similar transaction or event, to a Deferral Period of not less than one (1) year, as determined by the Committee at the date of grant.

10.4 VOTING RIGHTS. During the Deferral Period, unless otherwise determined by the Committee in its discretion, the Participant shall have no rights of ownership in the Shares of Deferred Stock and shall have no right to vote them.

10.5 DIVIDENDS. During the Deferral Period, the Committee may, at or after the date of grant, authorize payment of dividend equivalents on any Shares of Deferred Stock on either a current, deferred, or contingent basis, either in cash or in additional Shares.

10.6 NONTRANSFERABILITY. During the Deferral Period, the Participant shall have no right to transfer any rights with respect to its Award of Shares of Deferred Stock.

ARTICLE 11. LONG-TERM PERFORMANCE AWARDS

11.1 LONG-TERM PERFORMANCE AWARDS. Subject to the terms and provisions of the Plan, a Participant shall have the opportunity to receive an Award of cash, Shares, or a combination thereof, in such amounts and upon such terms and at such times as determined by the Committee in its sole discretion.

11.2 TERMS OF LONG-TERM PERFORMANCE AWARDS. The Committee shall set performance objectives in its discretion which, depending on the extent to which they are met, will determine the number of Shares and/or value of Long-Term Performance Awards that will be paid to the Participant. The Committee shall establish the Performance Cycle for each Long-Term Performance Award and shall impose such other conditions and/or restrictions on any Long-Term Performance Awards as it may deem advisable including, without limitation, restrictions based upon the achievement of specific performance objectives (Corporation-wide, business unit, and/or individual), Qualifying Performance Criteria, time-based restrictions, and/or restrictions under applicable Federal or state securities laws.

11.3 EARNING OF LONG-TERM PERFORMANCE AWARDS. Subject to the terms of this Plan and Article 11, after the applicable Performance Cycle has ended, the Participant shall be entitled to receive a payment of the number of Shares and/or cash earned by the Participant over the applicable Performance Cycle. Notwithstanding the attainment of specific performance objectives, the Committee has the discretion to reduce or eliminate an Award that would otherwise be payable based on its evaluation of Extraordinary Events and other factors.

11.4 FORM AND TIMING OF PAYMENT OF LONG-TERM PERFORMANCE AWARDS. Payment of Long-Term Performance Awards shall be made as soon as practical following the close of the applicable Performance Cycle in a manner designated by the Committee, in its sole discretion. Subject to the terms of this Plan, the Committee, in its sole discretion, may pay Long-Term Performance Awards in the form of cash or in Shares (or in a

combination thereof) which have an aggregate Fair Market Value equal to the value of the Long-Term Performance Awards at the close of the applicable Performance Cycle. Such Shares may be granted subject to any restrictions deemed appropriate by the Committee.

11.5 REQUIREMENT OF EMPLOYMENT. Except as otherwise provided in this Plan and as specified in Article 16, a Participant must remain in the employment of the Corporation until the payment of a Long-Term Performance Award in order to be entitled to payment; provided, however, that the Committee may, in its sole discretion, provide for a partial or full payment in the event the Participant is not so employed.

11.6 NONTRANSFERABILITY. A Long-Term Performance Award may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution.

ARTICLE 12. CODE SECTION 162(m) DEDUCTION QUALIFICATIONS

12.1 AWARDS FOR COVERED EMPLOYEES. At all times when Code Section 162(m) is applicable, all Awards granted to a Covered Employee under this Plan shall comply with the Performance-Based Exception requirements of Code Section 162(m). In addition, in the event that changes are made to Code Section 162(m) to permit greater flexibility with respect to any Award available under the Plan, the Committee may, subject to this Article 12, make any adjustments it deems appropriate. Notwithstanding the above, the Committee may, in its sole discretion, with respect to any Award under the Plan, determine that compliance with Code Section 162(m) is not desired after consideration of the goals of the Corporation's executive compensation philosophy and whether it is in the best interests of the Corporation to have such Award so qualified.

12.2 DESIGNATION OF COVERED EMPLOYEES. For each Performance Cycle, the Committee will designate which Participants are Covered Employees within ninety (90) days of the beginning of the Performance Cycle (or such earlier or later date as is permitted or required by Code Section 162(m)).

12.3 ESTABLISHMENT OF QUALIFYING PERFORMANCE CRITERIA AND AWARDS FOR COVERED EMPLOYEES. Within ninety (90) days of the beginning of a Performance Cycle (or such earlier or later date as is permitted or required by Code Section 162(m)), the Committee shall, in its sole discretion, for each such Performance Cycle, determine and establish in writing one or more performance goals based on one or more Qualifying Performance Criteria applicable to the Performance Cycle for each Covered Employee. The Committee may establish any number of differing Performance Cycles, performance goals, Qualifying Performance Criteria, and Awards for Covered Employees running concurrently, in whole or in part.

12.4 CERTIFICATION OF ACHIEVEMENT OF QUALIFYING PERFORMANCE CRITERIA AND AMOUNT OF AWARDS. After the end of each Performance Cycle, or such earlier date if the Qualifying Performance Criteria are achieved (and such date otherwise

complies with Code Section 162(m)), the Committee shall certify in writing, prior to the payment of any Award to a Covered Employee, that the performance goal based on the Qualifying Performance Criteria for the Performance Cycle and all other material terms of the Plan were satisfied. The Committee may not, under any circumstances, increase an Award to a Covered Employee above the amount payable pursuant to the pre-established performance goal based on the Qualifying Performance Criteria for the Performance Criteria for the Performance Criteria for the pre-established performance goal based on the Qualifying Performance Criteria for the Performa

12.5 MAXIMUM AWARD TO PARTICIPANTS. Notwithstanding any other provision of this Plan to the contrary, the maximum aggregate number of Shares which may be subject to option by one or more Option Awards or SAR Awards (whether settled in cash, Shares, or a combination thereof) to a Participant shall be four million (4,000,000) Shares over any five (5) year period. Notwithstanding any of provision of this Plan to the contrary, the maximum amount of compensation (whether represented by Shares, cash, or a combination thereof) that may be payable to a Participant with respect to any specified Performance Cycle, pursuant to the attainment of a performance goal associated with a Long-Term Performance Award, Restricted Stock Award, or Deferred Stock Award shall be four million dollars (\$4,000,000) for each type of Award.

12.6 TAX AND SECURITY LAWS. In the event that applicable tax and/or securities laws change to permit the Committee discretion to alter the governing performance measures without obtaining shareholder approval of such changes, the Committee shall have the sole discretion to make such changes without obtaining shareholder approval.

ARTICLE 13. BENEFICIARY DESIGNATION

Each Participant under the Plan may, from time to time, name any beneficiary or beneficiaries (who may be named contingently or successively) to whom any benefit under the Plan is to be paid in case of his or her death before he or she receives any or all of such benefit. Each such designation shall revoke all prior designations by the same Participant, shall be in a form prescribed by the Corporation, and will be effective only when filed by the Participant in writing with the Corporation during the Participant's lifetime. In the absence of any such designation, benefits remaining unpaid at the Participant's death shall be paid to the Participant's estate.

ARTICLE 14. DEFERRALS

The Committee may permit or require a Participant to defer such Participant's receipt of the payment of cash or the delivery of Shares that would otherwise be due to such Participant by virtue of the (1) exercise of an Option, (2) the lapse or waiver of restrictions with respect to Restricted Stock, RSUs, SARs, Deferred Stock, or (3) the satisfaction of any requirements or objectives with respect to Long-Term Performance Awards. If any such deferral election is required or permitted, the Committee shall, in its sole discretion, establish rules and procedures for such payment of deferrals including the crediting of interest or dividend equivalents.

ARTICLE 15. DISCRETION TO REDUCE AWARDS

Notwithstanding any provision to the contrary, except in the event of a Change in Control, the Committee has the discretion to reduce or eliminate an Award that would otherwise be paid to any Participant, including any Covered Employee, based on the Committee's evaluation of Extraordinary Events or other factors.

ARTICLE 16. RIGHTS OF EMPLOYEES

16.1 EMPLOYMENT. Nothing in the Plan shall interfere with or limit in any way the right of the Corporation to terminate any Participant's employment at any time, with or without Cause, nor confer upon any Participant any right to continue in the employ of the Corporation.

16.2 PARTICIPATION. No Employee shall have the right to be selected to receive an Award under this Plan, or, having been so selected, to be selected to receive a future Award.

ARTICLE 17. CHANGE IN CONTROL

17.1 TREATMENT OF AWARDS. Notwithstanding any provision in this Plan to the contrary, upon the occurrence of a Change in Control, unless otherwise specifically prohibited under applicable laws, or by the rules and regulations of any governing governmental agencies or national securities exchanges:

(a) Any and all Options or SARs granted hereunder shall become immediately exercisable in full, and all such Options or SARs shall remain exercisable throughout their entire term notwithstanding the death, Retirement or termination of employment or directorship of the Participant;

(b) Any nonperformance-based restriction periods or restrictions imposed on Shares of Restricted Stock, RSUs, or Shares of Deferred Stock shall lapse; and

(c) All Long-Term Performance Awards and performance-based Awards of Shares of Restricted Stock, RSUs, and Shares of Deferred Stock shall be measured as of the effective date of the Change in Control, and shall be paid out to Participants within thirty (30) days following the effective date of the Change in Control, in a pro rata amount based upon (i) the actual results measured as of the effective date of the Change in Control, and (ii) the length of time within the Performance Cycle which has elapsed prior to the Change in Control.

17.2 TERMINATION, AMENDMENT, AND MODIFICATIONS OF CHANGE-IN-CONTROL PROVISIONS. Notwithstanding any other provision of this Plan or any Award Agreement provision, the provisions of this Article 17 may not be terminated, amended, or modified on or after the date of a Change in Control to affect adversely any Award theretofore granted under the Plan without the prior written consent of the Participant with respect to said Participant's outstanding Awards.



ARTICLE 18. AMENDMENT, MODIFICATION, AND TERMINATION

Subject to Article 17.2 herein, the Board or Committee may at any time and from time to time, alter, amend, suspend, or terminate the Plan in whole or in part; provided, however, that the Committee shall not have the authority to, without shareholder approval, (1) change the limits set forth in Article 4.1, (2) change the minimum Option Price, (3) change eligible Participants to receive Awards, or (4) reprice or alter the Option Price of any Option.

ARTICLE 19. WITHHOLDING

19.1 TAX WITHHOLDING. The Corporation shall have the power and the right to deduct or withhold, or require a Participant to remit to the Corporation, an amount sufficient to satisfy Federal, state, and local taxes, domestic or foreign, required by law or regulation to be withheld with respect to any taxable event arising as a result of this Plan.

19.2 SHARE WITHHOLDING. With respect to withholding required upon the exercise of Options, upon the lapse of restrictions on Restricted Stock, RSUs, SARS, or Deferred Stock, or upon any other taxable event arising as a result of Awards granted hereunder, Participants may elect to satisfy the tax withholding requirement, in whole or in part, by (i) having the Corporation withhold Shares having a Fair Market Value on the date the tax is to be determined equal to the minimum statutory tax withholding rates which could be withheld on the transaction or (ii) the delivery of Shares that have been held for a minimum of six (6) months to the Corporation (including attestation) having a Fair Market Value equal to the amount of the tax withholding obligations related to the transaction. All such elections shall be irrevocable and shall be subject to any restrictions or limitations that the Committee, in its sole discretion, deems appropriate. Delivery or withholding of fractional Shares shall not be permitted.

ARTICLE 20. FORFEITURE

Except on or after a Change in Control or as otherwise provided in the applicable Award Agreement, and notwithstanding any other provisions in the Plan, in the event of (1) a serious breach of conduct by a Participant or former Participant (including, without limitation, any conduct prejudicial to or in conflict with the Corporation or any securities laws violations including any violations under the Sarbanes-Oxley Act of 2002) or (2) any activity of a Participant or former Participant in which the Participant or former Participant solicits or takes away customers or potential customers with whom the Participant or former Participant had contact with or responsibility for during the Participant's or former Participant's employment with the Corporation (individually and collectively referred to as "Misconduct"), the Committee may (a) terminate any outstanding Award granted to the Participant to repay the Corporation any gain realized or payment received upon the exercise or payment of such Award (with such gain

or repayment valued as of the date of exercise or payment), without regard to when such Misconduct is actually discovered by the Corporation. Such termination or repayment obligation shall be effective as of the date specified by the Committee. Any repayment obligation may be satisfied in Shares or cash or a combination thereof (based upon the Fair Market Value of the Shares on the day prior to the repayment) and the Committee may provide for an offset of any future payments owed by the Corporation to such person if necessary to satisfy the repayment obligation. The determination of whether any Participant or former Participant has engaged in a serious breach of conduct or any prohibited solicitation shall be determined by the Committee in good faith and in its sole discretion.

ARTICLE 21. INDEMNIFICATION

Each person who is or shall have been a member of the Committee, or of the Board, shall be indemnified and held harmless by the Corporation against and from any loss, cost, liability, or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit, or proceeding to which he or she may be a party or in which he or she may be involved by reason of any action taken or failure to act under the Plan and against and from any and all amounts paid by him or her in settlement thereof, with the Corporation's approval, or paid by him or her in satisfaction of any judgement in any such action, suit, or proceeding against him or her, provided he or she shall give the Corporation an opportunity at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such persons may be entitled under the Corporation's Charter or Bylaws, as a matter of law, or otherwise, or any power that the Corporation may have to indemnify them or hold them harmless.

ARTICLE 22. SUCCESSORS

All obligations of the Corporation under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Corporation, whether the existence of such successor is the result of a direct or indirect purchase of all or substantially all of the business and/or assets of the Corporation, or a merger, consolidation, or otherwise.

ARTICLE 23. UNFUNDED PLAN

The Plan shall be unfunded and the Corporation shall not be required to segregate any assets that may at any time be represented by Awards under the Plan. Any liability of the Company to any person with respect to any Awards under the Plan shall be based solely upon any contractual obligations that may be effected pursuant to the Plan. Except as provided herein, no such obligation of the Corporation shall be deemed to be secured by any pledge of, or other encumbrance on, any property of the Corporation.

ARTICLE 24. NOTIFICATION UNDER CODE SECTION 83(b)

If the Participant, in connection with the exercise of any Option, or the grant of Shares from an Award of SARs, Restricted Stock, or RSUs, desires to make the election permitted under Code Section 83(b) to include in such Participant's gross income in the year of transfer the amounts specified in Code Section 83(b), then such Participant shall notify the Corporation of the desired election within ten (10) days before the filing of the notice of the election with the Internal Revenue Service in addition to any filing and notification required under regulations issued under Code Section 83(b). The Committee may, in connection with the grant of an Award or at any time thereafter before such an election being made, prohibit a Participant from making the election described above.

ARTICLE 25. OTHER PLANS

Nothing in this Plan shall be construed as limiting the authority of the Committee, the Board of Directors, the Corporation or any Subsidiary to establish any other compensation plan, or as in any way limiting its or their authority to pay bonuses or supplemental compensation to any persons employed by the Company or a Subsidiary, whether or not such person is a Participant in this Plan and regardless of how the amount of such compensation or bonus is determined. However, no such plan will be established or operated in a way that entitles or allows a Covered Employee to receive an award under such plan as a substitution or supplement for not achieving goals under this Plan.

ARTICLE 26. LEGAL CONSTRUCTION

25.1 GENDER AND NUMBER. Except where otherwise indicated by the context, any masculine term used herein also shall include the feminine; the plural shall include the singular and the singular shall include the plural.

25.2 SEVERABILITY. In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.

25.3 REQUIREMENTS OF LAW. The granting of Awards and the issuance of Shares under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.

25.4 GOVERNING LAW. To the extent not preempted by Federal law, the Plan, and all agreements hereunder, shall be construed in accordance with and governed by the laws of the state of Ohio, without reference to its choice of law rules.

EXECUTIVE DEFERRED COMPENSATION PLAN

OF

HUNTINGTON BANCSHARES INCORPORATED

EFFECTIVE OCTOBER 1, 2001

AMENDED AND RESTATED OCTOBER 1, 2002

AMENDED AND RESTATED FEBRUARY 18, 2004

EXECUTIVE DEFERRED COMPENSATION PLAN OF HUNTINGTON BANCSHARES INCORPORATED

TABLE OF CONTENTS

ARTICLE 1 - DEFINITIONS

1.01 Accounts

- 1.02 **Beneficiary**
- 1.03 Bonus
- 1.04 Bonus Deferral Agreement 1.05
- Bonus Deferrals
- 1.06 Cash Deferral Account 1.07
- Change in Control
- 1.08 Code 1.09
- Common Stock 1.10
- Common Stock Deferral Account
- 1.11 Compensation
- 1.12 Compensation Committee
- 1.13 Corporation
- 1.14 **Deferrals**
- Distribution Election Form 1.15
- 1.16 Effective Date
- 1.17 Eligible Employee
- 1.18 **ERISA**
- 1.19 Investment Funds
- 1.20 Long-Term Incentive
- 1.21 Long-Term Incentive Deferral Agreement
- 1.22 Long-Term Incentive Deferrals
- 1.23 Member
- 1.24 Net Shares
- 1.25 <u>Plan</u>
- 1.26 Plan Year
- 1.27 Restricted Stock Award
- 1.28 Restricted Stock Deferral Agreement
- Restricted Stock Deferrals 1.29
- 1.30 Salary Deferral Agreement
- Salary Deferrals 1.31
- Stock Option Deferral Account 1.32
- 1.33 Stock Option Deferral Agreement
- 1.34 Stock Option Deferrals
- 1.35 <u>Trust</u>
- 1.36 Valuation Date

6

PAGE

EXECUTIVE DEFERRED COMPENSATION PLAN OF HUNTINGTON BANCSHARES INCORPORATED

TABLE OF CONTENTS

ARTICLE 2 - PARTICIPATION

2.01 <u>Eligibility</u>

2.02 Participation

ARTICLE 3 - CONTRIBUTIONS

| 3.01 <u>Amount of</u> | <u>Deferral</u> | <u>Contributions</u> |
|-----------------------|-----------------|----------------------|
|-----------------------|-----------------|----------------------|

- 3.02 Stock Option Exercise
- 3.03 Investment of Accounts
- 3.04 <u>Valuation of Accounts</u>
- 3.05 Vesting of Account
- 3.06 Individual Accounting

ARTICLE 4 - PAYMENT OF ACCOUNT

- 4.01 Payment of Accounts
- 4.02 <u>Method of Payment</u>
- 4.03 <u>Hardship Distributions</u>
- 4.04 <u>Other Distributions</u>
- 4.05 Accelerated Tax Distributions
- 4.06 Compensation Committee Deferral of Distributions
- 4.07 <u>Death Benefit</u>
- 4.08 Designation of Beneficiary

ARTICLE 5 - CHANGE IN CONTROL

- 5.01 Definition of Change in Control
- 5.02 Application in Change in Control
- 5.03 Payments to and by the Trust
- 5.04 Legal Fees and Expenses

PAGE

EXECUTIVE DEFERRED COMPENSATION PLAN OF HUNTINGTON BANCSHARES INCORPORATED

TABLE OF CONTENTS

ARTICLE 6 - GENERAL PROVISIONS

| 6.01 | Establishment of Rules | 29 |
|---------------|------------------------------|----|
| 6.02 | Funding | 30 |
| 6.03 | No Contract of Employment | 31 |
| 6.04 | Facility of Payment | 31 |
| 6.05 | Withholding Taxes | 32 |
| 6.06 | Nonalienation | 32 |
| 6.07 | Construction | 32 |
| 6.08 | Limitations on Liability | 33 |
| 6.09 | Administrative Expense | 33 |
| 6.10 | Claims and Review Procedures | 33 |
| 6.11 | Illegal or Invalid Provision | 35 |
| 6.12 | Gender and Number | 36 |
| 6.13 | Successors | 36 |
| ARTICLE 7 - A | MENDMENT OR TERMINATION | 37 |

EXECUTIVE DEFERRED COMPENSATION PLAN OF HUNTINGTON BANCSHARES INCORPORATED

INTRODUCTION

This Executive Deferred Compensation Plan of Huntington Bancshares Incorporated is authorized by the Board of Directors of Huntington Bancshares Incorporated effective on and after October 1, 2001. The purpose of this Plan is to provide a means to defer receipt of compensation and current income tax liability thereon for selected managers and highly compensated employees in addition to the amounts that can be deferred under the Corporation's qualified retirement plan.

This Plan is intended to constitute a nonqualified unfunded deferred compensation plan for a select group of management or highly compensated employees under Title I of the ERISA. All benefits payable under the Plan shall be paid out of the general assets of Huntington Bancshares Incorporated. Huntington Bancshares Incorporated may establish and fund a trust as provided herein in order to aid it in providing benefits due under the Plan.

EXECUTIVE DEFERRED COMPENSATION PLAN OF HUNTINGTON BANCSHARES INCORPORATED

ARTICLE 1. DEFINITIONS

- 1.01 "Accounts" means the device used to measure and determine the amount of deferred compensation to be paid to a Member or Beneficiary under the Plan and may include the Cash Deferral Account, the Common Stock Deferral Account, and the Stock Option Deferral Account.
- 1.02 "Beneficiary" means the individual, trust or other recipient to whom a deceased Member's benefits are payable, as provided in Section 4.07.
- 1.03 "Bonus" means any annual bonus or other similar bonus payable to an Eligible Employee and any other bonus payable to an Eligible Employee and designated by the Corporation through other program documents or otherwise as eligible for deferral under this Plan.
- 1.04 "Bonus Deferral Agreement" means the agreement entered into by the Member pursuant to Section 2.02 under which he elects to defer all or a portion of his Bonus under this Plan.
- 1.05 "Bonus Deferrals" means the amount of contributions credited to a Member's Accounts under Section 3.01, with respect to his Bonus.

-1-

- 1.06 "Cash Deferral Account" means the account credited with Salary Deferrals, cash Bonus Deferrals and cash Long-Term Incentive Deferrals and the earnings credited to that Account pursuant to Section 3.03.
- 1.07 "Change in Control" means the events described in Section 5.01.
- 1.08 "Code" means the Internal Revenue Code of 1986, as amended from time to time.
- 1.09 "Common Stock" means the Huntington Bancshares Incorporated common stock with no par value per share.
- 1.10 "Common Stock Deferral Account" means the account credited with Common Stock from deferred Common Stock in relation to Bonus Deferrals, Long-term Incentive Deferrals and Restricted Stock Deferrals and the dividend equivalents with respect to the Common Stock credited to that Account pursuant to Section 3.03.
- 1.11 "Compensation" means base salary or commissions earned by an Eligible Employee from the Corporation for each payroll period during a Plan Year, including any salary reduction contributions made under any plan maintained by the Corporation pursuant to Sections 401(k), 125 and 132(f) of the Code.
- 1.12 "Compensation Committee" means the Compensation and Stock Option Committee of the Board of Directors of the Corporation.

- 1.13 "Corporation" means Huntington Bancshares Incorporated, a Maryland corporation, together with any and all subsidiaries, and any successor thereto as provided in Section 6.13. A subsidiary or subsidiaries means any corporation or other entity whose financial statements are consolidated with the Corporation.
- 1.14 "Deferrals" means collectively the Salary, Bonus, Long-Term Incentive, Restricted Stock and Net Shares, unless indicated otherwise.
- 1.15 "Distribution Election Form" means the form or forms completed by a Member to elect alternative payment options and/or alternative payout timing.
- 1.16 "Effective Date" means October 1, 2001.
- 1.17 "Eligible Employee" means a manager or highly compensated employee of the Corporation who has been selected by the Compensation Committee or its delegate to participate in this Plan as described in Section 2.01.
- 1.18 "ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time.
- 1.19 "Investment Funds" mean such investment funds that the Corporation may, in its discretion, make available in determining the earnings (or losses) on a Member's Cash Deferrals.

- 1.20 "Long-Term Incentive" means any long-term award payable to an Eligible Employee by the Corporation and designated by the Corporation through other program documents or otherwise as eligible for deferral under this Plan.
- 1.21 "Long-Term Incentive Deferral Agreement' means the agreement entered into by the Member pursuant to Section 2.02 under which he elects to defer all or a portion of his Long-Term Incentive under this Plan.
- 1.22 "Long-Term Incentive Deferrals" means the amount of contributions credited to a Member's Accounts under Section 3.01, with respect to his Long-Term Incentive.
- 1.23 "Member" means each Eligible Employee who has made the election described in Section 2.02.
- 1.24 "Net Shares" with respect to an election made pursuant to Section 3.02, means the difference between the number of shares of Common Stock subject to the stock option exercise and the number of shares of Common Stock delivered to satisfy the stock option exercise price, less any shares used to satisfy FICA or any other taxes due upon the option exercise as may be designated by the Corporation.
- 1.25 "Plan" means the Executive Deferred Compensation Plan of Huntington Bancshares Incorporated.

- 1.26 "Plan Year" means the twelve (12)-month period beginning on any January 1; the first Plan Year shall begin October 1, 2001 and ends December 31, 2001.
- 1.27 "Restricted Stock Award" means any restricted stock awards of Common Stock payable to an Eligible Employee by the Corporation and designated by the Corporation through a restricted stock award document or otherwise as eligible for deferral under this Plan.
- 1.28 "Restricted Stock Deferral Agreement" means the agreement entered into by the Member pursuant to Section 2.02 under which he elects to defer all or a portion of his Restricted Stock Award under the Plan.
- 1.29 "Restricted Stock Deferrals" means the amount of contributions credited to a Member's Accounts under Section 3.01, with respect to his Restricted Stock Award.
- 1.30 "Salary Deferral Agreement" means the agreement entered into by the Member pursuant to Section 2.02 under which he elects to defer all or a portion of his Compensation under this Plan.
- 1.31 "Salary Deferrals" means the amount of contributions credited to a Member's Account under Section 3.01, with respect to his Compensation.
- 1.32 "Stock Option Deferral Account" means the account credited with the deferred Net Shares and the dividend equivalents with respect to those Net Shares credited to that Account pursuant to Section 3.03.

- 1.33 "Stock Option Deferral Agreement" means the agreement entered into by the Member pursuant to Section 2.02 under which he elects to defer all or a portion of his Net Shares under this Plan.
- 1.34 "Stock Option Deferrals" means the amount of contributions credited to a Member's Accounts under Section 3.01, with respect to his Net Shares.
- 1.35 "Trust" means the trust that may be established by the Corporation as provided in Section 6.02(b).
- 1.36 "Valuation Date" means the last business day of each calendar month. All distributions under the Plan shall be based upon the value of the Member's Accounts as of the Valuation Date specified in Article 4 with respect to the distribution and shall include any contributions made by a Member, but not yet credited to the Member's Accounts.

ARTICLE 2. PARTICIPATION

2.01 Eligibility

- (a) In addition to a designation of eligibility to participate in this Plan contained in other Corporation programs for its managers and highly compensated employees, the Compensation Committee or its delegate, in its sole discretion, shall select from time-to-time either individually or by class, those managers and highly compensated employees of the Corporation who shall be eligible to participate in this Plan under Section 2.02. The Compensation Committee may select different employees or classes of employees to participate in the various Deferrals. The Compensation Committee shall make its selection from those managers and highly compensated employees who it deems to be in a select group of management or highly compensated employees as defined under Title I of ERISA. Said employees shall be notified of their eligibility for participation in the Plan by the Corporation as soon as practicable after the Compensation Committee has made its selection and, for purposes of authorizing any Deferrals under Section 3.01, in any event prior to the first day of eligibility after said employee is designated an Eligible Employee. An Eligible Employee may become a Member for any or all Deferrals he is eligible for under Section 3.01.
- (b) In its sole discretion, the Compensation Committee may withdraw its approval for participation in the Plan for any Member at any time with respect to any future deferral opportunity whether or not a deferral election has been made by the Member. In the event of such withdrawal, the Member with existing Accounts will remain a Member in relation to the right to direct investment and elect distribution options from those Accounts. Such Member shall be notified of such

-7-

withdrawal in writing as soon as practicable following such action. Notwithstanding the foregoing, in the event a Change in Control occurs, the Compensation Committee may not thereafter withdraw its participation approval for any Member.

2.02 Participation

- (a) Initial Plan Year.
 - (i) Salary Deferrals will be available beginning in 2002 and to be effective must be made by an Eligible Employee by executing and delivering to the Corporation a Salary Deferral Agreement by December 31, 2001. The Salary Deferral Agreement shall apply to Compensation earned by the Eligible Employee in the payroll periods beginning after December 31, 2001.
 - (ii) An election to become a Member and authorize a Bonus Deferral to be effective for a 2001 Bonus (payable in 2002) must be made by an Eligible Employee, by executing and delivering to the Corporation, an irrevocable Bonus Deferral Agreement by December 31, 2001.
 - (iii) An election to become a Member and authorize a Long-Term Incentive Deferral to be effective for cycle eight (8) payments ending on December 31, 2002 payable in 2003 must be made by the Eligible Employee by executing and delivering to the Corporation an irrevocable Long-term Incentive Deferral Agreement as provided in Section 2.02(b)(iii).
 - (iv) Stock Option Deferral will be available beginning in 2003 and to be effective must be made by an Eligible Employee by executing and delivering to the Corporation a Stock Option Deferral Agreement as provided in Section 2.02(b)(iv).

-8-

(b) Initial Participation for 2002 Plan Year and Thereafter.

- (i) In the case of a newly hired Eligible Employee or an employee who first becomes an Eligible Employee during a Plan Year, that Eligible Employee must execute and deliver to the Corporation a Salary Deferral Agreement within 30 days of notification of eligibility to participate in the Plan. Salary Deferrals will begin in the first full payroll period following receipt of the Salary Deferral Agreement.
- (ii) For purposes of Bonus Deferrals, the initial irrevocable Bonus Deferral Agreement must be executed and delivered to the Corporation before the later of the date established for current Members to submit their Plan Year Bonus Deferral Agreement or thirty (30) days after being designated an Eligible Employee, to apply to the Bonus attributable to that Plan Year, which becomes payable in the next following Plan Year. The Bonus Deferral Agreement for subsequent Plan Years may be changed in accordance with the procedures in paragraph (c)(ii) below. Notwithstanding the foregoing, in relation to any other bonus, the irrevocable Bonus Deferral Agreement must be made at such time and in such manner as established by the Corporation in its discretion.
- (iii) For purposes of Long-Term Incentive Deferrals and for Restricted Stock Deferrals, the initial irrevocable Long-Term Incentive Deferral Agreement or Restricted Stock Deferral Agreement, as the case may be, must be executed and delivered to the Corporation before the later of the date

established for current Members to submit their next plan year's payment Deferral Agreement or thirty (30) days after being designated an Eligible Employee for purposes of the Long-Term Incentive Deferral and/or Restricted Stock Award Deferral.

- (iv) For purposes of **Stock Option Deferrals** effective initially in 2003, the initial Stock Option Deferral Agreement as it applies to a specific Common Stock option must be executed and delivered to the Corporation:
 - (A) no later than December 31 of the Plan Year immediately preceding the Plan Year in which the Member will exercise such option, and
 - (B) at least six (6) months before the exercise of the Common Stock option.
 - This Stock Option Deferral Agreement is irrevocable, but will only apply to the stock option(s) designated in the Agreement.

(c) Continuing Participation.

(i) After the initial year for eligibility, an Eligible Employee may authorize or change a previously authorizedSalary Deferral for any Plan Year by executing and delivering to the Corporation a Salary Deferral Agreement at a date determined by the Corporation, but no later than December 31 of the Plan Year immediately preceding the Plan Year in which the Compensation would otherwise be paid to the Member. Once a Salary Deferral Agreement is made, it will remain in effect for that Plan Year and subsequent Plan Years until suspended or changed in accordance with these terms.

- (ii) A Bonus Deferral Agreement must be executed for any Plan Year that aBonus Deferral is desired. An irrevocable Bonus Deferral Agreement must be delivered to the Corporation prior to the last business day of March in order to defer that Plan Year's Bonus, payable in the following Plan Year. A Bonus Deferral Agreement is irrevocable, but may be changed for subsequent Plan Years in accordance with these terms.
- (iii) A Long-Term Incentive Deferral Agreement must be executed for each payout year that aLong-Term Incentive Deferral is desired. The Long-Term Incentive Deferral Agreement must be delivered to the Corporation prior to the last business day of March of the year prior to the plan year of payment. This Long-Term Incentive Deferral Agreement is irrevocable, but will only apply to the Long-Term Incentive designated in the Agreement.
- (iv) A Restricted Stock Deferral Agreement must be executed for each payment that aRestricted Stock Deferral is desired. The Restricted Stock Deferral Agreement must be delivered to the Corporation more than six (6) months prior to the vesting in the Restricted Stock Award and in the Plan Year prior to the year in which the Restricted Stock Award vests. This Deferral Agreement is irrevocable, but will only apply to the Restricted Stock Award designated in the Agreement.
- (v) A Stock Option Deferral Agreement must be executed for each Common Stock option exercise that aStock Option Deferral is desired. The Stock Option Deferral Agreement must be delivered to the Corporation:
 - (A) no later than December 31 of the Plan Year immediately preceding the Plan Year in which the Member will exercise such option, and

-11-

(B) at least six (6) months before the exercise of the Common Stock option.

This Stock Option Deferral Agreement is irrevocable, but will only apply to the stock option(s) designated in the Agreement.

- (d) **Participation Procedure.** In order to commence participation in the Plan, an Eligible Employee must complete and timely deliver to the Corporation the following forms:
 - (i) one or more of the Deferral Agreements referred to above;
 - (ii) a Distribution Election Form which applies to one or more of the Accounts for which Deferrals are elected; and
 - (iii) a Beneficiary Designation Form which applies to the Accounts.
- (e) Election Not to Defer. An Eligible Employee or Member may elect from time-to-time not to defer any amount he is eligible to defer hereunder without affecting the eligibility to defer at any time in the future as long as he remains an Eligible Employee.

ARTICLE 3. CONTRIBUTIONS

3.01 Amount of Deferral Contributions

The amount of contributions to be recorded on the books of the Corporation on behalf of a Member's Accounts pursuant to this Section 3.01 shall be equal to the total of the Deferrals described herein.

- (a) Salary Deferrals. For each payroll period beginning on or after the effective date of an Eligible Employee's Salary Deferral Agreement, his Cash Deferral Account shall be credited with an amount of Salary Deferrals, if applicable. The amount of Salary Deferrals shall be equal to the designated percentage of Compensation elected by the Member in his Salary Deferral Agreement by agreeing to accept a reduction in Compensation equal to a stated whole percentage of Compensation per payroll period which is not less than five percent (5%) nor more than ninety (90%) of Compensation.
- (b) Bonus Deferral. The amount of Bonus Deferrals shall be elected by the Member by agreeing to accept a reduction in Bonus equal to a stated whole percentage in his Bonus Deferral Agreement for that year, which is not less than five percent (5%) or one hundred (100) shares of Common Stock nor more than ninety (90%) of Bonus.
- (c) Long-Term Incentive Deferral. The amount of a Long-Term Incentive Deferral shall be elected by the Member by agreeing to accept a reduction in the Long-Term Incentive equal to a stated whole percentage in the applicable Long-Term Incentive Deferral Agreement, which is not less than five percent (5%) or one hundred (100) shares of Common Stock nor more than one hundred percent (100%) of the Long-Term Incentive.

-13-

- (d) Restricted Stock Deferral. The amount of a Restricted Stock Deferral shall be elected by the Member by agreeing to accept a reduction in the Restricted Stock Award equal to a stated whole percentage of his Common Stock covered in the applicable Restricted Stock Deferral Agreement, which is not less than one hundred (100) shares of Common Stock nor more than one hundred percent (100%) of the award (in whole shares).
- (e) Stock Option Deferral. The amount of a Stock Option Deferral shall be elected by the Member by agreeing to accept a reduction in Net Shares deliverable to him equal to a stated whole percentage of his Net Shares for that Stock Option Deferral Agreement, which is not less than one hundred (100) shares of Common Stock nor more than one hundred percent (100%) of the Net Shares (in whole shares).
- (f) The Corporation, in its sole discretion, may allow Deferrals in dollar amounts, or a combination of percentages and dollar amounts.
- (g) No election for Deferrals will be effective unless the cash amount or Common Stock amount payable to the Member to which the Deferral applies is sufficient to satisfy such election.
- (h) In no event shall the Deferrals of Compensation, Bonus, Long-Term Incentive, Common Stock or Net Shares be reduced below the amount required for federal, state and local tax and any other required or elected withholding amounts (including amounts elected under the Corporation's various benefit plans).

3.02 Stock Option Exercise

- (a) This Section describes the special procedures for deferring the delivery and receipt of Common Stock which a Member may receive from the exercise of a nonqualified stock option granted to the Member by the Corporation. The stock options are governed by the stock option plan under which they are granted. No stock options or shares of Common Stock are authorized to be issued under the Plan. A Member who elects to defer receipt of Common Stock issuable upon the exercise of stock options will have no rights as a stockholder of the Corporation with respect to allocations made to his Stock Option Deferral Account except the right to receive dividend equivalent allocations as hereafter described.
- (b) A Member may elect to defer receipt of Net Shares of Common Stock resulting from a stock-for-stock exercise of an exercisable stock option issued to the Member by completing and submitting to the Corporation his Stock Option Deferral Agreement as provided in Section 2.02. The stock option exercise must occur on or prior to the expiration date of the stock option and must be accomplished by delivering Common Stock or using another acceptable method, such as, attestation, on or prior to the exercise date, shares of Common Stock which have been personally owned by the Member for at least six (6) months prior to the exercise date and have not been used in a stock swap in the prior six (6) months. A Member's Stock Option Deferral Agreement shall not be effective if the stock option as to which the Member has made the deferral election terminates prior to the exercise date selected by the Member. If the Member dies or fails to deliver shares of Common Stock which have been personally owned by the Ameber. If the Member dies or fails to deliver shares of Common Stock which have been personally owned by the Member. If the Member dies or fails to deliver shares of Common Stock which have been personally owned by the Member to the exercise date (and have not been used in a stock

swap in the prior six (6) months) in payment of the exercise price, then the Stock Option Deferral Agreement shall not be effective. Only whole Net Shares may be deferred.

3.03 Investment of Accounts

- (a) The Accounts of each Member shall be credited with an additional amount of hypothetical net earnings (or losses) determined under this Section.
- (b) Except for the Common Stock Deferral Account and the Stock Option Deferral Account, each Member shall elect the manner in which his Accounts are to be credited with net earnings (and losses) by designating how the Accounts are to be invested on a hypothetical basis from among the Investment Funds. The election shall be made in writing on a form provided by the Corporation. An investment election shall be effective for the Valuation Date established by the Corporation following its receipt. Modifications may be made to investment elections on the same basis.
- (c) If the Corporation exercises its discretion to establish the Trust, it reserves the right to determine the amount of contributions to the Trust and the types of investments used, including, but not limited to, mutual funds, annuities and life insurance contracts.
- (d) Bonus Deferrals, Long-Term Incentive Deferrals, Restricted Stock Deferrals and Stock Option Deferrals of Common Stock may be maintained in their respective Accounts on the books of the Corporation or the Common Stock may be held in the Trust.
- (e) Notwithstanding any of the other provisions of the Plan, the Member may elect to

-16-

have any Salary Deferrals, cash Bonus Deferrals and cash Long-Term Incentive Deferrals invested in Common Stock. The number of shares of Common Stock to be credited to a Member's Cash Deferral Account by virtue of an election to invest a portion of a Salary Deferral, cash Bonus Deferral or a cash Long-Term Incentive Deferral in Common Stock shall be determined on the date of such Deferral in accordance with such procedures as the Corporation shall establish. Such Common Stock may be maintained in the Cash Deferral Account on the books of the Corporation or the Common Stock may be held in the Trust.

3.04 Valuation of Accounts

- (a) On the first Valuation Date, the Member's Accounts shall equal:
 - (i) the amount of the Member's Deferrals, if any, credited to the Member's Accounts during the period from the Effective Date through the first Valuation Date; plus
 - the proportionate share of the net earnings or losses, if any, since the Effective Date on the Investment Funds, made available in determining the net earnings or losses on a Member's Deferrals;
 - (iii) for purposes of the Common Stock valuation, all dividend equivalents payable in relation to Common Stock shall be credited in the form of additional whole and fractional shares of Common Stock since the Effective Date.
- (b) On each subsequent Valuation Date, the Member's Accounts shall equal:
 - (i) the Member's Accounts balance as of the immediately preceding Valuation Date; plus

-17-

- the proportionate share of the net earnings or losses, if any, since the immediately preceding Valuation Date on the Investment Funds, made available in determining the net earnings or losses on a Member's Deferrals; plus
- (iii) all dividend equivalents payable in relation to the Common Stock (credited in the form of additional shares) since the immediately preceding Valuation Date; plus
- (iv) the then value of the Member's Deferrals, if any, credited since the immediately preceding Valuation Date; and less
- (v) any payments or distributions made in accordance with the terms of the Plan from the Member's Accounts since the immediately preceding Valuation Date.
- (c) The Corporation reserves the right to change from time-to-time the procedures used in valuing the Accounts or crediting (or debiting) the Accounts if it determines that such an action is justified in that it results in a more accurate reflection of the fair market value of assets. In the event of a conflict between the provisions of this Section and such new administrative procedures, those new administrative procedures shall prevail.
- (d) In the event that the Corporation determines that any recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase or exchange of Common Stock or other securities of the Corporation, issuance of warrants or other rights to purchase Common Stock or other securities of the Corporation, or other similar corporate transactions or events affects the Common Stock, an appropriate adjustment to the Member's Accounts shall be made to prevent reduction or enlargement of the Member's benefits under the Plan.

-18-

3.05 Vesting of Account

The Member shall be fully vested in his Deferrals and earnings credited to his Accounts.

3.06 Individual Accounting

The Corporation shall maintain, or cause to be maintained, records showing the individual balances of each of the Member's Accounts. At least once a year, each Member shall be furnished with a statement setting forth the value of his Accounts.

ARTICLE 4. PAYMENT OF ACCOUNT

4.01 Payment of Accounts

A Member shall be entitled to receive payment of his Accounts upon the Member's termination of employment from the Corporation for any reason.

4.02 Method of Payment

- (a) Unless otherwise elected as provided hereinafter, payment of the Accounts shall be made in a single lump sum payment on or as soon as administratively possible following the Valuation Date for the month following the month in which the Member terminates employment. Any portion of the Cash Deferral Account consisting of Common Stock shall be payable only in kind.
- (b) In lieu of receiving an immediate lump sum payment, the Member may elect from time-to-time on a Distribution Election Form to receive his Accounts:
 - (i) in annual installments over a period allowed by the Corporation ranging from two (2) to ten (10) years as follows:
 - (A) with respect to Cash Deferral Accounts, except for any portion of the Cash Deferral Account consisting of Common Stock, annual cash payments calculated by multiplying (I) the balance in the Cash Deferral Account as of the Valuation Date on which such installment payment is being made, times (II) a fraction equal to the reciprocal of the number of years remaining in the annual installment period elected by the Member; and

-20-

- (B) with respect to Common Stock Deferral Accounts and Stock Option Deferral Accounts and any Common Stock portion of the Cash Deferral Account, approximately equal annual installments of Common Stock, or
- (ii) in a lump sum payment or in installments as provided in subparagraph (i) above beginning as of a specified Valuation Date either before or after his termination of employment in accordance with terms established by the Corporation.
- (c) An installment payout election under paragraph (b) above is subject to the approval of the Corporation and to be effective must be made no later than:
 - (i) the end of the Plan Year prior to the Plan Year in which the election will be effective; and
 - (ii) at least six (6) months prior to the commencement payout date.
- (d) A Member may complete one Distribution Election Form at the time of the Member's initial Deferral hereunder electing a date or dates on which any Member Deferrals will be distributed or begin to be distributed to the Member other than the Member's termination of employment. The initial Deferral election period on the Distribution Election Form must be at least three (3) Plan Years from the Plan Year in which that Deferral occurs.

The Member may subsequently extend any Deferral payout date, if the Member's subsequent Distribution Election Form is executed and delivered to the Corporation at least two (2) Plan Years prior to the then elected payout date.

(e) The Corporation may establish rules and procedures which allow a Member to complete a separate Distribution Election Form for each Account within

-21-
the Member's Accounts or for an amount in any one of his Accounts based on the value of separate Deferral Agreements. In either situation, the Corporation shall revise the Distribution Election Form procedures under paragraph (d) above to allow current and subsequent elections to apply to each Account or to amounts in the Accounts based on separate deferral agreements.

4.03 Hardship Distributions

The Corporation may, pursuant to rules adopted by it and applied in a uniform manner, accelerate the date of distribution of a Member's Accounts (except the Common Stock Deferral Account and the Stock Option Deferral Account) because of hardship at any time. "Hardship" shall include an unforeseeable, severe financial condition resulting from (a) a sudden and unexpected illness or accident of the Member or his dependents (as defined in Section 152(a) of the Code); (b) loss of the Member's property due to casualty; or (c) other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Member, but which may not be relieved through other available resources of the Member, as determined by the Corporation. The amount of the distribution may not be in excess of the amount of the financial need to satisfy the hardship, including any amounts necessary to pay any federal, state or local tax reasonably anticipated to result from the distribution.

4.04 Other Distributions

A Member shall be permitted to accelerate payment from his Accounts on or as soon as administratively possible following the Valuation Date coincident with or occurring after at least thirty (30) days written notice to the Corporation. Such payment shall be subject to a

| | 2 | 2 | |
|---|----|---|---|
| - | -2 | 2 | 1 |
| | | | |

ten percent (10%) penalty reduction to the Accounts and the Member's future Deferrals shall be suspended. The Member shall be permitted to reauthorize Deferrals for the Plan Year following the expiration of a twelve (12)-month period after said distribution, provided he still is an Eligible Employee.

4.05 Accelerated Tax Distributions

In the event any Deferral is finally determined to be income taxable to the Eligible Employee prior to any distribution event hereunder, then the Account holding that Deferral shall be distributable to the Eligible Employee. The Eligible Employee shall notify the Corporation of the final determination of taxability and will provide all information required by the Corporation.

4.06 Compensation Committee Deferral of Distributions

The Compensation Committee may, in its sole discretion, defer payment to a future date of any otherwise scheduled distribution that would otherwise result in the loss of a corporate income tax deduction under Internal Revenue Code Section 162(m).

4.07 Death Benefit

If a Member dies before payment of the entire balance of his Accounts, an amount equal to the unpaid portion thereof as of the date of his death shall be made in a single lump sum payment to his Beneficiary on or as soon as administratively possible following the Valuation Date for the month following the month in which the Member dies.

4.08 Designation of Beneficiary

Each Member shall file with the Corporation a written designation of one or more persons, trust or other recipient as the Beneficiary who shall be entitled to receive the amount, if any, payable from his Accounts under the Plan upon his death pursuant to this Section 4.07. A Member may, from time to time revoke or change his Beneficiary designation without the consent of any prior Beneficiary by filing a new designation with the Corporation. The last such designation received by the Corporation shall be controlling; provided however, that no designation, or change or revocation thereof, shall be effective unless received by the Corporation prior to the Member's death, and in no event shall it be effective as of a date prior to such receipt. If no such Beneficiary designation is in effect at the time of the Member's death, or if no designated Beneficiary survives the Member, the Corporation shall distribute the Member's benefits to the following persons in the following order of priority:

- (a) The Member's surviving spouse;
- (b) The Member's surviving children, in equal shares; or
- (c) The legal representative of the Member's estate.

ARTICLE 5. CHANGE IN CONTROL

5.01 Definition of Change in Control

Change in Control means any of the following events:

- (a) Any "person" (as such term is used in Sections 13(d) and 14(d) of the Exchange Act as in effect as of the date of this Plan), other than the Corporation or any "person" who as of the Effective Date is a director or officer of the Corporation or whose shares of Common Stock of the Corporation are treated as "beneficially owned" (as such term is used in Rule 13d-3 of the Exchange Act as in effect as of the Effective Date) by any such director or officer, becomes the beneficial owner, directly or indirectly, of securities of the Corporation representing twenty-five percent (25%) or more of the combined voting power of the Corporation's then outstanding securities;
- (b) Individuals who, as of the Effective Date, constitute the Board of Directors of the Corporation (the "incumbent board") cease for any reason to constitute at least a majority of such Board of Directors, provided, however, that any individual becoming a director subsequent to the date hereof whose election, or nomination for election, was approved by a vote of at least a majority of the directors comprising the "incumbent board" shall be considered as though such individual were a member of the "incumbent board," but excluding for this purpose any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a "person" other than such Board of Directors;

| 2 | 5 | |
|---|---|---|
| 2 | э | • |

- (c) A merger or consolidation of the Corporation, other than a merger or consolidation in which the voting securities of the Corporation immediately prior to the merger or consolidation continue to represent (either by remaining outstanding or being converted into securities of the surviving entity) fifty-one percent (51%) or more of the combined voting power of the Corporation or surviving entity immediately after the merger or consolidation with another entity;
- (d) A sale, exchange, lease, mortgage, pledge, transfer, or other disposition (in a single transaction or a series of related transactions) of all or substantially all of the assets of the Corporation which shall include, without limitation, the sale of assets or earning power aggregating more than fifty percent (50%) of the assets or earning power of the Corporation on a consolidated basis;
- (e) A liquidation or dissolution of the Corporation;
- (f) A reorganization, reverse stock split, or recapitalization of the Corporation which would result in any of the foregoing; or
- (g) A transaction or series of related transactions having, directly or indirectly, the same effect as any of the foregoing.

5.02 Application in Change in Control

To the extent applicable, the provisions of this Article shall control and shall supersede any other provisions of the Plan to the extent inconsistent with the provisions of this Article.

5.03 Payments to and by the Trust

If the Corporation determines that it is probable that a Change in Control may occur within the six (6)-month period immediately following the date of determination, or if a Change in Control in fact occurs in those situations where the Corporation has not otherwise made such a determination, the Corporation shall make a contribution to the Trust (if in existence at the date of determination or the date of the Change in Control, as the case may be) in accordance with the provisions of the Trust. Solely for purposes of determining the amount of such contribution (but in no way in limitation of the Corporation's liability under the Plan as determined under other provisions of the Plan), the Corporation's total liability under the Plan shall be equal to the value of all Accounts established under the Plan, which remain unpaid by the Corporation as of the date of determination or the date as soon as possible after the date of determination or of the Change in Control, as the case may be, whether or not amounts are otherwise currently payable to Members or Beneficiaries in cash and/or Common Stock. Further the Corporation may, in its discretion, make other contributions to the Trust from time-to-time for purposes of providing benefits hereunder, whether or not a Change in Control has occurred or may occur.

Notwithstanding the foregoing, any contributions to the Trust, as well as any net earnings or losses thereon, shall be at all times subject to the provisions of the Trust.

5.04 Legal Fees and Expenses

The Corporation shall reimburse any Member or Beneficiary for all reasonable legal fees and expenses incurred by such Member or Beneficiary after the date of any Change in Control in seeking to obtain any right or benefit provided by the Plan.

ARTICLE 6. GENERAL PROVISIONS

6.01 Establishment of Rules

The Corporation shall be responsible for providing for the general administration of the Plan and for carrying out the provisions of the Plan. The Corporation from time-totime shall establish rules for the administration of the Plan and the transaction of its business. The Corporation shall have total and complete discretion to interpret the Plan, including, but not limited to, the discretion to:

- (a) determine all questions arising in the administration, interpretation and application of the Plan, including the power to construe and interpret the Plan;
- (b) decide all questions relating to an individual's eligibility for benefits and the amounts thereof;
- (c) decide all facts relevant to the determination of eligibility for benefits or participation;
- (d) make such adjustments which it deems necessary or desirable to correct any arithmetical or accounting errors; and
- (e) determine the amount, form and timing of any distribution to be made hereunder.

In making its decision, the Corporation shall be entitled to, but need not rely upon, information supplied by a Member, Beneficiary, or representative thereof. The Corporation may correct any defect, supply any omission, or reconcile any inconsistency in such manner and to such extent as it shall deem necessary to carry out the purposes of this Plan. The Corporation's decisions in such matters shall be binding and conclusive as to all parties. In providing for the administration of the Plan, the Corporation may delegate responsibilities for the operation and administration of the Plan by a written document or documents filed with the Plan records.

-29-

6.02 Funding

- (a) All amounts payable in accordance with this Plan shall constitute a general unsecured obligation of the Corporation. Such amounts, as well as any administrative costs relating to the Plan, shall be paid out of the general assets of the Corporation, to the extent not paid by a grantor trust established pursuant to paragraph (b) below. To the extent the Member or any other person acquires a right to receive benefits under this Plan, such right shall be no greater than the right of any unsecured general creditor of the Corporation.
- (b) The Corporation may, for administrative reasons, establish a grantor trust for the benefit of Members participating in the Plan. The Corporation reserves the right to determine the amount of contributions to the Trust and the types of investments used, including, but not limited to, mutual funds, annuities and life insurance contracts. The assets of said trust will be held separate and apart from other Corporation funds, and shall be used exclusively for the purposes set forth in the Plan and the applicable trust agreement, subject to the following conditions:
 - (i) the creation of said trust shall not cause the Plan to be other than "unfunded" for purposes of Title I of ERISA;
 - (ii) the Corporation shall be treated as "grantor" of said trust for purposes of Section 677 of the Code; and
 - (iii) said trust agreement shall provide that its assets may be used to satisfy claims of the grantor's general creditors in the event of its insolvency, and the rights of such general creditors are enforceable by them under federal and state law.

-30-

(c) All Accounts under the Plan shall be for bookkeeping purposes only and shall not represent a claim against specific assets of the Corporation. Nothing contained in this Plan shall be deemed to create a trust of any kind or create any fiduciary relationship. Payments of amounts credited to Accounts under the Plan with respect to those Members and Beneficiaries for whom Trust contributions are made shall be made at the Corporation's option by the Corporation or from the Trust in accordance with the terms of the Trust, but, to the extent not paid by the Corporation, shall be paid by the Trust.

6.03 No Contract of Employment

The establishment of the Plan shall not be construed as conferring any legal rights upon any person for a continuation of employment, nor shall it interfere with the rights of the Corporation to discharge any officer and to treat him without regard to the effect which such treatment might have upon him as a Member of the Plan.

6.04 Facility of Payment

In the event that the Corporation shall find that a Member is unable to care for his affairs because of illness or accident, the Corporation may direct that any benefit payment due him, unless claim shall have been made therefor by a duly appointed legal representative, be paid to his spouse, a child, a parent or other blood relative, or to a person with whom he resides, and any such payment so made shall be a complete discharge of the liabilities of the Plan therefor.

6.05 Withholding Taxes

The Corporation shall have the right to deduct from each payment to be made under the Plan any required withholding taxes. In the event employment tax liability or state or local tax liability is assessed on amounts paid or payable under this Plan, the Corporation shall have the right to deduct from the payment or from the Member's other Compensation any required employee portion of the employment tax liability or income tax withholding.

6.06 Nonalienation

Subject to any applicable law, no benefit under the Plan shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance or charge, and any attempt so to do shall be void, nor shall any such benefit be in any manner liable for or subject to garnishment, attachment, execution or levy, or liable for or subject to the debts, contracts, liabilities, engagements or torts of the Member or any liability for equitable distribution, alimony or other payments for the support of a spouse or former spouse, or for any other relative of any Member.

6.07 Construction

The Plan is intended to constitute an unfunded deferred compensation arrangement for a select group of management or highly compensated employees. Except to the extent superseded by Federal law, all rights hereunder shall be governed by and construed in accordance with the laws of the State of Ohio (including its statute or limitations provisions). The Plan shall be construed to effectuate its purpose and the Corporation's intent that the Plan be exempt from ERISA, as amended and that amounts deferred hereunder not be subject to Federal income tax until distributed.

-32-

Except as otherwise provided by law, any action to enforce a right or obligation hereunder shall be brought in a court of competent jurisdiction in the County of Franklin and State of Ohio.

6.08 Limitations on Liability

Notwithstanding any of the preceding provisions of the Plan, neither the Corporation nor any individual acting as employee or agent of the Corporation shall be liable to any Member, former Member or other person for any claim, loss, liability or expense incurred in connection with the Plan. The Corporation does not undertake any responsibility to any Member for the tax consequences of a particular Member's election to defer income under this Plan.

6.09 Administrative Expense

All expenses of administering this Plan shall be paid by the Corporation and no part of the expenses or taxes on the Corporation shall be charged against any Member's Accounts or any benefits distributed under the Plan.

6.10 Claims and Review Procedures

(a) In accordance with any rules and procedures adopted by the Corporation, applications for benefits shall be submitted to the Corporation on a prescribed form signed by the Member or, in the case of a death benefit, by his Beneficiary.

The Corporation shall, within 90 days after the receipt of a written claim, send written notification to the Participant or Beneficiary (referred to in the remainder of this Section as the "claimant") as to its disposition, unless special circumstances require an extension of time for processing the claim. If such an extension is required, written notice of the extension shall be furnished to the claimant prior to the termination of the initial ninety (90)-day period. In no event shall such extension exceed a period of ninety (90) days from the end of such initial period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Corporation expects to render the final decision.

In the event the claim is wholly or partially denied, the written notification shall state the specific reason or reasons for the denial, include specific references to pertinent Plan provisions on which the denial is based, provide an explanation of any additional material or information necessary for the claimant to perfect the claim and a statement of why such material or information is necessary, and set forth the procedure by which the claimant may appeal the denial of the claim. If the claim has not been granted and notice is not furnished within the time period specified in the preceding paragraph, the claim shall be deemed denied for the purpose of proceeding to appeal in accordance with paragraph (b) below.

(b) In the event a claimant wishes to appeal the denial of his claim, he may request a review of such denial by making written application to the Corporation within sixty (60) days after receipt of the written notice of denial (or the date on which such claim is deemed denied if written notice is not received within the applicable time

period specified in paragraph (a) above). Such claimant (or his duly authorized representative) may, upon written request to the Corporation, review documents which are pertinent to such claim, and submit in writing issues and comments in support of his position. Within sixty (60) days after receipt of the written appeal (unless an extension of time is necessary due to special circumstances or is agreed to by the parties, but in no event more than one hundred twenty (120) days after such receipt), the Corporation shall notify the claimant of its final decision. Such final decision shall be in writing and shall include specific reasons for the decision and specific references to the pertinent Plan provisions on which the decision is based. If an extension of time for review is required because of special circumstances, written notice of the extension shall be furnished to the claimant prior to the commencement of the extension. If the claim has not been granted and written notice is not provided within the time period specified above, the appeal shall be deemed denied.

(c) If the claimant does not follow the procedures set forth in paragraphs (a) and (b) above, the claimant shall be deemed to have waived his right to appeal benefit determinations under the Plan. In addition, the decisions, actions, and records of the Corporation shall be conclusive and binding upon the Corporation and all persons having or claiming to have any right or interests in or under the Plan.

6.11 Illegal or Invalid Provision

In case any provision of the Plan shall be held illegal or invalid for any reason, such illegal or invalid provision shall not affect the remaining parts of the Plan, but the Plan shall be construed and enforced without regard to such illegal or invalid provision.

-35-

6.12 Gender and Number

Except as otherwise indicated by context, masculine terminology used herein also includes the feminine, and terms used in the singular may also include the plural.

6.13 Successors

All obligations of the Corporation under the Plan shall be binding on any successor to the Corporation, whether the existence of such successor is the result of a direct or indirect purchase of all or substantially all of the business and/or assets of the Corporation, or a merger, consolidation or otherwise.

ARTICLE 7. AMENDMENT OR TERMINATION

The Corporation reserves the right to modify or to amend, in whole or in part, or to terminate this Plan any time by action of its Compensation Committee, taken at a meeting held either in person or by telephone or other electronic means, or by unanimous written consent by lieu of a meeting. However, no modification or amendment of the Plan shall adversely affect the right of any Member to receive the benefits granted under the Plan by the Corporation in respect to such Member as of the date of modification or amendment.

If the Plan is terminated, payments from the Accounts of all Members and Beneficiaries shall be made as soon as administratively convenient in the form of monthly payments over a three (3)-year period; however, the Compensation Committee, in its sole discretion, may pay benefits in a lump sum.

Notwithstanding the foregoing, following a Change in Control no modification, amendment or termination of the Plan shall change the right of any Member to direct investments, to direct the investment forms, to make distribution elections whether in-service or at termination of service from those rights the Member had at the date of modification, amendment or termination, without the written consent of a majority in number of the Members, except when to comply with legal or regulatory requirements necessary to maintain the tax deferred status of any Deferrals.

CERTIFICATION

I, Thomas E. Hoaglin, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Huntington Bancshares Incorporated;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared; and
 - b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2004

/s/ Thomas E. Hoaglin

Thomas E. Hoaglin Chief Executive Officer

CERTIFICATION

I, Donald R. Kimble, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Huntington Bancshares Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared; and
 - b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2004

/s/ Donald R. Kimble

Donald R. Kimble Chief Financial Officer and Controller

SECTION 1350 CERTIFICATION

In connection with the Quarterly Report of Huntington Bancshares Incorporated (the "Company") on Form 10-Q for the three month period ended June 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Thomas E. Hoaglin, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Thomas E. Hoaglin

Thomas E. Hoaglin Chief Executive Officer August 9, 2004

SECTION 1350 CERTIFICATION

In connection with the Quarterly Report of Huntington Bancshares Incorporated (the "Company") on Form 10-Q for the three month period ended June 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Donald R. Kimble, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Donald R. Kimble

Donald R. Kimble Chief Financial Officer and Controller August 9, 2004